UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Schedule 14A

	Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No.)
File	d by the Registrant
File	d by a party other than the Registrant \Box
Che	ck the appropriate box:
	Preliminary Proxy Statement
	Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))
\boxtimes	Definitive Proxy Statement
	Definitive Additional Materials
	Soliciting Material under §240.14a-12
	Enzo Biochem, Inc. (Name of Registrant as Specified In Its Charter)
	(Name of Person(s) Filing Proxy Statement, if other than the Registrant)
Pay	ment of Filing Fee (Check all boxes that apply):
	No fee required.
\boxtimes	Fee paid previously with preliminary materials.
	Fee computed on table in exhibit required by Item 25(b) per Exchange Act Rules 14a6(i)(1) and 0-11.



ENZO BIOCHEM, INC. 81 Executive Blvd. Suite 3 Farmingdale, New York 11735

NOTICE OF SPECIAL MEETING OF SHAREHOLDERS To be held on May 22, 2023

To All Shareholders of Enzo Biochem, Inc.:

NOTICE IS HEREBY GIVEN that a Special Meeting of Shareholders (the "Special Meeting") of Enzo Biochem, Inc., a New York corporation ("we" or the "Company"), will be held virtually via live webcast at www.virtualshareholdermeeting.com/ENZ2023SM on May 22, 2023 at 9:00 a.m., EDT, for the following purposes.

Our board of directors (our "Board") adopted and approved, and on March 16, 2023 we entered into, an Asset Purchase Agreement (the "Asset Purchase Agreement") with Laboratory Corporation of America Holdings, a Delaware corporation ("Labcorp" or the "Buyer"), whereby our subsidiary Enzo Clinical Labs, Inc. will sell substantially all the operating assets and assign certain liabilities of our clinical laboratory business (the "Business") to the Buyer (the "Asset Sale"). If the Asset Sale is completed, we may be required to provide certain transition services to the Buyer for a period of time following the closing of the Asset Sale as described more fully in the attached Proxy Statement.

At the Special Meeting or any postponement or adjournment thereof, you will be asked to approve the Asset Sale and adopt the Asset Purchase Agreement. You will also be asked to approve a proposal to adjourn the Special Meeting if there are insufficient votes at the time of the Special Meeting to approve the Asset Sale.

After careful consideration, our Board adopted and approved the Asset Purchase Agreement and determined that the Asset Sale is fair to, advisable and in the best interests of, the Company and its shareholders. **Our Board recommends that you vote:**

- "FOR" the approval of the Asset Sale and adoption of the Asset Purchase Agreement (the "Asset Sale Proposal"); and
- "FOR" the proposal to adjourn the Special Meeting to solicit additional proxies if there are insufficient votes to approve the Asset Sale Proposal at the time of the Special Meeting (the "Adjournment Proposal").

The proxy statement attached to this letter provides you with information about the proposed Asset Sale and the Special Meeting. I encourage you to read the entire proxy statement carefully. You may also obtain additional information about us from documents filed with the U.S. Securities and Exchange Commission (the "SEC"), including our financial statements and notes thereto included in our most recent filing on Form 10-K for the fiscal year ended July 31, 2022 and subsequent periodic and interim report filings (see the section entitled "Where You Can Find More Information" beginning on page 55).

Only shareholders of record as of the close of business on April 17, 2023, the date fixed by the Board as the record date for the Special Meeting, are entitled to notice of, and to vote at, the Special Meeting. This Special Meeting will be a completely virtual meeting. There will be no physical meeting location. The meeting will be conducted via live webcast. Shareholders will have the same rights and opportunities to participate in our virtual Special Meeting as they would at an in-person meeting. For more information on our virtual Special Meeting, including details on how to attend the meeting, see the instructions under "Instructions for the Virtual Special Meeting" on page iii of this Proxy Statement.

Your vote is very important. The Asset Sale cannot be completed unless the Asset Sale Proposal is approved by the affirmative vote of a majority of the shares of our common stock, par value \$0.01 per share ("Common Stock"), outstanding and entitled to vote thereon. Failing to vote on the Asset Sale Proposal, or to instruct your broker or other nominee on how to vote, will have the same effect as a vote "AGAINST" the Asset Sale Proposal.

Whether or not you plan to attend the Special Meeting, and regardless of the number of shares of Common Stock you own, please vote as soon as possible. We urge you to date, sign and return the proxy card in the envelope provided to you, or to use the telephone or Internet method of voting described on your proxy card, even if you plan to attend the Special Meeting, so that if you are unable to attend the Special Meeting, your shares can be voted. Any shareholder of record who submits a proxy card retains the right to revoke such proxy card by: (i) submitting a written notice of such revocation to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; (ii) submitting a duly signed proxy card bearing a later date than the previously signed and dated proxy card to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; or (iii) attending the Special Meeting virtually and voting thereat the shares represented by such proxy card. Attendance at the Special Meeting will not, in and of itself, constitute revocation of a completed, signed and dated proxy card previously returned. All such later-dated proxy cards or written notices revoking a proxy card should be sent to Enzo Biochem, Inc., 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Hamid Erfanian, Chief Executive Officer. If you hold shares of Common Stock in street name, you must contact the firm that holds your shares to change or revoke any prior voting instructions.

Please carefully read the enclosed Proxy Statement, which explains the proposals to be considered by you and acted upon at the Special Meeting. Please consult your financial, income tax or other professional advisor as appropriate.

The Board has fixed the close of business on April 17, 2023 as the record date for the determination of holders of record of Common Stock entitled to notice of, and to vote at, the Special Meeting. A list of shareholders of record of the Company as of the record date will remain open for inspection during the Special Meeting until the closing of the polls.

If you have any questions about the procedures for admission to the Special Meeting, please contact Investor Relations at (631)-755-5500.

April 24, 2023	By Order of the Board of Directors,
	/s/ Hamid Erfanian
	Hamid Erfanian
	Chief Executive Officer

Neither the Securities and Exchange Commission nor any state securities regulatory agency has approved or disapproved the Asset Sale or passed upon the adequacy or accuracy of the disclosure in the attached proxy statement. Any representation to the contrary is a criminal offense.

ALL HOLDERS OF RECORD OF COMMON STOCK (WHETHER THEY INTEND TO ATTEND THE SPECIAL MEETING OR NOT) ARE STRONGLY ENCOURAGED TO COMPLETE, SIGN, DATE AND RETURN PROMPTLY THE PROXY CARD ENCLOSED WITH THE ACCOMPANYING PROXY STATEMENT.

ENZO BIOCHEM, INC. PROXY STATEMENT SPECIAL MEETING OF SHAREHOLDERS To be held on May 22, 2023

This Proxy Statement is being furnished in connection with the solicitation of proxies by and on behalf of the Board of Directors (the "Board") of Enzo Biochem, Inc., a New York corporation ("we" or the "Company") and is being first made available to our shareholders on April 24, 2023, for use at a Special Meeting of Shareholders (the "Special Meeting") to be held virtually via live webcast at www.virtualshareholdermeeting.com/ENZ2023SM, on May 22, 2023 at 9:00 a.m., EDT, for the following purposes:

- (1) to consider and vote on a proposal to approve the sale (the "Asset Sale") of substantially all the assets and assignment of certain liabilities of our clinical laboratory business (the "Business") to Laboratory Corporation of America Holdings, a Delaware corporation ("Labcorp" or the "Buyer") and adopt the Asset Purchase Agreement, dated as of March 16, 2023 (as such agreement may be amended from time to time, the "Asset Purchase Agreement"), by and among the Company, Enzo Clinical Labs, Inc., a New York corporation (the "Seller") and the Buyer (the "Asset Sale Proposal"); and
- (2) to consider and vote on a proposal to approve the adjournment of the Special Meeting, if necessary or appropriate, to solicit additional proxies if there are insufficient votes at the time of the Special Meeting to approve the Asset Sale Proposal (the "Adjournment Proposal").

Only shareholders of record of our Common Stock as of the close of business on April 17, 2023 are entitled to notice of, and to vote at, the Special Meeting and any adjournment of the Special Meeting. If you hold shares through a broker or other nominee, you must follow the procedures provided by your broker or other nominee in order to vote your shares at the Special Meeting.

The Board may use the services of the Company's directors, officers and other regular employees to solicit proxies personally or by telephone and may request brokers, fiduciaries, custodians and nominees to send proxies, proxy statements and other materials to their principals and reimburse them for their out-of-pocket expenses in so doing. We have also retained Kingsdale Advisors, a proxy advisory and solicitation firm, to assist with the solicitation of proxies and provide related advice and informational support, for a services fee, plus customary disbursements, which are not expected to exceed \$20,000 in total. The cost of services and solicitation of proxies, estimated to be approximately \$20,000, will be borne by the Company. Each proxy duly executed and returned to the Company is revocable. The procedure a shareholder must follow to revoke its proxy depends on how such shareholder holds its shares. Registered holders of our Common Stock may revoke a previously submitted proxy by submitting another valid later dated proxy or by providing a written notice of such revocation to the Chief Executive Officer of the Company before the closing of the polls at the Special Meeting. Only the latest dated validly executed proxy will count. Shareholders also may revoke any previously submitted proxy by attending the Special Meeting and voting their shares during the Special Meeting. Note that simply attending the Special Meeting without taking one of the above actions will not revoke a proxy. In general, shareholders holding shares in "street name" may revoke previously submitted voting instructions by submitting to their custodian another valid voting instruction or a signed letter of revocation. Shareholders holding shares in "street name" should contact their custodian for detailed instructions on how to revoke their voting instruction and the applicable deadlines.

Instructions for the Virtual Special Meeting

The Special Meeting will be a completely virtual meeting. There will be no physical meeting location. The Special Meeting will be conducted via live webcast. Shareholders will have the same rights and opportunities to participate in the virtual Special Meeting as they would at an in-person meeting.

To attend the virtual Special Meeting, visit www.virtualshareholdermeeting.com/ENZ2023SM and enter the 16-digit control number included on your proxy card or voting instruction form. The meeting will start at 9:00 a.m., EDT on May 22, 2023. We encourage you to access the virtual Special Meeting prior to the start time to familiarize yourself with the virtual platform and ensure you can hear the streaming audio. Online access will be available starting at 8:30 a.m., EDT on May 22, 2023.

The virtual Special Meeting platform is fully supported across browsers (Firefox, Chrome, and Safari) and devices (desktops, laptops, tablets and cell phones) running the most updated version of applicable software and plugins. Participants should ensure that they have a strong WiFi connection from wherever they intend to participate in the virtual Special Meeting.

While we strongly encourage you to vote your shares prior to the virtual Special Meeting, shareholders may also vote during the meeting. Once logged in, you will be able to vote your shares by clicking the "Vote Here!" button.

Shareholders may submit written questions once logged into the virtual Special Meeting platform. Questions pertinent to meeting matters will be answered during the question and answer portion of the virtual Special Meeting, subject to a time limit prescribed by the Rules of Conduct that will be posted to the virtual Special Meeting platform on the day of the meeting. The Rules of Conduct will also provide additional information about the relevancy of questions to meeting matters.

If you encounter any difficulties accessing the virtual Special Meeting during the check-in or meeting time, you should call the technical support number that will be posted on the virtual shareholder meeting login page.

Voting Securities and Votes Required

The Board has fixed the close of business on April 17, 2023 as the record date for determination of shareholders entitled to notice of, and to vote at, the Special Meeting (the "Record Date"). As of the Record Date, the Company had outstanding 49,662,821 shares of Common Stock. Each share of Common Stock entitles the record holder thereof to one vote on each matter brought before the Special Meeting. No shares of preferred stock were outstanding as of the Record Date. The holders of a majority of the outstanding shares of Common Stock, represented in person or by proxy, constitutes a quorum.

Under New York law, you have the right to file a written objection including a notice of election to dissent to the Asset Sale and demand payment of the "fair value" of your shares with the Company. If you intend to dissent and demand payment of the "fair value" of your shares, you must comply with the provisions of Section 623 of the New York Business Corporation Law in order to receive payment for your shares. We will require strict compliance with the statutory procedures. For more information about your right to dissent, please turn to "The Asset Sale — Dissenters' Rights" beginning on page 35.

A shareholder who abstains from voting on any or all proposals will be included in the number of shareholders present at the Special Meeting for the purpose of determining the presence of a quorum. Brokers are not permitted to vote shares without instructions on proposals that are not considered "routine." Your broker will only be permitted to vote your shares on the approval of the Asset Sale Proposal and the Adjournment Proposal if you instruct your broker how to vote. You should follow the procedures provided by your broker regarding the voting of your shares. If you do not instruct your broker to vote your shares, your shares will not be voted, and your shares will not be counted as present for purposes of determining the presence or absence of a quorum at the Special Meeting.

The approval of the Asset Sale Proposal (Proposal 1) will require the affirmative vote of a majority of the outstanding shares of Common Stock entitled to vote. Shareholders may vote "for," "against" or "abstain" with respect to the Asset Sale Proposal.

The approval of the Adjournment Proposal (Proposal 2) will require the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting.

If you are a shareholder whose shares are registered directly in your name with American Stock Transfer & Trust Company, LLC, our transfer agent, and you return a signed proxy card without giving specific voting instructions, then the individuals designated as proxy holders on the proxy card will vote your shares in accordance with the recommendations of the Board: (i) "FOR" Proposal 1, and (ii) "FOR" Proposal 2.

Please vote your proxy so your vote can be counted.

The enclosed materials require the Company's shareholders to make important decisions with respect to the Company. Please read carefully this Proxy Statement and its annexes, as these documents contain detailed information relating to the Asset Sale. If you are in doubt as to how to make these decisions, please consult your financial, legal or other professional advisors.

Company shareholders who have questions about matters to be voted on at the Special Meeting or who desire additional copies of this Proxy Statement or additional proxy or voting instruction cards should contact:

Enzo Biochem, Inc.

81 Executive Blvd. Suite 3 Farmingdale, New York 11735 (631) 755-5500 Attn.: Investor Relations

or



Kingsdale Advisors

+1 (888) 518-1554 (toll-free in North America) +1 (917) 341-5244 (call collect outside of North America) e-mail: contactus@kingsdaleadvisors.com

IMPORTANT NOTICE REGARDING AVAILABILITY OF PROXY MATERIALS

This Proxy Statement, the form of proxy card, and our Annual Report on Form 10K for our fiscal year ended July 31, 2022, are available to you on our website at www.enzo.com. Shareholders may also obtain a copy of these materials by writing to Enzo Biochem, Inc., 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Hamid Erfanian, Chief Executive Officer. Upon payment of a reasonable fee, shareholders may also obtain a copy of the exhibits to our Annual Report on Form 10-K for our fiscal year ended July 31, 2022.

TABLE OF CONTENTS

	Page
SUMMARY	1
The Companies	1
The Asset Sale Summary Term Sheet	1
The Special Meeting	4
Record Date; Stock Entitled to Vote; Quorum	5
Vote Required	5
Voting	5
Submitting Proxies Via the Internet or by Telephone	5
Revocability of Proxies	5
Recommendations of Our Board of Directors	5
QUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE ASSET SALE	6
The Special Meeting	6
The Asset Sale	7
General	9
SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS	10
THE SPECIAL MEETING	11
Date, Time, Place and Purpose of the Special Meeting	11
Recommendation of Our Board of Directors	11
Record Date; Stock Entitled to Vote; Quorum	11
Vote Required	11
Voting	11
Householding of Proxy Materials	12
Solicitation of Proxies	12
Other Business	13
THE ASSET SALE	14
General	14
Background of the Proposed Sale	14
Reasons for the Asset Sale and Recommendation of Our Board of Directors	17
Opinion of the Company's Financial Advisor	19
Unaudited Pro Forma Financial Information	26
Certain Effects of the Asset Sale	32
Effects on the Company if the Asset Sale is Not Completed	32
Interest of Certain Persons in the Asset Sale	32
Stockholdings and Stock Awards	33
Bonus Compensation	34
Indemnification of Officers and Directors	34
Regulatory Matters	34
Dissenters' Rights	35
MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE ASSET SALE	40
PROPOSAL 1 — TO APPROVE THE ASSET SALE	41
Structure of the Asset Sale	41
Principal Provisions of the Asset Purchase Agreement	41
Benefit Arrangements	45
Further Assurances and Cooperation	46
Other Significant Provisions	46
Risks Related to the Asset Sale	47
Vote Required and Board Recommendation	49

	Page
PROPOSAL 2 — TO APPROVE THE ADJOURNMENT OF THE SPECIAL MEETING, IF NECESSARY OR APPROPRIATE, TO SOLICIT ADDITIONAL PROXIES IF THERE ARE INSUFFICIENT VOTES AT THE TIME OF THE SPECIAL MEETING TO APPROVE THE	
ASSET SALE PROPOSAL	50
Adjournment of the Special Meeting	50
Vote Required and Board Recommendation	50
MARKET PRICE AND DIVIDEND DATA	51
SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	52
FUTURE SHAREHOLDER PROPOSALS	54
WHERE YOU CAN FIND MORE INFORMATION	55
ANNEX A — ASSET PURCHASE AGREEMENT	A-1
ANNEX B — OPINION OF JEFFERIES LLC	B-1
ANNEX C — NEW YORK BUSINESS CORPORATION LAW SECTIONS 623 AND 910	C-1

vii

SUMMARY

This summary highlights information included elsewhere in this Proxy Statement. This summary does not contain all of the information you should consider before voting on the Proposals presented in this Proxy Statement. You should read the entire Proxy Statement carefully, including the unaudited pro forma financial statements contained herein and the annexes attached hereto. For your convenience, we have included cross references to direct you to a more complete description of the topics described in this summary.

The Companies

ENZO BIOCHEM, INC. 81 Executive Blvd. Suite 3 Farmingdale, New York 11735 (631) 755-5500

Enzo Biochem, Inc. (the "Company", "we", "our" or "Enzo") is an integrated diagnostics, clinical lab, and life sciences company focused on delivering and applying advanced technology capabilities to produce affordable reliable products and services that enable our customers to meet their clinical needs. We provide advanced biotechnology solutions to the global community as affordable and flexible quality products and services. We develop, manufacture and sell our proprietary technology solutions and platforms to clinical laboratories, specialty clinics, and researchers globally. Enzo's structure and business strategy represent the culmination of years of extensive planning and work. The Company has the ability to offer low cost, high performance products and services for diagnostic testing, which ideally positions us to capitalize on the reimbursement pressures facing diagnostic labs. Our pioneering work in genomic analysis coupled with our extensive patent estate and enabling platforms have positioned the Company to continue to play an important role in the rapidly growing molecular medicine marketplaces. Our common stock is currently registered under the Securities Exchange Act of 1934, which we refer to as the Exchange Act, and is listed on The New York Stock Exchange under the symbol "ENZ".

Laboratory Corporation of America Holdings ("Labcorp" or the "Buyer") 358 South Main Street Burlington, North Carolina 27215 (336) 229-1127

Labcorp is a leading global life sciences company that provides vital information to help doctors, hospitals, pharmaceutical companies, researchers, and patients make clear and confident decisions. Through Labcorp's diagnostics and drug development capabilities, it provides insights and accelerates innovations to improve health and improve lives. With over 80,000 employees, Labcorp serves customers in more than 100 countries. Labcorp's common stock is listed on the New York Stock Exchange under the symbol "LH".

The Asset Sale Summary Term Sheet

This following discussion of the Asset Sale is qualified by reference to the Asset Purchase Agreement, which is attached to this proxy statement as **Annex A**. You should read the entire Asset Purchase Agreement carefully as it is the legal document that governs the Asset Sale.

- The Asset Sale. On March 16, 2023, we entered into an Asset Purchase Agreement with the Buyer, which provides for the sale of substantially all the operating assets and the assignment of certain liabilities of our clinical laboratory business (the "Business") to the Buyer for an aggregate purchase price of approximately \$146 million (the "Purchase Price") payable in cash at closing, subject to certain adjustments and conditions described below and set forth in their entirety in the attached Asset Purchase Agreement. For more information, see the section below captioned "THE ASSET SALE" beginning on page 14, and "PROPOSAL 1 TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement" beginning on page 41.
- Purchase Price Adjustment. Pursuant to the attached Asset Purchase Agreement, the Purchase
 Price to be paid by the Buyer is subject to the following price adjustments: (i) less, (A) the dollar
 value of any accrued, but unused employee vacation time of Business employees Buyer hires at
 closing, (B) the Escrow Amount, (C) the dollar amount of debt underlying certain liens on the assets
 of the Business,

which is to be paid off at the closing, and (D) any amounts deducted for certain motor vehicles not delivered at the closing and (ii) plus, (X) the amount of all reserves, security and other deposits, advances and prepaid expenses and credits that are assignable to Buyer and related to the assigned contracts or any other acquired asset and (Y) a portion of the retention bonuses paid by the Company to its Business employees. For more information, see the section below captioned "PROPOSAL I—TO APPROVE THE ASSET SALE—Principal Provisions of the Asset Purchase Agreement-Consideration to be Received in the Asset Sale and Purchase Price Adjustments" beginning on page 41.

- Assets to be Sold to the Buyer. The assets to be purchased by the Buyer constitute substantially all the assets of the Business, including customer accounts, sales data, records, and promotional materials, customer and vendor lists, inventory and supplies, tangible personal property and equipment, reserves, security deposits, prepaid expenses, and credits related to assigned contracts. The acquired assets also include certain intellectual property assets used exclusively in the Business, such as trademarks and domain names used in the Business. We will retain our cash, cash equivalents and accounts receivable, among other assets. For more information, see the section below captioned "PROPOSAL I TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement-Assets to be Sold" beginning on page 41.
- Liabilities to be Assumed by the Buyer. The Buyer will assume only certain liabilities arising from
 certain leases and other contracts. All other liabilities will remain our obligations, including
 transaction costs. For more information, see the section below captioned "PROPOSAL 1 TO
 APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement-Liabilities
 to be Assumed" beginning on page 41.
- Conditions to the Closing of the Asset Sale. The consummation of the Asset Sale is subject to the satisfaction or waiver of customary conditions on or prior to the closing as set forth in the Asset Purchase Agreement. Such conditions include the expiration or termination of the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 ("HSR") and the approval of the Asset Sale by shareholders of the Company holding a majority of the shares of its common stock outstanding entitled to vote thereon. Subject to the terms and conditions in the Asset Purchase Agreement, after shareholder approval of the Asset Sale, Buyer will be obligated to pay a fee of \$75,000 to the Company for each day after the date of such shareholder approval until the closing of the Asset Sale, provided that such fee shall not exceed an aggregate of \$7,000,000. At the closing, such fee will be wholly or partially credited against the purchase price. For more information, see the section below captioned "PROPOSAL 1 TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement-Conditions to the Closing of the Asset Sale" beginning on page 42.
- Terms of Indemnification and Escrow. Except as otherwise provided in the Asset Purchase Agreement, the representations and warranties of the Company and the right of the Buyer to indemnification under the Asset Purchase Agreement generally survive the Closing for 15 months. The Asset Purchase Agreement provides that Buyer shall deposit at the closing \$5 million (the "Escrow Amount") into an escrow account for the purpose of satisfying claims, if any, made by Buyer against the Seller pursuant to the indemnification provisions in the Asset Purchase Agreement relating to breaches of the Company's representations and warranties. The Escrow Amount will be held in the escrow account for a period of fifteen months after the closing of the Asset Sale. Subject to the survival period for representations and warranties of the Company and the Seller, and other than in the case of fraud or breach of certain specified representations and warranties (the "Fundamental Representations") or breach of representations and warranties related to certain tax matters, the Seller shall not be liable to the Buyer for indemnification unless and until the aggregate amount of the indemnification obligations exceeds \$584,000 (the "Deductible"), in which event the Seller shall then be liable for losses in excess of the Deductible up to an amount equal to \$584,000 (payable from the Escrow Amount) and the remainder of the losses shall be satisfied from Buyer's representations and warranty insurance policy ("R&W Policy"). With respect to claims by Buyer regarding certain other tax matters and branches of Fundamental Representations, subject to certain limitations, Buyer is entitled to satisfy losses first from the Escrow Amount (up to an amount equal to the Deductible), second from its R&W Policy, and last from the Escrow Amount or, if exhausted, from Seller directly up to the amount of the Purchase Price. With respect to any claim by the Buyer regarding (i) excluded liabilities, excluded assets or the excluded

business specified in the Asset Purchase Agreement, or (ii) breach, default of, or failure by Seller or the Company to perform the covenants or obligations of the Asset Purchase Agreement, Buyer is entitled to satisfy losses first from the Escrow Fund and then from the Seller directly.

- Transition Period. Pursuant to the Asset Purchase Agreement, we may be required to provide transition services to the Buyer for up to six months following the closing (the "Transition Period"). For more information, see the section below captioned "PROPOSAL 1 TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement-Transition Period" beginning on page 44.
- Termination. The Asset Purchase Agreement may be terminated by either the Company or the
 Buyer subject to customary conditions. For more information, see the section below captioned
 "PROPOSAL 1 TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase
 Agreement-Termination" beginning on page 44.
- Breakup Fee. The Asset Purchase Agreement provides that, in connection with the termination of the Asset Purchase Agreement, under specified circumstances, the Company will be required to pay Buyer a termination fee of \$5,000,000 (including termination by the Company in order to enter into a definitive agreement with respect to a superior proposal) or reimburse Buyer's expenses of up to \$5,000,000 (in connection with termination for failure to obtain shareholder approval). For more information, see the section below captioned "PROPOSAL I TO APPROVE THE ASSET SALE Principal Provisions of the Asset Purchase Agreement-Breakup Fee" beginning on page 44.
- No Solicitation. The Asset Purchase Agreement contains a "no shop" restriction on the Company's ability to solicit proposals from, provide confidential information to, and engage in discussions and negotiations with, potential third-party bidders of the Business. The "no shop" restriction contains a "fiduciary out" provision whereby the Company may engage with a third-party bidder from whom the Company receives an unsolicited bona fide written acquisition proposal after the date of the Asset Purchase Agreement and prior to obtaining shareholder approval which our Board determines in good faith after consultation with outside legal counsel that (i) failure to take action would reasonably be expected to breach the Board's fiduciary duties under applicable law, (ii) the proposal is, or would reasonably be expected to lead to, an alternative transaction that is more favorable to the Company's shareholders from a financial point of view than the Asset Sale and (iii) the proposal is reasonably likely to be consummated on the terms proposed and to constitute a Superior Proposal, as defined in the Asset Purchase Agreement.

At any time prior to obtaining shareholder approval of the Asset Sale, in response to a Superior Proposal, our Board may, among other actions, withdraw or modify its recommendation contained in this proxy statement to approve the Asset Sale Proposal or recommend the approval of an alternative acquisition proposal, if our Board concludes in good faith (after consultation with outside counsel) that the failure to take such action would reasonably likely to cause our Board to breach its fiduciary duties under applicable law (see previous section entitled "Breakup Fee" for details related to any fees we may be obligated to pay if our Board exercises its option to withdraw or materially modify its recommendation contained in this proxy statement). For more information, see the section below captioned "PROPOSAL I — TO APPROVE THE ASSET SALE — Principal Provisions of the Asset Purchase Agreement-No Solicitation" beginning on page 45.

• Opinion of the Company's Financial Advisor. The Company has engaged Jefferies LLC ("Jefferies") as financial advisor to the Company in connection with the Asset Sale. In connection with the Asset Sale, Jefferies delivered a written opinion, dated March 15, 2023, to the Board as to the fairness, from a financial point of view and as of such date, to the Company of the Purchase Price to be received for the Business in the Asset Sale pursuant to the Asset Purchase Agreement. The full text of Jefferies' opinion, which describes the various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, is attached as Annex B to this proxy statement and is incorporated herein by reference. Jefferies' opinion was provided for the use and benefit of the Board (in its capacity as such) in its evaluation of the Purchase Price from a financial point of view to the Company and did not address any other aspect of the Asset Sale or any other matter. Jefferies' opinion did not address the relative merits of the Asset Sale or other transactions contemplated thereby as compared to any alternative transaction or opportunity that might be available to the

Company or the Seller or with respect to the Business, nor did it address the underlying business decision by the Company to engage in the Asset Sale. Jefferies' opinion does not constitute a recommendation as to how the Board, any securityholder or any other party should vote or act with respect to the Asset Sale or any other matter. For more information, see the section below captioned "THE ASSET SALE — Opinion of the Company's Financial Advisor" beginning on page 19.

- Material U.S. Federal Income Tax Consequences. Our shareholders will not realize any gain or loss for U.S. federal income tax purposes as a result of the Asset Sale. The proposed Asset Sale will be treated as a sale of corporate assets in exchange for cash and the assumption of certain liabilities. The adjusted tax basis in the assets being sold is less than the sales proceeds that will be received from the Buyer in which case we will incur U.S. federal and state income tax as a result of a realized gain from the Asset Sale. However, we anticipate that our tax attributes, including our U.S. federal and state net operating loss carryforwards ("NOLs"), will be available to offset a substantial portion of our U.S. federal and state income tax liabilities resulting from such gain. For more information, see the section below captioned "THE ASSET SALE MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE ASSET SALE" beginning on page 40.
- Interests of Certain Persons in the Asset Sale. Our directors and executive officers may have interests in the Asset Sale that may be in addition to, or different from, the interests of our shareholders. In considering the recommendation of our Board with respect to the approval of the Asset Sale, our shareholders should be aware that our executive officers may have interests in the Asset Sale that are different from, or in addition to, those of our shareholders generally. Our Board was aware of these interests and considered them, among other matters, in approving the Asset Purchase Agreement and the Asset Sale making its recommendation that our shareholders vote "FOR" the Asset Sale Proposal. For more information, see the section below captioned "THE ASSET SALE Interests of Certain Persons in the Asset Sale" beginning on page 32.
- Voting Agreements. In connection with the entry into the Asset Purchase Agreement, on March 16, 2023, Labcorp entered into separate Voting Agreements (the "Voting Agreements") with certain of our directors and executive officers who collectively beneficially own up to approximately 11% of the Company's outstanding shares of Common Stock. The Voting Agreements provide that each shareholder shall, among other things, vote all of the Common Stock that it beneficially owns (i) in favor of adoption and approval of the Asset Purchase Agreement and the Asset Sale, (ii) against any alternative transaction (as described in the Asset Purchase Agreement) and (iii) against any other action that is intended or could reasonably be expected to impede or interfere with or materially delay the Asset Sale or any other transactions contemplated by the Asset Purchase Agreement. For more information, see the section below captioned "PROPOSAL 1 TO APPROVE THE ASSET SALE Voting Agreements" beginning on page 45.
- Dissenters' Rights. Under New York law, you have the right to file a written objection including a notice of election to dissent to the Asset Sale and demand payment of the "fair value" of your shares with the Company. If you intend to dissent and demand payment of the "fair value" of your shares, you must comply with the provisions of Section 623 of the New York Business Corporation Law in order to receive payment for your shares. We will require strict compliance with the statutory procedures. For more information about your right to dissent, please turn to "THE ASSET SALE Dissenters' Rights" beginning on page 35.

The Special Meeting (page 11)

The Special Meeting will be conducted via live webcast. Shareholders will have the same rights and opportunities to participate in the virtual Special Meeting as they would at an in-person meeting. To attend the virtual Special Meeting, visit www.virtualshareholdermeeting.com/ENZ2023SM and enter the 16-digit control number included on your proxy card or voting instruction form. The meeting will start at 9:00 a.m., EDT on May 22, 2023. We encourage you to access the virtual Special Meeting prior to the start time to familiarize yourself with the virtual platform and ensure you can hear the streaming audio. Online access will be available starting at 8:30 a.m., EDT on May 22, 2023.

At the Special Meeting, you will consider and vote upon proposals to (i) approve the Asset Sale Proposal, and (ii) approve the Adjournment Proposal.

Record Date; Stock Entitled to Vote; Quorum (page 11)

If you owned shares of our common stock at the close of business April 17, 2023, the record date for the Special Meeting, you are entitled to notice of and to vote at the special meeting. You have one vote for each share of our common stock that you own on the record date. As of the close of business on April 17, 2023, there were 49,662,821 shares of our common stock outstanding and entitled to be voted at the Special Meeting. The holders of a majority of the shares entitled to vote, represented in person or by proxy, shall constitute a quorum at the Special Meeting.

Vote Required (page 11)

For the Asset Sale Proposal and as a condition to the completion of the Asset Sale, approval requires the affirmative vote of a majority of the shares of our common stock outstanding and entitled to vote thereon.

Failing to vote or to instruct your broker or other nominee on how to vote will have the same effect as a vote "AGAINST" the Asset Sale Proposal.

For the Adjournment Proposal, approval requires the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting.

Each holder of a share of our common stock is entitled to one vote per share.

Voting (page 11)

You may grant a proxy by completing and returning the enclosed proxy card. If you hold your shares through a broker or other nominee, you should follow the procedures provided by your broker or nominee.

Submitting Proxies via Internet or by Telephone (page 13)

We urge you to date, sign and return the proxy card in the envelope provided to you, or to use the telephone or Internet method of voting described on your proxy card, even if you plan to attend the Special Meeting, so that if you are unable to attend the Special Meeting, your shares can be voted.

Revocability of Proxies (page 13)

Any shareholder of record who submits a proxy card retains the right to revoke such proxy card by: (i) submitting a written notice of such revocation to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; (ii) submitting a duly signed proxy card bearing a later date than the previously signed and dated proxy card to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; or (iii) attending the Special Meeting virtually and voting thereat the shares represented by such proxy card. Attendance at the Special Meeting will not, in and of itself, constitute revocation of a completed, signed and dated proxy card previously returned. All such laterdated proxy cards or written notices revoking a proxy card should be sent to Enzo Biochem, Inc., 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Hamid Erfanian, Chief Executive Officer. If you hold shares of Common Stock in street name, you must contact the firm that holds your shares to change or revoke any prior voting instructions.

Recommendations of Our Board of Directors (page 11)

After careful consideration, our Board has determined that the Asset Purchase Agreement and the Asset Sale are fair to, advisable and in the best interests of the Company and its shareholders. For a description of the factors considered by our Board in reaching its decision to adopt and approve the Asset Purchase Agreement and recommend its approval, see "THE ASSET SALE — Reasons for the Asset Sale and Recommendation of Our Board of Directors" on page 17. Accordingly, our Board has adopted and approved the Asset Purchase Agreement and recommends that you vote (i) "FOR" the approval of the Asset Sale Proposal, and (ii) "FOR" the approval of the Adjournment Proposal.

OUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE ASSET SALE

The following are some questions that you, as a shareholder of the Company, may have regarding the Special Meeting, including the Asset Sale Proposal, and brief answers to such questions. We urge you to carefully read this entire proxy statement, including the unaudited pro forma financial statements contained herein and the annexes to this proxy statement and the documents referred to in this proxy statement because the information in this section does not provide all the information that may be important to you as a shareholder of the Company with respect to the Asset Sale Proposal. See "Where You Can Find More Information" beginning on page 55.

Q. Who is soliciting my proxy?

A. This proxy is being solicited by our board of directors (our "Board").

Q. What matters will be voted on at the Special Meeting?

A. You will be asked to vote on the following proposals:

- to approve the Asset Sale Proposal; and
- to approve the Adjournment Proposal.

Q. How is a quorum calculated?

A. The holders of a majority of the votes of the shares of common stock of the Company issued and outstanding and entitled to vote must be present in person or by proxy to constitute a quorum for the transaction of any business. Signed proxies received but not voted will be included in the calculation of the number of shares considered to be present at the meeting.

Q. How does the Company's Board of Directors recommend that I vote on the proposals?

A. Our Board recommends that you vote in favor of the Asset Sale Proposal and the Adjournment Proposal.

Q. What vote is required for the Company shareholders to approve the proposals?

A. In order to approve the Asset Sale Proposal, the proposal must be approved by the affirmative vote of a majority of the shares of Common Stock outstanding and entitled to vote thereon.

Approval of the Adjournment Proposal requires the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting.

On April 17, 2023 (the record date), there were 49,662,821 outstanding shares of our common stock.

Pursuant to the Voting Agreements, certain of our directors and executive officers who collectively beneficially own up to approximately 11% of the Company's outstanding shares of Common Stock have agreed to, among other things, vote all of the Common Stock that each of them beneficially owns (i) in favor of adoption and approval of the Asset Purchase Agreement and the Asset Sale, (ii) against any alternative transaction (as described in the Asset Purchase Agreement); and (iii) against any other action that is intended or could reasonably be expected to impede or interfere with or materially delay the Asset Sale or any other transactions contemplated by the Asset Purchase Agreement. For more information, see the section below captioned "PROPOSAL 1 — TO APPROVE THE ASSET SALE — Voting Agreements" beginning on page 45.

O. Who is entitled to vote at the Special Meeting?

A. Holders of record of our common stock as of the close of business on April 17, 2023, the record date for the Special Meeting, are entitled to receive notice of and to vote at the special meeting. On the record date, 49,662,821 shares of our common stock, held by approximately 742 holders of record, were outstanding and entitled to vote. You may vote all shares you owned as of the record date. You are entitled to one vote per share.

Q. What should I do now?

A. After carefully reading and considering the information contained in this proxy statement, including the annexes hereto, please submit a proxy for your shares by returning the enclosed proxy card. Alternatively, you may follow the instructions on the proxy card to submit your proxy by telephone or via the Internet. You may also attend the special meeting and vote in person. Please do NOT enclose or return your stock certificate(s) with your proxy.

Q. If my shares are held in "street name" by my broker, will my broker vote my shares for me?

A. Your broker will only be permitted to vote your shares on the approval of the Asset Sale Proposal and the Adjournment Proposal if you instruct your broker how to vote. You should follow the procedures provided by your broker regarding the voting of your shares. If you do not instruct your broker to vote your shares, your shares will not be voted, and your shares will not be counted as present for purposes of determining the presence or absence of a quorum at the Special Meeting.

Q. How are the votes to approve the Asset Sale Proposal counted?

A. Approval of the Asset Sale Proposal, as a condition to the completion of the Asset Sale, requires the affirmative vote of a majority of the shares of our common stock outstanding and entitled to vote thereon. Accordingly, (1) an abstention from voting, (2) a shareholder's failure to submit a proxy card or to vote in person at the Special Meeting or (3) the failure of a shareholder who holds shares in "street name" through a broker or other nominee to give voting instructions to such broker or other nominee, will have the same effect as a vote "AGAINST" the approval of the Asset Sale Proposal.

Q. How are votes to approve the Adjournment Proposal counted?

A. For the Adjournment Proposal, approval requires the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting. A shareholder's failure to submit a proxy card or to vote in person at the Special Meeting will cause those shares to be not present for purposes of determining a quorum but, assuming a quorum is present, will have no effect on the outcome of the voting with respect to the Adjournment Proposal. An abstention will have no effect on the outcome of the vote with respect to the Adjournment Proposal.

Q. What happens if I return my proxy card, but I do not indicate how to vote?

A. If you return a signed proxy card without indicating your vote, your shares will be voted "FOR" the approval of the Asset Sale Proposal and the Adjournment Proposal.

Q. When should I send in my proxy card?

A. You should send in your proxy card as soon as possible so that your shares will be voted at the Special Meeting. Alternatively, you may follow the instructions on the proxy card to submit your proxy by telephone or via the Internet.

Q. What does it mean if I get more than one proxy or vote instruction card?

A. If your shares are registered differently and are in more than one account, you will receive more than one proxy or (if your shares are held in "street name" by your broker or by another nominee) vote instruction card. Please complete and return all of the proxy and vote instruction cards you receive, or submit your proxy by telephone or the Internet, to ensure that all of your shares are voted.

The Asset Sale Proposal

Q. What is the proposed transaction?

A. The proposed transaction is the sale of substantially all the operating assets and assignment of certain liabilities of our clinical laboratory business to the Buyer, pursuant to an Asset Purchase Agreement, dated as of March 16, 2023, as such agreement may be amended from time to time.

Q. What is the Purchase Price of the Asset Sale?

A. The Purchase Price for the Asset Sale is \$146 million, subject to adjustment as described herein.

Q: What are the material terms of the Asset Purchase Agreement?

A. See the section of this proxy statement entitled "PROPOSAL 1 — TO APPROVE THE ASSET SALE — Principal Provisions of the Asset Purchase Agreement" on pages 41 through 45 of this proxy statement.

Q: What will happen if the Asset Sale is approved by shareholders?

A. If the holders of at least a majority of all of the outstanding shares of Common Stock approve the Asset Sale, and the conditions to closing the Asset Sale as set forth in the Asset Purchase Agreement are satisfied or waived, the Company and its subsidiaries will sell substantially all the operating assets and assign of certain liabilities of our clinical laboratory business to the Buyer for cash. Subject to the terms and conditions stated in the Asset Purchase Agreement, after shareholder approval of the Asset Sale, Buyer will be obligated to pay a fee of \$75,000 to the Company for each day after the date of such shareholder approval until the closing of the Asset Sale, provided that such fee shall not exceed an aggregate of \$7,000,000. At the closing, such fee will be wholly or partially credited against the purchase price. For more information, see the section below captioned "PROPOSAL I — TO APPROVE THE ASSET SALE — Principal Provisions of the Asset Purchase Agreement-Conditions to the Closing of the Asset Sale" beginning on page 42.

As soon as practicable after the closing of the Asset Sale, the Company plans to evaluate options in order to maximize shareholder value with the approval of the Board for a certain portion of proceeds to be distributed to the shareholders.

Q. Why is our Board recommending the Asset Sale?

A. Our Board believes that the Asset Sale and the Asset Purchase Agreement are fair to, advisable and in the best interests of, the Company and its shareholders. Our Board recommends that you vote "FOR" the approval of the Asset Sale Proposal. To review our Board's reasons for recommending the Asset Sale, see the section entitled "THE ASSET SALE — Reasons for the Asset Sale and Recommendation of Our Board of Directors" on pages 17 through 19 of this proxy statement.

O. Will the Asset Sale be a taxable transaction to me?

A. No. Our shareholders will not realize any gain or loss for U.S. federal income tax purposes as a result of the Asset Sale. See "MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE ASSET SALE" beginning on page 40.

Q. When is the Asset Sale expected to be completed?

A. We are working towards closing the Asset Sale as soon as possible, subject to the terms of the Asset Purchase Agreement. We currently expect to close the Asset Sale by the later of July 14, 2023 (which is 120 days after the signing of the Asset Purchase Agreement) or as soon as all of the conditions to the Asset Sale are satisfied or waived, including shareholder approval of the Asset Sale Proposal at the Special Meeting.

Q. Are there any risks related to the Asset Sale?

A. Yes. You should carefully read the section entitled "PROPOSAL 1 — TO APPROVE THE ASSET SALE — Risks Related to the Asset Sale" beginning on page 47.

Q. Are holders of the Company's common stock entitled to dissenters' rights?

A. Holders of Common Stock may be entitled to dissenters' rights under New York law in connection with the Asset Sale, provided such rights are properly perfected. Holders of Common Stock must comply with the requirements of the New York Business Corporation Law (the "NYBCL") in order to be entitled to have the "fair value" (as described in Section 623 of the NYBCL) of their shares of Common Stock determined by judicial proceedings and to receive payment based on that valuation. The ultimate amount a holder would receive in a dissenters' rights proceeding may be more than, the same as or less than the value of the holder's shares of Common Stock. To exercise dissenters' rights, holders of Common Stock must precisely comply with the requirements of the NYBCL. See "THE ASSET SALE — Dissenters' Rights" (page 35) and the text of the New York dissenters' rights statutes, Sections 623 and 910 of the NYBCL, which is reproduced in its entirety as Annex C to this proxy statement.

Q. What will happen if the shareholders do not approve the Asset Sale at the Special Meeting?

A. If our shareholders do not approve the Asset Sale at the Special Meeting, the Company and Buyer can terminate the Asset Purchase Agreement. If the Asset Purchase Agreement is terminated, the Company will pay to Buyer all of the reasonable and documented out-of-pocket expenses incurred by the Buyer in connection with the Asset Purchase Agreement and the transactions contemplated thereby up to a maximum of \$5,000,000. Our Board, in discharging its fiduciary obligations to our shareholders, will evaluate other strategic alternatives that may be available, which alternatives may not be as favorable to our shareholders as the Asset Sale and may include the bankruptcy and liquidation of the Company.

The Adjournment Proposal

Q. What is the effect of voting to approve the proposal to adjourn the Special Meeting, if necessary or appropriate, to solicit additional proxies?

A. If you vote in favor of the proposal to adjourn the Special Meeting, if necessary or appropriate, to solicit additional proxies, you will be voting to permit the Special Meeting to be adjourned, even if a quorum is present, in order to solicit additional votes in favor of the proposal to approve the Asset Sale. In the event that the proposal to approve the Asset Sale has not received the requisite shareholder approval, it is likely that our Board would seek to adjourn the Special Meeting to solicit additional proxies in favor of that proposal. The receipt of sufficient votes in favor of the Adjournment Proposal would allow our Board to take such action.

General

Q. What should I do if I have questions?

A. If you have more questions about the Special Meeting, the Asset Sale or this proxy statement, or would like additional copies of this proxy statement or the proxy card, you should contact:

Enzo Biochem, Inc.

81 Executive Blvd. Suite 3 Farmingdale, New York 11735 (631) 755-5500 Attn.: Investor Relations

or



Kingsdale Advisors

+1 (888) 518-1554 (toll-free in North America)

+1 (917) 341-5244 (call collect outside of North America)

e-mail: contactus@kingsdaleadvisors.com

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement, and the documents to which we refer you in this proxy statement, contain forwardlooking statements about our plans, objectives, expectations and intentions. Forward-looking statements include information concerning possible or assumed future results of operations of our company, the expected completion and timing of the Asset Sale and other information relating to the Asset Sale. You can identify these statements by words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "estimate," "may," and "continue" or similar words. You should read statements that contain these words carefully. They discuss our future expectations or state other forward-looking information, and may involve known and unknown risks over which we have no control, including, without limitation,

- the satisfaction of the conditions to consummate the Asset Sale, including the approval of the Asset Sale Proposal by our shareholders;
- the occurrence of any event, change or other circumstance that could give rise to the termination of the Asset Purchase Agreement;
- the outcome of any legal proceedings that may be instituted against us and others following the announcement of the Asset Purchase Agreement;
- the amount of the costs, fees, expenses and charges related to the Asset Sale;
- the effect of the announcement of the Asset Sale on our customer relationships, operating results and business generally, including the ability to retain key employees;
- the amount of cash we will hold after the Asset Sale or our net proceeds from the Asset Sale;

and other risks detailed in this proxy statement and our current filings with the SEC, including our most recent filings on Forms 10-Q and 10-K. See "WHERE YOU CAN FIND MORE INFORMATION" on page 55 of this proxy statement. You should not place undue reliance on forward-looking statements. We cannot guarantee any future results, levels of activity, performance or achievements. The statements made in this proxy statement represent our views as of the date of this proxy statement, and you should not assume that the statements made herein remain accurate as of any future date. Moreover, we assume no obligation to update forward-looking statements or update the reasons actual results could differ materially from those anticipated in forward-looking statements, except as required by law.

THE SPECIAL MEETING

We are furnishing this proxy statement to you, as a shareholder of the Company, as part of the solicitation of proxies by our Board for use at the Special Meeting of shareholder.

Date, Time, Place and Purpose of the Special Meeting

The Special Meeting will be held virtually over the internet using the link, www.virtualshareholdermeeting.com/ENZ2023SM, on May 22, 2023 at 9:00 a.m., EDT, for the following purposes:

- to consider and vote on a proposal to approve the Asset Sale and adopt the Asset Purchase Agreement (the "Asset Sale Proposal"); and
- (2) to consider and vote on a proposal to approve the adjournment of the Special Meeting, if necessary or appropriate, to solicit additional proxies if there are insufficient votes at the time of the Special Meeting to approve the Asset Sale Proposal (the "Adjournment Proposal").

Recommendation of Our Board of Directors

Our Board has determined that the Asset Purchase Agreement and the Asset Sale are fair to, advisable and in the best interests of, the Company and its shareholders. Accordingly, our Board has adopted and approved the Asset Purchase Agreement recommends that our shareholders vote (i) "FOR" the approval of the Asset Sale Proposal, and (ii) "FOR" the approval of the Adjournment Proposal.

Record Date; Shares Entitled to Vote; Quorum

The holders of record of shares of our common stock as of the close of business on Aprill 7, 2023, which is the record date for the Special Meeting, are entitled to receive notice of and to vote at the Special Meeting.

On the record date, there were 49,662,821 shares of our common stock outstanding held by approximately 742 shareholders of record. Holders of a majority of the votes of the shares of common stock of the Company issued and outstanding and entitled to vote shall be present in person or by proxy to constitute a quorum for the transaction of any business at the Special Meeting. Abstentions (but not broker "non-votes") will be counted as present for purposes of determining the existence of a quorum.

Vote Required

The approval of the Asset Sale Proposal requires the affirmative vote of a majority of shares of Common Stock outstanding and entitled to vote thereon. For the Adjournment Proposal, approval requires the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting.

Each holder of a share of our common stock is entitled to one vote per share.

Abstentions will have the same effect as a vote "AGAINST" the Asset Sale Proposal and will have no effect on the outcome of the vote with respect to the Adjournment Proposal.

Brokers or other nominees who hold shares of our common stock in "street name" for customers who are the beneficial owners of such shares may not give a proxy to vote those customers' shares in the absence of specific instructions from those customers. These non-voted shares of our common stock will not be counted as present for purposes of determining the presence or absence of a quorum at the Special Meeting, and will have the same effect as a vote "AGAINST" the Asset Sale Proposal. These non-voted shares will have no effect on the outcome of any vote on the Adjournment Proposal, assuming a quorum is present.

Voting

The Special Meeting will be a completely virtual meeting. There will be no physical meeting location. The Special Meeting will be conducted via live webcast. Shareholders will have the same rights and opportunities to participate in the virtual Special Meeting as they would at an in-person meeting.

To attend the virtual Special Meeting, visit www.virtualshareholdermeeting.com/ENZ2023SM and enter the 16-digit control number included on your proxy card or voting instruction form. The meeting will start at 9:00 a.m., EDT on May 22, 2023. We encourage you to access the virtual Special Meeting prior to the start time to familiarize yourself with the virtual platform and ensure you can hear the streaming audio. Online access will be available starting at 8:30 a.m., EDT on May 22, 2023.

The virtual Special Meeting platform is fully supported across browsers (Firefox, Chrome, and Safari) and devices (desktops, laptops, tablets and cell phones) running the most updated version of applicable software and plugins. Participants should ensure that they have a strong WiFi connection from wherever they intend to participate in the virtual Special Meeting.

While we strongly encourage you to vote your shares prior to the virtual Special Meeting, shareholders may also vote during the meeting. Once logged in, you will be able to vote your shares by clicking the "Vote Here!" button

Shareholders may submit written questions once logged into the virtual Special Meeting platform. Questions pertinent to meeting matters will be answered during the question and answer portion of the virtual Special Meeting, subject to a time limit prescribed by the Rules of Conduct that will be posted to the virtual Special Meeting platform on the day of the meeting. The Rules of Conduct will also provide additional information about the relevancy of questions to meeting matters.

If you encounter any difficulties accessing the virtual Special Meeting during the check-in or meeting time, you should call the technical support number that will be posted on the virtual shareholder meeting login page.

Householding of Proxy Materials

The SEC has adopted rules that permit companies and intermediaries (e.g., brokers) to satisfy the delivery requirements for proxy statements and annual reports with respect to two or more shareholders sharing the same address by delivering a single Notice or Proxy Statement addressed to those shareholders. This process, which is commonly referred to as "householding," potentially means extra convenience for shareholders and cost savings for companies.

Brokers with account holders who are shareholders may be "householding" our proxy materials. A single Notice or Proxy Statement may be delivered to multiple shareholders sharing an address unless contrary instructions have been received from the affected shareholders. Once you have received notice from your broker that it will be "householding" communications to your address, "householding" will continue until you are notified otherwise or until you notify your broker or the company that you no longer wish to participate in "householding."

If, at any time, you no longer wish to participate in "householding" and would prefer to receive a separate proxy statement and annual report, you may (1) notify your broker, (2) direct your written request to: Enzo Biochem, Inc., 81 Executive Blvd., Suite 3, Farmingdale, New York 11735, Attention: Secretary, or contact us at (631) 755-5500. Shareholders who currently receive multiple copies of the proxy statement at their address and would like to request "householding" of their communications should contact their broker. In addition, we will promptly deliver, upon written or oral request to the address or telephone number above, a separate copy of the annual report and proxy statement to a shareholder at a shared address to which a single copy of the documents was delivered.

Solicitation of Proxies

In addition to solicitation by mail, our directors, officers and employees may solicit proxies by telephone, other electronic means or in person. These people will not receive any additional compensation for their services, but we will reimburse them for their out-of-pocket expenses. We will reimburse banks, brokers, nominees, custodians and fiduciaries for their reasonable expenses in forwarding copies of this proxy statement to the beneficial owners of shares of our common stock and in obtaining voting instructions from those owners. We will pay the costs of this proxy solicitation, including all expenses of filing, printing and mailing this proxy statement. We have also retained Kingsdale Advisors, a proxy advisory and solicitation firm, to assist with the solicitation of proxies and provide related advice and informational support, for a services fee, plus customary disbursements, which are not expected to exceed \$20,000 in total.

Submitting Proxies via Internet or by Telephone

Whether or not you plan to attend the Special Meeting, and regardless of the number of shares of Common Stock you own, please vote as soon as possible. We urge you to date, sign and return the proxy card in the envelope provided to you, or to use the telephone or Internet method of voting described on your proxy card, even if you plan to attend the Special Meeting, so that if you are unable to attend the Special Meeting, your shares can be voted.

Revocability of Proxies

Any shareholder of record who submits a proxy card retains the right to revoke such proxy card by: (i) submitting a written notice of such revocation to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; (ii) submitting a duly signed proxy card bearing a later date than the previously signed and dated proxy card to the Chief Executive Officer of the Company so that it is received no later than 5:00 p.m., EDT on May 21, 2023; or (iii) attending the Special Meeting virtually and voting thereat the shares represented by such proxy card. Attendance at the Special Meeting will not, in and of itself, constitute revocation of a completed, signed and dated proxy card previously returned. All such later-dated proxy cards or written notices revoking a proxy card should be sent to Enzo Biochem, Inc., 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Hamid Erfanian, Chief Executive Officer. If you hold shares of Common Stock in street name, you must contact the firm that holds your shares to change or revoke any prior voting instructions.

Other Business

As of the date of this proxy statement, our Board does not intend to present at the Special Meeting of shareholders any matters other than those described herein and does not presently know of any matters that will be presented by other parties. If any other matter requiring a vote of the shareholders should come before the meeting, it is the intention of the persons named in the proxy to vote with respect to any such matter in accordance with the recommendation of our Board or, in the absence of such a recommendation, in accordance with the best judgment of the proxy holder.

THE ASSET SALE

This discussion of the Asset Sale is qualified by reference to the Asset Purchase Agreement, which is attached to this proxy statement as **Annex** A. You should read the entire Asset Purchase Agreement carefully as it is the legal document that governs the Asset Sale.

General

Our Board has adopted and approved an Asset Purchase Agreement with the Buyer whereby we will sell substantially all the operating assets and assign certain liabilities of our clinical laboratory business to the Buyer. If the Asset Sale is completed, we may be required to provide certain transition services to the Buyer for up to six months following the closing of the Asset Sale.

Background of the Proposed Sale

Enzo Biochem, Inc. (the "Company", "we", "our", or "Enzo") is an integrated clinical labs, diagnostics, and life sciences company that was formed in 1976. Our mission is to harness genetic processes in order to develop research tools, provide diagnostic services, and develop enabling technologies for detecting and identifying genes and modifying gene expression. We are focused on delivering and applying advanced technological capabilities to develop reliable products and services for the global community. Our proprietary technology solutions platform is comprised of access to 200,000 products, over 10,000 of which are produced by the Company. Throughout the course of our research and development activities, we have built a significant patent position, consisting of over 450 patents, both domestic and foreign. A pioneer of the industry, we were the first biotechnology company to go public with our initial public offering in June 1980.

Our activities primarily consist of our two operating segments: Enzo Clinical Labs ("ECL") and Enzo Life Sciences ("ELS"). ECL provides a broad range of clinical laboratory services to hospitals, physicians, researchers, and pharmaceutical companies and offers an extensive menu of clinical laboratory tests and procedures used by physicians to establish or support a diagnosis or monitor treatment and medication. ELS develops and manufactures products for clinical research and drug development customers. Underpinned by extensive technological capabilities, ELS has developed products used in the identification of genomic information by laboratories worldwide.

Between 2020 and early 2022, we pursued the potential sale of ECL through an auction run by Cain Brothers, our financial advisor during that process. That auction yielded strong interest from two bidders, one of whom was the Buyer. We initially granted exclusivity to the other bidder, but upon expiration of the exclusivity period with such bidder, we ceased negotiations and entered into exclusivity with the Buyer. By the beginning of 2022, we had been negotiating the purchase and sale of ECL with the Buyer for over six months, at which time we abandoned the potential transaction with the Buyer. We had determined that both bidders failed to put forth acceptable proposals.

As previously disclosed, since the fiscal year ended July 31, 2018, other than during the fiscal years ended July 31, 2021 and 2022, our ECL segment has suffered significant losses or nearly broke even. The exception to this came as a result of the COVID-19 pandemic, which began in March 2020. At the start of the pandemic, the Company made substantial investments to expand and maintain COVID-19 testing for the communities it serves. The Company's technical expertise in molecular diagnostics allowed it to develop next generation COVID-19 diagnostic and antibody testing options, which were approved under the FDA Emergency Use Authorization (EUA). During the fiscal year ended July 31, 2022, the Company generated substantial COVID-19-related services revenues, representing 44% of all services revenues. The testing revenues had a significantly positive impact on the profitability and cash flow of the ECL segment for most of the fiscal year ending July 31, 2022. Revenues from COVID-19 testing during the three and six months ended January 31, 2022 represented 56% and 52%, respectively of all services revenues.

However, the rate of transmission of COVID-19 and the severity of its variants have dramatically declined in the U.S. over the last year. Increases in vaccination rates and booster shots, the development of new therapeutics and greater availability of rapid COVID-19 tests has resulted in a continued, significant decline in demand for our COVID-19 testing. As a result, volume, revenues, profitability, and cash flow from COVID19 testing during the current periods were all substantially and materially lower than the prior year period levels. Revenues from COVID-19 testing during the three and six months ended January 31, 2023 represented approximately 4% and 6%.

respectively, of all of the Company's services revenues. Thus, COVID-19 testing is no longer a material part of our ECL business segment. The ECL business segment has subsequently returned to losses and our management and our board of directors expect that the ECL segment will continue to experience losses for the foreseeable future.

At a special meeting held on March 11, 2022, our Board authorized management to pursue various strategic options, including selecting and interviewing appropriate financial advisors to assist us with analyzing potential strategic sale transactions. Between March 11, 2022 and May 27, 2022, we contacted and interviewed six financial advisors that we believed had the requisite experience, particularly in our industry, to advise the Company on our capital needs and potential strategic sale transactions. Leading up to the selection of Jefferies LLC ("Jefferies") as our financial advisor, our Board met (through in-person or telephonic meetings) on each of March 22, 2022, April 4, 2022, April 7, 2022, April 27, 2022, May 9, 2022, and May 16, 2022, in each case to monitor, deliberate and advise our management on how to proceed in connection with the evaluation of potential strategic sale transactions, and considering alternatives such as raising additional capital and exploring potential asset sales.

On May 16, 2022, our Board proposed, and on July 12, 2022, established a special mergers and acquisitions committee (the "M&A committee") to oversee and advise management in connection with a potential sale of the ECL business segment. The M&A committee consisted of Mary Tagliaferri, Brad Radoff and Ian Walters and provided our Board with regular updates with respect to a potential sale transaction.

Our Board, in close consultation with our M&A committee, evaluated and discussed potential strategic alternatives for the Company and weighed the diversion of management's time and effort and expected transaction expenses and costs against the need and desire to enhance shareholder value given our current stock price and future prospects of our businesses including ECL and ELS. Our Board directed management, with the assistance of Jefferies, to focus its efforts on pursuing a potential sale transaction of the ECL business segment in a third-party solicitation process (the "Sale Process"), while also considering future plans with respect to our retained businesses. Between September 2022 and October 2022, our management, with the assistance of Jefferies, prepared confidential information materials (a "CIM") and identified potential strategic buyers for the ECL business segment. During this time, with the assistance of our advisors, we began preparing for a potential sale transaction and assembling a virtual data room containing due diligence information to be provided to potential counterparties in connection with the Sale Process.

Between October 2022 and December 2022, at the direction of our board of directors, Jefferies contacted a total of 105 potential buyers, consisting of 59 potential strategic parties and 46 financial sponsors that could be interested in acquiring the ECL and ELS business segments. Of the potential buyers, 42 parties executed a confidentiality agreement and received the CIM and a first round process letter in connection with the Sale Process, which directed the parties to submit initial indications of interest by December 14, 2022. During the first half of December 2022, management conducted executive summary presentations with 14 potential buyers and held due diligence calls as requested. At the conclusion of the first round, ten parties conveyed initial indications of interest, and our board of directors determined to progress five parties to the second round of the Sale Process, including the Buyer.

Between October 2022 and January 2023, our Board and M&A committee regularly held special meetings during which members of management and representatives of Jefferies provided updates on the Sale Process, including the number of potential buyers contacted in accordance with the directives of our board of directors and any developments following the meetings with potential buyers.

On January 12, 2023, at a meeting of our Board, management updated our board of directors with respect to its efforts in the Sale Process, and Jefferies updated our Board regarding the efforts specifically to sell assets of ECL, as directed and prioritized by our Board, including the potential buyers contacted. Management further updated our Board on the projections of revenue, expenses and cash of each of ECL and ELS.

Following this meeting, with instruction from, and in consultation with, the management team, our legal counsel, McDermott Will & Emery LLP ("McDermott"), began drafting the Asset Purchase Agreement contemplating the sale of the ECL business, which was later made available to potential buyers.

On or around January 15, 2023, a second round process letter was sent to the remaining potential buyers of the assets of ECL in which bidders were directed to submit comments to the draft Asset Purchase Agreement by February 9, 2023 and final bids by February 15, 2023.

As of mid-January 2023 and over the following several weeks, three potential buyers including the Buyer expressed continued interest in acquiring the assets of ECL. During this time, all three potential buyers had site visits and meetings with our management team to facilitate and expedite due diligence. During this time, McDermott negotiated clean team agreements with bidders to facilitate due diligence review in compliance with applicable competition laws.

On January 31, 2023, the Company received an issues list to our auction draft Asset Purchase Agreement from a potential buyer ("Party A"). Over the next few days McDermott and Party A's counsel discussed the issues list, in particular such Party A's termination rights and the minimum-volume closing condition, among other terms. Between February 9, 2023 and February 10, 2023, the Company received comments to the draft Asset Purchase Agreement from Party A and the Buyer. Over the next several days, representatives of McDermott discussed these comments with management, evaluating the critical legal and business issues raised by each bidder's draft of the Asset Purchase Agreement, which issues included (in addition to those noted above) the scope of assumed and excluded assets and liabilities, the survival of representations and warranties, and post-closing indemnification. Over the next few days, representatives of McDermott discussed these and other issues with each Party A's and Buyer's respective counsel. Jefferies also provided our board of directors with disclosure regarding certain material investment banking relationships during the prior two-year period with the three remaining bidders, including the Buyer with which no such material investment banking relationships were indicated.

On February 15, 2023, we received three final bids from the remaining interested buyers of the assets of ECL, including the Buyer. The Buyer and Party A submitted updated draft letters of intent and markups of the Asset Purchase Agreement (as discussed above) and a third bidder submitted only an updated draft letter of intent. Following execution of clean team agreements, we provided clean room data room access to selected personnel of each bidder. The clean room contents included fee schedules, an employee census and revenue reports.

On February 17, 2023, our Board held a meeting to consider developments in the Sale Process and discuss the terms of the transaction as proposed by each of the final bidders. At the meeting, management and representatives of McDermott outlined the terms of each bidder's proposed letter of intent and discussed Party A's and the Buyer's markups of the Asset Purchase Agreement (noted above). In particular, our counsel outlined key provisions of the draft Asset Purchase Agreement, including the issues described above, the expectation that the Buyer would have negotiated post-closing employment arrangements with certain key employees of ECL prior to signing the Asset Purchase Agreement, the limitation on Buyer's efforts to obtain HSR approval or to assume all required filing fees, a closing condition requiring the delivery of landlord consents, the Company's obligation to compensate the Buyer for expenses under certain circumstances, the scope of the Company's post-closing restrictive covenants and the possibility of post-closing transition services. After comprehensive evaluations of all critical aspects of the transactions proposed by each of the final bidders, our Board authorized management to enter into an exclusivity period as requested by the Buyer. It was noted that the Buyer's proposal was superior among the ECL bidders from a financial and deal certainty perspective.

Between February 25, 2023 and March 16, 2023, our management had a series of calls with the Buyer and its legal and financial advisors to discuss the potential sale transaction and related matters, including ongoing due diligence, the scope of the purchased assets and excluded liabilities, adjustments to the purchase price and relevant deal terms, regulatory considerations that would impact deal certainty, the timing and logistics around the transfer of customer accounts and employees and post-closing transition services. Based on contemporaneous discussions with the Buyer, our management repeatedly directed McDermott to make revisions to the Asset Purchase Agreement, in each case with the intent of enhancing shareholder value and achieve deal certainty. In tandem, McDermott actively engaged with the Buyer's legal counsel to negotiate legal terms and drafting points and exchanged successive drafts of the Asset Purchase Agreement during that time.

Between March 5, 2023 and March 10, 2023, McDermott and the Buyer's counsel discussed closing deliverables and post-closing covenants.

The Buyer's final bid required a two-week exclusivity period which expired on March 6, 2023. On March 6, 2023, as the parties continued to negotiate in good faith, we and Buyer agreed to extend the exclusivity agreement through March 9, 2023 and then again through March 16, 2023 to allow additional time to finalize the Asset Purchase Agreement.

Between March 9, 2023 and March 16, 2023, we and the Buyer exchanged multiple revised drafts of the Asset Purchase Agreement. During such time, we and McDermott continued to have numerous internal discussions and conversations with the Buyer and its counsel with respect to the remaining open points in the Asset Purchase Agreement, including the scope of purchased and excluded assets and liabilities, the scope and timing of our obligations contemplated by the Asset Purchase Agreement, the scope of fundamental representations, the scope of closing deliverables, our interim operating covenants, the treatment of certain intellectual property, and Buyer's intent to offer employment to certain of our employees.

Throughout the Sale Process, the M&A committee met regularly to discuss the status, progress and developments affecting the Sale Process and updated our board of directors contemporaneously (as discussed above) with respect to the same. These meetings were held on February 22, 2023, February 23, 2023, March 1, 2023, March 2, 2023, March 5, 2023, March 7, 2023, March 12, 2023, March 13, 2023, March 15, 2023, and March 16, 2023.

On March 14, 2023, our Board met with our management and legal and financial advisors to receive an update regarding, and to discuss, the proposed Asset Sale. In advance of this meeting, Jefferies provided its preliminary financial analysis of the proposed Purchase Price to our board of directors.

On March 15, 2023, at a special meeting of our Board held to discuss the latest developments with respect to the potential transaction with the Buyer, our board of directors again reviewed the overall Sale Process as conducted to date

Our management and representatives of McDermott then provided an overview of the negotiations with the Buyer and summarized key terms of the proposed Asset Purchase Agreement, including the resolution of outstanding open points thereunder, such as the breakup fee, the closing conditions and deliverables, and the survival of indemnification obligations. Representatives of McDermott and members of management updated our Board with respect to the expected timing and conditions to closing (and potential exposure and related risks), post-closing transition services and related matters.

Also at this meeting, Jefferies reviewed its financial analysis of the Purchase Price with the Board and rendered an oral opinion, confirmed by delivery of a written opinion dated March 15, 2023, to the Board to the effect that, as of that date and based on and subject to various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, the Purchase Price to be received for the Business in the Asset Sale pursuant to the Asset Purchase Agreement was fair, from a financial point of view, to the Company.

After carefully considering the various factors described under the section captioned "—Reasons for the Asset Sale and Recommendation of our Board of Directors," our Board then approved entry into the Asset Purchase Agreement by majority vote. Mr. Erfanian, Mr. Radoff, Dr. Walters and Dr. Tagliaferri voted in favor of approving our entry into the Asset Purchase Agreement, and Dr. Rabbani abstained.

On March 16, 2023, the Company and the Parent executed the Asset Purchase Agreement with the Buyer and issued a press release announcing the Asset Sale.

Reasons for the Asset Sale and Recommendation of Our Board of Directors

Our Board: (i) determined that the Asset Sale is fair, advisable and in the best interests of us and our shareholders (ii) approved the Asset Purchase Agreement and the Asset Sale, and (iii) recommends that our shareholders vote in favor of the approval of the Asset Sale and adoption of the Asset Purchase Agreement.

In evaluating the Asset Sale, our Board consulted with our management and legal and financial advisors and, in the course of reaching that determination and recommendation, our Board considered a number of potentially supportive factors in its deliberations including:

the fact that the Business is located in an attractive market in the tristate region of New York, New
Jersey and Connecticut and therefore was potentially of greater value to a third party than the value
implied by the Company's then-current stock price. It serves a number of physicians that find
services by the laboratory to be an asset and the capabilities of the laboratory are suitable to be
operated by a large reference laboratory;

- the fact that since the fiscal year ended July31, 2018, other than during the fiscal years ended July 31, 2021 and 2022, which were the peak COVID-19 periods, the Enzo Clinical Labs ("ECL") business suffered significant losses or nearly broke even and management and our Board expect to continue to experience losses for the foreseeable future;
- the determination by management and our Board, after evaluating various potential strategic
 alternatives and conducting an extensive review of our financial condition, results of operations and
 business prospects, that continuing to operate as a going concern or filing for bankruptcy protection
 was not reasonably likely to create greater value for our shareholders as compared to the value
 obtained pursuant to the Asset Sale primarily for the reasons stated below:
 - the substantial doubt about ECL's ability to continue as a going concern in the near term and
 insufficient cash resources available to continue funding the operations of ECL beyond the near
 future:
 - our need to obtain significant additional capital to finance ECL's operations, to make payments
 pursuant to agreements with our sales representatives, to keep current with healthcare-related
 regulatory requirements and to repay our debts and operational expenses; and
 - the belief by our Board, based on our sales and marketing efforts to date, that the pace of
 commercial adoption of our clinical lab solution and revenue derived from sale of our products
 was unlikely to increase sufficiently for us to achieve profitability for the foreseeable future, if
 at all.
- the fact that the consideration to be paid in the Asset Sale is all cash and we will retain our existing
 cash, cash equivalents and accounts receivable, which we believe provides significantly greater value
 to our shareholders than any other alternative available;
- management's belief that as a result of the extent of negotiations with the Buyer, we obtained the
 highest consideration that the Buyer was willing to pay or that we were likely to obtain from other
 potentially interested buyers in the Sale Process;
- the fact that after shareholder approval of the Asset Sale, Buyer will be obligated to pay a daily
 ticking fee of \$75,000 (with an aggregate ticking fee not to exceed \$7,000,000) to the Company for
 each day after the date of such shareholder approval until the closing of the Asset Sale, which ticking
 fee may be partially or wholly credited against the Purchase Price paid by the Buyer at the closing;
- Buyer's agreement to limit the survival period applicable to the representations and warranties, to limit the Company's liability for indemnification of breaches of certain representations and warranties, and to make certain indemnification claims subject to a Deductible;
- the results of the Sale Process, including early discussions and meetings with other potential thirdparty acquirors, and the fact that none of the bona fide proposals to acquire the ECL assets were superior to the Buyer's proposal;
- · the lack of a financing condition to the obligations of Buyer;
- the Asset Sale is subject to the approval of our shareholders;
- significant investment is required to be competitive in the market;
- the Company is undersized in comparison to a lab that can be profitable;
- the provisions in the Asset Purchase Agreement allowing our Board to withdraw its recommendation
 that our shareholders vote in favor of the Asset Sale if our board of directors receives a Superior
 Proposal (as defined in the Asset Purchase Agreement) subject to certain confidentiality and notice
 provisions;
- the provisions in the Asset Purchase Agreement allowing our Board to terminate the Asset Purchase Agreement in order to accept a Superior Proposal subject to certain conditions contained in the Asset Purchase Agreement and the payment to Buyer of a breakup fee of \$5 million; and

 the conclusion of our Board that such termination fees and transaction expenses were reasonable in light of the benefits of the Asset Sale and were at customary levels for termination fees and transaction expenses for comparable sized transactions.

In the course of its deliberations, our Board also considered a variety of risks and other countervailing factors, including:

- the conditions to closing that must be satisfied or waived, including obtaining from our lender consent to release certain liens and certain other third parties and acknowledgments that are outside of our control:
- the risks and costs to us if the Asset Sale is not consummated, including the diversion of management
 and employee attention, employee attrition and the effect of non-consummation of a transaction on
 customer, payor, licensor and vendor relationships;
- the general restrictions on our ability to solicit or engage in discussions or negotiations with a third
 party regarding specified transactions and the requirement that we pay the Buyer a breakup fee of
 \$5 million, if the Asset Purchase Agreement is terminated under certain circumstances;
- the fact that our shareholders will not participate in potential future growth and earnings, if any, as a
 result of potential increased commercial adoption of our clinical labs services and the intellectual
 property developed by us to date;
- the restrictions on the conduct of the Business prior to the consummation of the Asset Sale, requiring
 us to conduct the Business in accordance with our current practice, subject to specific limitations and
 exceptions;
- our potential obligations to provide transition services to the Buyer after consummation of the Asset Sale and the uncertainty regarding the costs we will incur to provide such transition services; and
- the interests of our officers and directors in the Asset Sale described below under "Interests of Certain Persons in the Asset Sale."

The foregoing discussion of the factors considered by our Board is not intended to be exhaustive, but does set forth material factors considered by our Board. Our Board reached the conclusion to approve and adopt the Asset Purchase Agreement in light of the various factors described above and other factors that each member of our Board felt were appropriate. In view of the wide variety of factors considered by our Board in connection with its evaluation of the Asset Sale and the complexity of these matters, our Board did not consider it practical, and did not attempt to quantify, rank or otherwise assign relative weights to the specific factors it considered in reaching its decision and did not undertake to make any specific determination as to whether any particular factor, or any aspect of any particular factor, was favorable or unfavorable to the ultimate determination of our Board. Rather, our board of directors made its recommendation based on the totality of information presented to it and the investigation conducted by it. In considering the factors discussed above, individual directors may have given different weights to different factors.

After evaluating these factors, our Board determined that the Asset Purchase Agreement was fair to and in the best interests of, our shareholders. Accordingly, our Board adopted and approved the Asset Purchase Agreement. Our Board recommends that you vote "FOR" the Asset Sale Proposal.

Opinion of the Company's Financial Advisor

The Company has retained Jefferies as financial advisor to the Company in connection with the proposed Asset Sale. In connection with this engagement, the Company requested that Jefferies evaluate the fairness, from a financial point of view, to the Company of the Purchase Price to be received for the Business in the Asset Sale pursuant to the Asset Purchase Agreement. At a meeting of the Board held on March 15, 2023, Jefferies rendered an oral opinion, confirmed by delivery of a written opinion dated March 15, 2023, to the Board to the effect that, as of such date and based on and subject to the various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, the Purchase Price to be received for the Business in the Asset Sale pursuant to the Asset Purchase Agreement was fair, from a financial point of view, to the Company.

The full text of Jefferies' opinion, which describes the various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, is attached as Annex B to this proxy statement and is incorporated herein by reference. Jefferies' opinion was provided for the use and benefit of the Board (in its capacity as such) in its evaluation of the Purchase Price from a financial point of view to the Company and did not address any other aspect of the Asset Sale or any other matter. Jefferies' opinion did not address the relative merits of the Asset Sale or other transactions contemplated thereby as compared to any alternative transaction or opportunity that might be available to the Company or the Seller or with respect to the Business, nor did it address the underlying business decision by the Company to engage in the Asset Sale. Jefferies' opinion does not constitute a recommendation as to how the Board, any securityholder or any other party should vote or act with respect to the Asset Sale or any other matter. The following summary is qualified in its entirety by reference to the full text of Jefferies' opinion.

In arriving at its opinion, Jefferies, among other things:

- reviewed a draft, dated March 15, 2023, of the Asset Purchase Agreement;
- reviewed certain publicly available information relating to the Business;
- reviewed certain internal financial and other information furnished to Jefferies by the management of
 the Company relating to the Business and its operations and prospects on a standalone basis,
 including certain historical financial and other information, and financial forecasts and estimates,
 provided to or discussed with Jefferies by the management of the Company;
- held discussions with members of the senior management of the Company regarding the Business and its operations and prospects and the other matters described in the second and third bullets immediately above;
- considered the discussions undertaken at the direction of the Company with selected third parties to solicit indications of interest in the possible acquisition of all or a portion of the Company;
- reviewed the implied trading multiples of certain publicly traded companies with operations that Jefferies deemed relevant in evaluating the Business;
- reviewed, to the extent publicly available, financial terms of certain transactions that Jefferies deemed relevant in evaluating the Asset Sale; and
- · conducted such other financial studies, analyses and investigations as Jefferies deemed appropriate.

In its review and analysis and in rendering its opinion, Jefferies assumed and relied upon, but did not assume any responsibility to independently investigate or verify, the accuracy and completeness of all financial and other information that was supplied or otherwise made available by or on behalf of the Company or that was publicly available to Jefferies (including, without limitation, the information described above) or otherwise reviewed by Jefferies. Jefferies relied on assurances of the management and other representatives of the Company that they were not aware of any facts or circumstances that would make such information incomplete, inaccurate or misleading. Jefferies did not make or obtain an independent evaluation or appraisal of or relating to any of the assets or liabilities (contingent, accrued, derivative, off-balance sheet or otherwise), nor did Jefferies conduct a physical inspection of any of the properties or facilities, of the Business or any entity, and Jefferies assumed no responsibility to obtain or conduct any such evaluations, appraisals or physical inspections. Jefferies also did not evaluate the solvency of the Business or any entity under any state or federal laws relating to bankruptcy, insolvency or similar matters. In addition, Jefferies' analyses and opinion did not consider any actual or potential arbitration, litigation, audits, claims or possible unasserted claims, investigations or other proceedings involving or affecting the Business or any entity.

Jefferies was advised that there were no audited standalone financial statements relating solely to the Business and assumed that if such financial statements were available, they would not reflect any information that would be meaningful in any respect to Jefferies' analyses or opinion. Jefferies also assumed that there were no material changes in the assets or liabilities associated with, or in the financial condition or prospects of, the Business since the dates on which financial and other information relating to the Business were last made available to Jefferies that would be meaningful in any respect to Jefferies' analyses or opinion. With respect to the financial forecasts and estimates provided to and reviewed by Jefferies, Jefferies noted that projecting future results of any company or

business is inherently subject to uncertainty. However, Jefferies was advised, and Jefferies assumed, that the financial forecasts and estimates relating to the Business that Jefferies was directed to utilize for purposes of its analyses and opinion were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of the Company as to, and were an appropriate basis upon which to evaluate, the future financial performance of the Business and the other matters covered thereby. Jefferies expressed no opinion as to any financial forecasts or estimates or the assumptions on which they were based.

Jefferies relied upon the assessments of the management of the Company as to, among other things, (i) the potential impact on the Business of market, competitive, seasonal, macroeconomic and other conditions, trends and developments in and prospects for, and governmental, regulatory and legislative matters relating to or otherwise affecting, the clinical lab services industry, including healthcare reform regulations and changes in healthcare reimbursement models and products, government payment and reimbursement systems and thirdparty payor policies, (ii) implications of the global COVID-19 pandemic on the Business, (iii) the ability to operate, secure leases and maintain sufficient patient service centers and other facilities for the operations of the Business, (iv) required certifications and licenses for, and the technology and other intellectual property utilized in, the Business and associated risks, including the validity and duration of such certifications, licenses, technology and other intellectual property, and (v) existing and future agreements and other arrangements involving, and ability to attract, retain and/or replace, key employees, customers, suppliers and other commercial relationships relating to the Business. Jefferies assumed that there would not be any developments with respect to any such matters, or any adjustments to or allocations of the Purchase Price, that would have an adverse effect on the Business or the Asset Sale or that otherwise would be meaningful in any respect to Jefferies' analyses or opinion. Jefferies also assumed that the full escrowed portion of the Purchase Price would be received by the Seller in connection with the Asset Sale.

Jefferies' opinion was based on economic, monetary, regulatory, market and other conditions existing, and which could be evaluated, as of the date of Jefferies' opinion. Jefferies expressly disclaimed any undertaking or obligation to advise any person of any change in any fact or matter affecting its opinion of which Jefferies becomes aware after the date of its opinion. As the Board was aware, the credit, financial and stock markets, and the industry in which the Business operates, have experienced and may continue to experience volatility and Jefferies expressed no view or opinion as to any potential effects of such volatility on the Business or the Asset Sale

Jefferies made no independent investigation of, and Jefferies expressed no view or opinion as to, any legal, regulatory, accounting or tax matters affecting or relating to the Business or the Asset Sale and Jefferies assumed the correctness in all respects meaningful to its analyses and opinion of all legal, regulatory, accounting and tax advice given to the Company and/or the Board, including, without limitation, with respect to changes in, or the impact of, accounting standards or tax and other laws, regulations and governmental and legislative policies affecting the Business or the Asset Sale and legal, regulatory, accounting and tax consequences to the Company or its securityholders or to the Seller of the terms of, and transactions contemplated by, the Asset Purchase Agreement and related documents. Jefferies assumed that the Asset Sale would be consummated in accordance with its terms without waiver, modification or amendment of any material term, condition or agreement and in compliance with all applicable laws, documents and other requirements and that, in the course of obtaining the necessary governmental, regulatory or third-party approvals, consents, waivers and releases for the Asset Sale or otherwise, including with respect to any divestitures or other requirements, no delay, limitation, restriction or condition would be imposed or occur that would have an adverse effect on the Business or the Asset Sale or that otherwise would be meaningful in any respect to Jefferies' analyses or opinion. Jefferies also assumed that appropriate indemnification and other arrangements or provisions had been or would be made in respect of liabilities to be assumed by the Buyer in connection with the Asset Sale and that neither the Company nor the Seller would directly or indirectly incur or retain any liabilities or other obligations that were contemplated to be excluded from the Asset Sale or otherwise. Jefferies further assumed that the final Asset Purchase Agreement, when signed by the parties thereto, would not differ from the draft reviewed by Jefferies in any respect meaningful to Jefferies' analyses or opinion.

Jefferies' opinion did not address the relative merits of the Asset Sale or other transactions contemplated thereby as compared to any alternative transaction or opportunity that might be available to the Company or the Seller or with respect to the Business, nor did it address the underlying business decision by the Company to engage in the Asset Sale or the terms of the Asset Purchase Agreement or related documents, including the form or structure of the Purchase Price or the Asset Sale, any adjustments to or allocations of the Purchase Price or any term, aspect or implication of any indemnification, escrow or ticking fee arrangements, any voting, transition services, intellectual

property or other ancillary agreements or any other agreement, arrangement or understanding entered into in connection with, or contemplated by or resulting from, the Asset Sale or otherwise. Jefferies' opinion was limited to the fairness, from a financial point of view and as of the date of such opinion, to the Company of the Purchase Price (to the extent expressly specified in such opinion). Jefferies was not asked to, and Jefferies' opinion did not, address the fairness, financial or otherwise, of any consideration to the holders of any class of securities, creditors or other constituencies of the Company or any other party. Furthermore, Jefferies did not express any view or opinion as to the fairness, financial or otherwise, of the amount or nature of any compensation or other consideration payable to or to be received by any officers, directors or employees, or any class of such persons, in connection with the Asset Sale relative to the Purchase Price or otherwise. Jefferies also expressed no view or opinion as to the prices at which the securities of the Company may trade or otherwise be transferable at any time, including following announcement or consummation of the Asset Sale. The issuance of Jefferies' opinion was authorized by the Fairness Committee of Jefferies LLC.

In connection with rendering its opinion to the Board, Jefferies performed a variety of financial and comparative analyses, including those described below. The following summary is not a complete description of all analyses performed and factors considered by Jefferies in connection with its opinion. The preparation of a financial opinion is a complex process involving subjective judgments and is not necessarily susceptible to partial analysis or summary description. With respect to the selected public companies and selected precedent transactions analyses summarized below, no company or transaction used as a comparison was identical or directly comparable to the Business or the Asset Sale. These analyses necessarily involved complex considerations and judgments concerning financial and operating characteristics and other factors that could affect the public trading, acquisition or other values of the companies or transactions concerned.

Jefferies believes that its analyses and the summary below must be considered as a whole and in context and that selecting portions of its analyses and factors or focusing on information presented in tabular format, without considering all analyses and factors or the narrative description of the analyses, could create a misleading or incomplete view of the processes underlying Jefferies' analyses and opinion. Jefferies did not draw, in isolation, conclusions from or with regard to any one factor or method of analysis for purposes of its opinion, but rather arrived at its ultimate opinion based on the results of all analyses undertaken by it and assessed as a whole.

The estimates of the future performance of the Business in or underlying Jefferies' analyses are not necessarily indicative of future results or values, which may be significantly more or less favorable than those estimates. In performing its analyses, Jefferies considered industry performance, general business and economic conditions and other matters, many of which were beyond the control of the Company. Estimates of the financial value of companies do not purport to be appraisals or necessarily reflect the prices at which companies or securities actually may be sold or acquired. Accordingly, the estimates used in, and the implied reference ranges resulting from, any particular analysis described below are inherently subject to substantial uncertainty and should not be taken as Jefferies' view of the actual value of the Business or its assets.

The Purchase Price was determined through negotiations between the Company and the Buyer, and the decision of the Company to enter into the Asset Purchase Agreement was solely that of the Board. Jefferies' opinion and financial analyses were only one of many factors considered by the Board in its evaluation of the Purchase Price and should not be viewed as determinative of the views of the Board or the Company's management with respect to the Asset Sale or the consideration payable in the Asset Sale.

Financial Analyses

The summary of the financial analyses described in this section entitled "— Financial Analyses" is a summary of the material financial analyses reviewed with the Board and performed by Jefferies in connection with its opinion. The financial analyses summarized below include information presented in tabular format. In order to fully understand Jefferies' financial analyses, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses. Considering the data below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a misleading or incomplete view of Jefferies' financial analyses. Except as otherwise noted, financial data utilized for the Business in the financial analyses described below was based on financial forecasts and estimates relating to the Business and its operations and prospects on a standalone basis provided to or discussed with Jefferies by the management of the Company.

<u>Selected Public Companies Analysis.</u> Jefferies reviewed publicly available financial, stock market and operating information relating to the Business and the following seven selected companies that Jefferies viewed as generally relevant for purposes of analysis as publicly traded companies with operations in the clinical lab services industry (collectively referred to in this section as the "selected companies"):

- Diagnósticos da América S.A.
- Fleury SA
- · Laboratory Corporation of America Holdings
- Medicover AB
- Quest Diagnostics Incorporated
- · Sonic Healthcare Limited
- SYNLAB AG

Jefferies reviewed, among other information, enterprise values, calculated as fully-diluted equity values based on closing stock prices on March 14, 2023 (or, in the case of SYNLAB AG, its unaffected closing stock price on March 10, 2023, the last trading day prior to public announcement of the receipt by SYNLAB AG of a non-binding acquisition proposal) plus total debt, non-controlling interests and preferred equity (as applicable) and less cash and cash equivalents and short-term investments, as a multiple of calendar year 2023 and calendar year 2024 estimated revenue. Financial data of the selected companies were based on publicly available Wall Street research analysts' estimates, public filings and other publicly available information. Financial data of the Business was based on the Company management's forecasts and other estimates, calendarized for comparative purposes.

The overall low to high calendar year 2023 and calendar year 2024 estimated revenue multiples observed for the selected companies were 0.9x to 2.1x (with a median of 1.6x) and 0.8x to 2.1x (with a median of 1.6x), respectively. Jefferies then applied selected ranges of calendar year 2023 and calendar year 2024 estimated revenue multiples derived from the selected companies of 1.1x to 2.1x and 1.1x to 2.1x, respectively, to corresponding data of the Business.

This analysis indicated the following approximate implied enterprise value reference ranges for the Business, as compared to the Purchase Price:

Implied Enterprise Value R	Purchase Price	
CY2023E Revenue		
\$65.6 million – \$125.3 million	\$82.2 million – \$157.0 million	\$146 million

<u>Selected Precedent Transactions Analysis</u>. Using publicly available information, Jefferies reviewed financial data relating to the following 11 selected transactions that Jefferies viewed as generally relevant for purposes of analysis as transactions involving target companies with operations in the clinical lab services industry (collectively referred to in this section as the "selected transactions"):

Announced	Acquiror	Target
April 2022	Fulgent Genetics, Inc.	Inform Diagnostics, Inc.
December 2021	Sonic Healthcare Limited	 ProPath Services, LLC
December 2018	Sonic Healthcare Limited	 Aurora Diagnostics, LLC
May 2018	Myriad Genetics, Inc.	Counsyl, Inc.
September 2017	 Avista Capital Partners, L.P. 	• Miraca Life Sciences, Inc.
July 2016	Laboratory Corporation of America Holdings	• Sequenom, Inc.
June 2015	Cinven Partners LLP	 SYNLAB Group
June 2015	Eurofins Scientific SE	Biomnis Group
June 2015	OPKO Health, Inc.	• Bio-Reference Laboratories, Inc.
June 2015	Eurofins Scientific SE	• Diatherix Laboratories, Inc.
May 2015	Cinven Partners LLP	• Labco S.A.

Jefferies reviewed, among other information, transaction values of the selected transactions, calculated as the enterprise values implied for the target companies involved in the selected transactions based on the consideration paid or payable in the selected transactions, as a multiple of the latest 12 months revenue of the target companies (excluding COVID-19 related revenue, as applicable, to the extent publicly available) as of the announcement date of the applicable selected transaction. Financial data of the selected transactions were based on publicly available Wall Street research analysts' estimates, public filings and other publicly available information. Financial data of the Business was based on the Company management's forecasts and estimates, calendarized for comparative purposes.

The overall low to high latest 12 months revenue multiples observed for the selected transactions was 0.7x to 3.2x (with a median of 1.7x). Jefferies then applied a selected range of latest 12 months revenue multiples derived from the selected transactions of 1.2x to 2.2x to the latest 12 months (as of March 31, 2023) estimated non-COVID-19 related revenue of the Business.

This analysis indicated the following approximate implied enterprise value reference range for the Business, as compared to the Purchase Price:

Implied Enterprise Value Reference Range		Purchase Price
	\$53.2 million – \$97.6 million	\$146 million

Discounted Cash Flow Analysis. Jefferies performed a discounted cash flow analysis of the Business by calculating the estimated present value of the standalone unlevered, after-tax free cash flows that the Business was forecasted to generate during the fiscal years ending July 31, 2023 through July 31, 2027 based on the Company management's forecasts and other estimates. Implied terminal values for the Business were calculated by applying to the fiscal year 2027 estimated revenue of the Business a selected range of revenue multiples of 1.2x to 2.2x. The present values (as of March 31, 2023) of the cash flows and terminal values were then calculated using a selected range of discount rates of 12.5% to 13.5%. This analysis indicated the following approximate implied enterprise value reference range for the Business, as compared to the Purchase Price:

Implied Enterprise Value Reference Range	Purchase Price
\$42.6 million – \$112.8 million	\$146 million

Miscellaneous

The Company has agreed to pay Jefferies for its financial advisory services in connection with the proposed Asset Sale an aggregate fee of approximately \$3.0 million, of which a portion was payable upon delivery of Jefferies' opinion to the Board and approximately \$2.3 million is payable contingent upon consummation of the Asset Sale. In addition, the Company has agreed to reimburse Jefferies for Jefferies' expenses, including fees and expenses of counsel, incurred in connection with Jefferies' engagement and to indemnify Jefferies and related parties against certain liabilities, including liabilities under federal securities laws, arising out of or in connection with the services rendered and to be rendered by Jefferies under its engagement.

As the Board was aware, Jefferies also currently is acting as financial advisor to the Company in connection with its strategic review and related matters, for which services Jefferies expects to receive an aggregate fee of approximately \$2.0 million from the Company, and Jefferies and its affiliates in the future may provide other financial advisory services or financing services to the Company and/or its affiliates, for which services Jefferies and/or its affiliates would expect to receive compensation. As the Board also was aware, although Jefferies and its affiliates had not provided financial advisory or financing services to the Buyer during the two-year period prior to the date of Jefferies' opinion for which Jefferies and its affiliates received compensation, Jefferies and its affiliates may in the future provide such services to the Buyer and/or its affiliates, for which services Jefferies and its affiliates would expect to receive compensation. In the ordinary course of business, Jefferies and its affiliates may trade or hold securities or financial instruments (including loans and other obligations) of the Company, the Buyer and/or their respective affiliates for Jefferies' own account and for the accounts of Jefferies' customers and, accordingly, may at any time hold long or short positions or otherwise effect transactions in those securities or financial instruments.

Jefferies was selected as the Company's financial advisor in connection with the Asset Sale because, among other things, Jefferies is an internationally recognized investment banking firm with substantial experience in merger and acquisition transactions and based on its familiarity with the Company and its businesses and industry. Jefferies is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, leveraged buyouts, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities and private placements.

Past Contacts, Transactions or Negotiations

The Company has two pre-existing commercial relationships with the Buyer. First, the Buyer is one of our frequently used reference labs. Second, on October 17, 2022, we entered into a STAT Services Agreement (as amended on March 16, 2023) with the Buyer pursuant to which we perform certain testing services for the Buyer. Other than as described herein or under "Background of the Proposed Sale" above, neither the Company nor any of its subsidiaries has had any negotiations, transactions or material contacts during the past two years with the Buyer, and other than as described under "Background of the Proposed Sale" above or set forth in the Asset Purchase Agreement (and the other agreements contemplated thereby), there are no present or proposed material agreements, arrangements, understandings or relationships between the Company's executive officers, directors or affiliates, on the one hand, and any of the Buyer's executive officers, directors or affiliates, on the other hand.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

(in thousands, except share and per share data)

The following unaudited pro forma condensed consolidated financial information is based upon the historical consolidated statements of Enzo Biochem, Inc., a New York corporation (the "Company"), adjusted to give effect to the sale (the "Asset Sale") of substantially all the operating assets and assign certain liabilities of the Company's clinical laboratory business (the "Business") in accordance with the Asset Purchase Agreement dated March 16, 2023 (the "Asset Purchase Agreement") between the Company, Enzo Clinical Labs, Inc., a New York corporation (the "Seller"), and Laboratory Corporation of America Holdings, a Delaware corporation (the "Buyer"). These unaudited pro forma condensed consolidated financial statements are derived from, and should be read in conjunction with the Company's Annual Report on Form 10-K for the year ended July 31, 2022 filed with the United States Securities and Exchange Commission (the "SEC") on October 14, 2022 as amended on November 24, 2022, and the Quarterly Report on Form 10-Q for the interim period ended January 31, 2023, filed with the SEC on March 20, 2023.

The unaudited pro forma condensed consolidated balance sheet gives effect to the proposed Asset Sale as if it had occurred on January 31, 2023. The estimated cash proceeds and impact of the resulting gain are only included in the January 31, 2023 unaudited pro forma condensed consolidated balance sheet. The unaudited pro forma condensed consolidated statements of operations for the six months ended January 31, 2023 and the years ended July 31, 2022 and 2021 give effect to the proposed Asset Sale as if it had occurred on August 1, 2020.

The pro forma adjustments related to the sale of the Business are based on available information and assumptions that management believes are (1) directly attributable to the Asset Sale; (2) factually supportable; and (3) with respect to the unaudited pro forma condensed consolidated statements of operations, expected to have a continuing impact on consolidated operating results. Certain of the most significant assumptions are set forth under the Notes to Unaudited Pro Forma Condensed Consolidated Financial Information.

We have included the following unaudited pro forma condensed consolidated financial information solely for the purpose of providing shareholders with information that may be useful for purposes of considering and evaluating the proposal to approve the Asset Sale. The unaudited pro forma condensed consolidated financial information is not necessarily indicative of the results of operations or financial position that might have been achieved for the dates or periods indicated, nor is it indicative of the results of operations or financial position that may occur in the future. Furthermore, the pro forma adjustments may differ from those that will be calculated for purposes of reporting discontinued operations in future filings. The Company cautions shareholders that its future results of operations, including uses of cash and financial position, may significantly differ from those described in these unaudited pro forma condensed consolidated financial statements, and accordingly, these unaudited pro forma condensed consolidated financial statements should be read in conjunction with the disclosures in the proxy statement to which they are attached regarding the nature of the Company's business following completion of the transactions contemplated by the Asset Purchase Agreement.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET As of January 31, 2023 (in thousands, except share and per share data)

	Н	listorical		Pro Forma djustments		I	ro Forma
ASSETS							
Current assets:							
Cash and cash equivalents		5,054		136,648	(a)		141,702
Accounts receivable, net		10,866		_			10,866
Inventories, net		15,723		(5,766)	(b)		9,957
Prepaid expenses and other current assets		4,738		(2,541)	(b)		2,197
Total current assets		36,381		128,341		_	164,722
Property, plant, and equipment, net		17,425		(5,136)	(b)		12,289
Right-of-use assets, net		13,569		(712)	(b)		12,857
Goodwill		7,452		(7,452)	(b)		_
Other, including restricted cash of \$1,000		1,625		(319)	(b)		1,306
Total assets	\$	76,452	\$	114,722		\$	191,174
	_		_			_	
LIABILITIES AND STOCKHOLDERS' EQUITY							
Current liabilities:							
Accounts payable – trade	\$	13,189	\$	_		\$	13,189
Accrued liabilities		11,590		17,353	(c)		28,943
Current portion of operating lease liabilities		3,425		(214)	(b)		3,211
Mortgage debt		3,684		_			3,684
Other current liabilities and finance leases short term		313		_			313
Total current liabilities		32,201		17,139			49,340
Operating lease liabilities, non-current, net		11,132		(498)	(b)		10,634
Long term debt, net		289		_			289
Total liabilities	\$	43,622	\$	16,641		\$	60,263
Stockholders' equity:							
Common Stock, \$.01 par value; authorized 75,000,000 shares; 48,733,054 shares issued and outstanding		487		_			487
Additional paid-in capital		340,407		_			340,407
Accumulated deficit	(310,593)		98,081	(c)		(212,512)
Accumulated other comprehensive income		2,529		_			2,529
Total stockholders' equity		32,830		98,081		_	130,911
Total liabilities and stockholders' equity	\$	76,452	\$	114,722		\$	191,174

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS For the six months ended January 31, 2023 (in thousands, except share and per share data)

]	Pro Forma Historical Adjustments			Pro Forma	
Revenues	\$	34,614	\$ (19,997)		(d)	\$ 14,617
Operating costs and expenses:						
Cost of revenues		29,750		(20,545)	(d)	9,205
Research and development		2,427		(603)	(d)	1,824
Selling, general and administrative		23,263		(13,493)	(d)	9,770
Legal and related expense, net		2,066		(171)	(d)	 1,895
Total operating costs and expenses		57,506		(34,812)		22,694
				_		
Operating loss (income)		(22,892)		14,815		(8,077)
Other income (expense):						
Interest, net		132		3	(d)	135
Other		130		(12)	(d)	118
Foreign exchange gain		675		_		675
Total other income (expense)		937		(9)		928
(Loss) income before income taxes		(21,955)		14,806		(7,149)
Income taxes		_		_		_
Net (loss) income	\$	(21,955)	\$	14,806		\$ (7,149)
Net loss per common share:						
Basic and diluted	\$	(0.45)				\$ (0.15)
	_					
Weighted average common shares outstanding:						
Basic and diluted		48,725				48,725

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS For the year ended July 31, 2022 (in thousands, except share and per share data)

	Historical	Pro Forma Adjustments		Pro Forma
Revenues	\$ 107,071	\$ (74,428)	(d)	\$ 32,643
Operating costs and expenses:				
Cost of revenues	65 104	(45.001)	(L)	10.212
	65,104	(45,891)	(d)	19,213
Research and development	3,767	(1,329)	(d)	2,438
Selling, general and administrative	48,018	(26,173)	(d)	21,845
Legal and related expense, net	5,689	(254)	(d)	5,435
Legal settlements	(500)			(500)
Total operating costs and expenses	122,078	(73,647)		48,431
Operating loss	(15,007)	(781)		(15,788)
Other income (expense):				
Interest, net	159	19	(d)	178
Other	(1,191)	(77)	(d)	(1,268)
Foreign exchange loss	(2,222)	_		(2,222)
Total other expense	(3,254)	(58)		(3,312)
Loss before income taxes	(18,261)	(839)		(19,100)
Income taxes	_	_		_
Net loss	\$ (18,261)	\$ (839)		\$ (19,100)
Net loss per common share:				
Basic and diluted	\$ (0.38)			\$ (0.39)
Weighted average common shares outstanding:				
Basic and diluted	48,594			48,594

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS For the year ended July 31, 2021 (in thousands, except share and per share data)

	Histor	ical	Pro Forma Adjustments		Pro Forma
Revenues	\$ 117	7,731	\$ (86,984)	(d)	\$ 30,747
Operating costs and expenses:					
Cost of revenues	64	1,154	(48,179)	(d)	15,975
Research and development		3,252	(615)	/	2,637
Selling, general and administrative	44	1,905	(26,028)	(d)	18,877
Legal and related expense, net		1,728	(264)	(d)	4,464
Total operating costs and expenses	117	7,039	(75,086)		41,953
Operating income (loss)		692	(11,898)		(11,206)
Other income (expense):					
Interest, net		8	17	(d)	25
Other	(5,905	18	(d)	6,923
Foreign exchange (loss) gain		270	_		270
Total other income		7,183	35		7,218
Income (loss) before income taxes	5	7,875	(11,863)		(3,988)
Income taxes		_			
Net income (loss)	\$ 7	7,875	\$ (11,863)		\$ (3,988)
Net income (loss) per common share:					
Basic	\$	0.16			\$ (0.08)
Diluted	\$	0.16			\$ (0.08)
Weighted average common shares outstanding:					
Basic	48	3,191			48,191
Diluted	48	3,325			48,191

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

(in thousands, except share and per share data)

1. Planned Asset Sale

On March 16, 2023, Enzo Biochem, Inc., a New York corporation (the "Company"), Enzo Clinical Labs, Inc., a New York corporation (the "Seller") entered into an Asset Purchase Agreement (the "Asset Purchase Agreement") with Laboratory Corporation of America Holdings, a Delaware corporation (the "Buyer"). Pursuant to the Asset Purchase Agreement, the Seller has agreed to sell substantially all the operating assets and assign certain liabilities of its clinical laboratory business (the "Business") to the Buyer in exchange for approximately \$146.0 million in cash, on and subject to the terms and conditions set forth therein (such transaction, the "Asset Sale"). The purchase price remains subject to adjustment pursuant to the terms of the Asset Purchase Agreement.

2. Unaudited Pro Forma Adjustments

The preparation of the unaudited pro forma consolidated financial information is based on financial statements prepared in accordance with accounting principles generally accepted in the United States. The pro forma adjustments reflected in the accompanying unaudited pro forma consolidated financial information reflect estimates and assumptions that we believe to be directly attributable to the Asset Sale, as further described in this proxy statement, and factually supportable. Actual results may differ from those estimates.

The unaudited pro forma consolidated financial information is provided for illustrative purposes only and does not purport to represent what the actual results of operations would have been had the Asset Sale occurred on the respective dates assumed, nor is it necessarily indicative of our future operating results. The unaudited pro forma consolidated financial information and the accompanying unaudited notes should be read in conjunction with our consolidated financial statements and notes thereto included by reference in this proxy statement.

The pro forma adjustments are based on the Company's preliminary estimates and assumptions which are subject to change. The following pro forma adjustments, related to the proposed Asset Sale, are included in the unaudited pro forma condensed consolidated balance sheet and/or the unaudited pro forma condensed consolidated statements of operations.

(a) Represents cash proceeds under the terms of the Asset Purchase Agreement of \$146.0 million (subject to certain adjustments and assuming full release of the \$5.0 million placed into an escrow account for the purpose of satisfying claims, if any, made by Buyer pursuant to the indemnification provisions included in the Asset Purchase Agreement), net of transaction-related expenses of \$9.4 million. Estimated transaction-related expenses are summarized as follows (amounts in thousands):

Banker success fees and expenses	\$ 3,075
Legal and accounting fees	2,700
Contingent transaction bonuses	 3,577
	\$ 9,352

- (b) Represents transfer of assets to the Buyer and the assumption of liabilities by the Buyer under the terms of the Asset Purchase Agreement.
- (c) The overall adjustment to accumulated deficit represents the estimated after-tax gain on the sale of the assets of \$98.1 million, which is calculated as follows (amounts in thousands):

Cash purchase price	\$ 146,000
Less estimated transaction-related expenses	9,352
Net proceeds	136,648
Less net assets sold under the Asset Purchase Agreement	21,214
Less estimated income taxes	17,353
After-tax gain on sale of assets	\$ 98,081

The actual after-tax gain will be a different amount. Management expects to utilize substantially all of the Company's historical tax attributes to offset a substantial portion of its taxable income arising from the Asset Sale. The estimated income taxes resulting from the Asset Sale are reflected as an adjustment to the accrued liabilities in the unaudited pro forma condensed consolidated balance sheet; however, the estimated after-tax gain on the Asset Sale has not been reflected in the unaudited pro forma condensed consolidated statements of operations as it is considered to be nonrecurring in nature.

(d) Represents the elimination of Revenue, Cost of Revenue, Research and Development, Selling, General and Administrative, Legal Fee Expense, and Other Income related to the Business for the periods presented. These adjustments have no effect on the Company's income taxes for all periods presented based on the Company being in a loss position with a full valuation allowance.

Certain Effects of the Asset Sale

If the shareholders approve the Asset Purchase Agreement and the closing occurs the, we may be requested by the Buyer to continue certain activities during the Transition Period. Services are to be proposed by the Buyer and agreed upon by the parties. In addition, for a period of five years after the closing of the Asset Sale, we agreed not to engage in any business activity that competes with the Business as conducted on the signing or closing date in New York, New Jersey and Connecticut, use names licensed to Buyer in any business that competes with the Business, or use certain trade names or certain other confusingly similar trade names in connection with any business of any kind.

Effects on the Company if the Asset Sale is Not Completed

The Asset Purchase Agreement includes customary termination provisions for both the Company and Buyer and provides that, in connection with the termination of the Asset Purchase Agreement under specified circumstances, including termination by the Company to accept and enter into a definitive agreement with respect to an unsolicited superior proposal, the Company will be required to pay Buyer a termination fee of \$5,000,000, or reimbursement of Buyer's expenses of up to \$5,000,000. If the Asset Sale is not completed, our Board, in discharging its fiduciary obligations to our shareholders, will evaluate other strategic alternatives that may be available, which alternatives may not be as favorable to our shareholders as the Asset Sale and may include a bankruptcy and liquidation of the Company.

Interests of Certain Persons in the Asset Sale

In considering the recommendation of our Board with respect to the Asset Purchase Agreement, holders of shares of our common stock should be aware that our executive officers and directors may have interests in the Asset Sale that may be different from, or in addition to, those of our shareholders generally. These interests may create potential conflicts of interest. Our board was aware of these potential conflicts of interest and considered them, among other matters, in reaching its decision to approve the Asset Purchase Agreement and to recommend that our shareholders vote in favor of the Asset Sale Proposal.

Stockholdings

The following table shows the aggregate number of shares of our common stock held, as of April 5, 2023, by each person who served as a director or executive officer of us at any time since the beginning of our last fiscal year

Title of Class	Name and Address of Beneficial Owner	Amount and Nature of Beneficial Ownership ⁽¹⁾	Percent of Class ⁽²⁾
Common Stock	Elazar Rabbani, Ph.D.	2,273,336(3)	4.5%
Common Stock	Hamid Erfanian	345,147 ⁽⁴⁾	*
Common Stock	Kara Cannon	285,928(5)	*
Common Stock	Patricia Eckert	36,067(6)	*
Common Stock	David Bench	10,138 ⁽⁷⁾	*
Common Stock	Bradley L. Radoff	4,663,726(8)	9.2%
Common Stock	Mary Tagliaferri, M.D.	96,633 (9)	*
Common Stock	Ian B. Walters, M.D.	96,633(10)	*
Common Stock	All directors, executive officers and key employees as a group (8 persons)	7,807,606(11)	15.4%

^{*} Represents beneficial ownership of less than 1%.

- (1) Except as otherwise noted in the footnotes to the table, all shares of Common Stock are beneficially owned and the sole investment and voting power is held by the persons named, and such persons' address is c/o Enzo Biochem, Inc., 81 Executive Blvd., Suite 3 Farmingdale, NY 11735.
- (2) For directors, executive officers and key employees, based upon 49,662,821 shares of Common Stock of the Company considered outstanding as of the close of business on April 5, 2023. Common Stock not outstanding but deemed beneficially owned by virtue of the right of an individual to acquire shares within 60 days from April 5, 2023 is treated as outstanding when determining the amount and percentage of Common Stock owned by directors and executive officers individually and as a group.
- (3) Includes (i) 348,100 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, (ii) 5,308 shares of Common Stock held in the name of Dr. Rabbani as custodian for certain of his children, (iii) 18,794 shares of Common Stock held in the name of Dr. Rabbani's wife as custodian for certain of their children, and (iv) 4,902 shares of Common Stock held in the Company's 401(k) plan.
- (4) Includes (i) 233,333 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023.
- (5) Includes (i) 237,667 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 35,661 shares of Common Stock held in the Company's 401(k) plan.
- (6) Includes (i) 16,667 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,400 shares of Common Stock held in the Company's 401(k) plan.
- (7) Mr. Bench resigned effective November 14, 2022.
- (8) Includes (i) 430,000 shares of Common Stock held by The Radoff Family Foundation of which Mr. Radoff is deemed to be the beneficial owner, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award. The address of Bradley L. Radoff is 2727 Kirby Drive Unit 29L Houston. Texas 77098.
- (9) Includes (i) 77,101 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award.
- (10) Includes (i) 77,101 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award.
- (11) Includes (i) 989,969 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 58,596 shares of Common Stock which are issuable within 60 days from April 5, 2023 under restricted stock unit awards.

Bonus Compensation

Summary Compensation Table

The following table sets forth summary information concerning compensation which will be paid to our named executive officers upon the closing of the Asset Purchase Agreement.

Name and Principal Position	Bonus ⁽¹⁾	Total	
Hamid Erfanian	\$ 2,482,000	\$ 2,482,0	000
Chief Executive Officer			
Kara Cannon	\$ 1,095,000	\$ 1,095,0	000
Chief Operating Officer			

Represents transaction bonus earned on closing of the Asset Purchase Agreement, as per terms of employment contract.

Former executive and current director Elazar Rabbani, Ph.D., and former executive David Bench will not receive any special compensation benefits upon the closing of the Asset Purchase Agreement.

Current independent directors Radoff, Tagliaferri, and Walters will not receive any special compensation benefits upon the closing of the Asset Purchase Agreement.

Indemnification of Officers and Directors

We have entered into indemnification agreements with each of our current directors and executive officers. These agreements require us to indemnify these individuals to the fullest extent permitted under New York law against liabilities that may arise by reason of their service to us, and to advance expenses incurred as a result of any proceeding against them as to which they could be indemnified.

Regulatory Matters

General

The Company and the Buyer have agreed to promptly as possible use their respective commercially reasonable efforts to obtain, or cause to be obtained, all consents and orders from all applicable governmental authorities and make, or cause to be made, all filings required by any applicable law that may be or become necessary for execution and delivery of the Asset Purchase Agreement and the performance of their respective obligations thereunder. The Company and the Buyer have agreed to cooperate with each other to promptly seek to obtain all such other consents and orders that are required to be made by any applicable law and not willfully take any action that will have the effect of delaying, impairing or impeding the receipt of any required consents and orders that are required to be made by such applicable law. This includes, to the extent required by any applicable law and, if so required, in no event later than ten business days following the signing date of the Asset Purchase Agreement: (i) filing with the Federal Trade Commission ("FTC") and the Department of Justice ("DOJ") the notification and report form required for the Asset Sale and any supplemental information requested in connection therewith pursuant to the HSR, (ii) furnishing to each other such necessary information and reasonable assistance as the other may request in connection with their respective preparation of any filing or submission which is necessary under the HSR, (iii) keeping each other apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC and the DOJ, (iv) promptly complying with any such inquiry or request, and (v) using their commercially reasonable efforts to obtain any clearance required under the HSR for the Asset Sale.

Although we expect that all required regulatory clearances and approvals will be obtained, we cannot assure that these regulatory clearances and approvals will be timely obtained or obtained at all, or that the granting of these regulatory clearances and approvals will not involve the imposition of additional conditions on the closing.

At any time before or after the consummation of the Asset Sale, notwithstanding the expiration or termination of any applicable waiting periods, the FTC or the DOJ, could take such action under the antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the consummation of the merger, seeking divestiture of substantial assets of the parties or requiring the parties to license, or hold separate, assets or terminate existing relationships and contractual rights. In addition, at any time before or after the closing, notwithstanding

the expiration or termination of the applicable waiting period under the HSR or the equivalent in any foreign jurisdictions, non-U.S. regulatory bodies and U.S. state attorneys general could take action under other applicable regulatory laws as they deem necessary or desirable in the public interest. Such action could include seeking to enjoin consummation of the Asset Sale or seeking divestiture of substantial assets of the parties. Private parties may also seek to take legal action under the antitrust laws under certain circumstances.

U.S. Antitrust

Under the HSR and the rules promulgated thereunder, the Asset Sale cannot be completed until the Company and the Buyer each file a notification and report form with the FTC and the DOJ, and the applicable waiting period thereunder has expired or been terminated. The parties made the filings required under the HSR on March 30, 2023, and the waiting period under the HSR expires May 1, 2023, at 11:59 p.m. Eastern time.

Other than obtaining HSR approval, neither the Company nor the Buyer is aware of any material federal, state or foreign regulatory approvals required to be obtained, or waiting periods required to expire, to execute the Asset Purchase Agreement or consummate the Asset Sale. If the parties discover that other approvals or filings and waiting periods are necessary, they will seek to obtain or comply with them, although, as is the case with the regulatory approvals described herein, there can be no assurance that they will be obtained on a timely basis, if at all.

Dissenters' Rights

Under Sections 623 and 910 of the New York Business Corporation Law (the "NYBCL"), you have the right to file a written objection including a notice of election to dissent to the Asset Sale and demand payment of the "fair value" of your shares of the Company. If you intend to dissent and demand payment of the "fair value" of your shares, you must comply with the provisions of Section 623 of the NYBCL in order to receive payment for your shares. We will require strict compliance with the statutory procedures by any shareholder wishing to exercise their dissenters' rights.

The following briefly summarizes the material provisions of the New York statutory procedures you must follow if you wish to dissent from the Asset Sale and receive payment of the "fair value" of your shares. Because such "fair value" could potentially be determined in a judicial proceeding, the outcome of which cannot be predicted, there is no guarantee that shareholders exercising dissenters' right would receive an amount per share equal to or greater than the amount that shareholders may receive through a public or private sale of shares of the Company.

This summary, however, is not a complete statement of all applicable requirements and is qualified in its entirety by reference to Sections 623 and 910 of the NYBCL, the full text of which appears in Annex C of this proxy statement.

This proxy statement constitutes the Company's notice to its shareholders of the availability of dissenters' rights in connection with the Asset Sale in compliance with the requirement of Section 623 of the NYBCL. Dissenters' rights are available to holders of shares of Common Stock of the Company. If you wish to consider exercising your dissenters' rights, you should carefully review the text of Section 623 of the NYBCL contained in Annex C because failure to timely and properly comply with the requirements of Section 623 will result in the loss of your rights under the NYBCL. In view of the complexity of Section 623 of the NYBCL, shareholders who may wish to dissent from the Asset Sale and pursue dissenters' rights should consult their own legal advisors to ensure that they fully and properly comply with the requirements of Section 623 of the NYBCL.

If you elect to dissent, you must satisfy the conditions stated below.

Notice of Election to Dissent

Section 623 of the NYBCL requires that if you intend to enforce your right to dissent in connection with the Asset Sale, you must file with the Company, before the Special Meeting, or at the Special Meeting, but before the vote, written objection to the Asset Sale. The written objection must include:

- a notice of your election to dissent;
- your name and residence address;

- · the number and class of shares as to which you dissent; and
- a demand for payment of the fair value of your shares if the Asset Sale is consummated.

The objection is not required from any shareholder to whom the Company did not give notice of the Special Meeting.

If you intend to exercise your dissenters' rights, you must not vote in favor of the approval of the Asset Sale. Voting against or failing to vote for the Asset Sale by itself does not constitute an election to dissent within the meaning of Section 623 of the NYBCL and will not satisfy the requirement of filing a written objection. An abstention or failure to vote will not waive your right to dissent so long as you have properly and timely filed the appropriate written objection to the Asset Sale in accordance with Section 623 and have not voted in favor of approval of the Asset Sale. A vote in favor of the Asset Sale, by proxy or online, will constitute a waiver of your election to dissent in respect of the shares voted and will nullify any previously filed written notices of election to dissent. Since a proxy left blank will be voted for approval of the Asset Sale, if you wish to exercise your dissenters' rights you must either vote against approval of the Asset Sale or abstain. You may not dissent as to less than all of your shares as to which you have a right to dissent. A nominee or fiduciary may not dissent on behalf of any beneficial owner as to less than all of the shares of the owner, as to which the nominee or fiduciary has a right to dissent, held of record by the nominee or fiduciary. The written notice of election to dissent must be in addition to and separate from any proxy or vote abstaining from or against the Asset Sale. Finally, you must continuously be the beneficial owner of your shares through the effective date of the Asset Sale. If you fail to comply with any of these conditions and the Asset Sale is completed, you will have no dissenters' rights with respect to your shares.

All written objections to the Asset Sale and notices of election to dissent pursuant to NYBCL Section 623(a) should be addressed to Enzo Biochem, Inc., 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Hamid Erfanian, Chief Executive Officer and should be executed by, or on behalf of, the record holder of the shares.

To be effective, a notice of election to dissent by a holder of our Common Stock must be made by or in the name of the registered shareholder, fully and correctly, as the shareholder's name appears on his, her or its stock certificate(s) and cannot be made by the beneficial owner if he, she or it does not also hold the shares of record. The beneficial holder must, in that case, have the registered owner submit the required notice of election to dissent.

If shares are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of a notice of election to dissent should be made in that capacity; and if the shares are owned of record by more than one person, as in a joint tenancy or tenancy in common, the demand should be executed by or for all joint owners. A record owner, such as a broker, who holds shares as a nominee for others, may exercise the record owner's dissenters' rights with respect to the shares held for one or more beneficial owners, while not exercising this right for other beneficial owners.

If you hold your shares in a brokerage account or in other nominee form and you wish to exercise dissenters' rights, you should consult with your broker or other nominee to determine the appropriate procedures.

Notice of Authorization by the Company

If the holders of at least a majority of the outstanding shares of our Common Stock approve the Asset Sale, then within ten days of the date of approval, we will give written notice of approval by registered mail to each shareholder who filed a timely notice of election to dissent, and who did not withdraw the notice of election to dissent prior to the vote or did not vote in favor of the Asset Sale, or from whom a dissent was not required.

Notice of Dissent by Shareholder Not Previously Required to Submit Notice of Election to Dissent

Within twenty days after the giving of notice to any shareholder from whom written objection was not required, and who elects to dissent, the dissenting shareholder must file with the Company a written notice of the election, stating the shareholder's name and residence address, the number and classes of shares as to which the shareholder dissents and a demand for payment of the fair value of the shares.

Rights of Dissenting Shareholders Upon Consummation of the Asset Sale

Upon consummation of the Asset Sale, each shareholder will cease to have any of the rights of a shareholder except the right to be paid the fair value of the shareholder's shares and any other rights under Section 623 of the NYBCL. A notice of election may be withdrawn by a shareholder at any time prior to the shareholder's acceptance in writing of an offer made by the Company, as provided below, but in no case later than sixty days from the date of consummation of the Asset Sale except that if the Company fails to make a timely offer, the time for withdrawing a notice of election will be extended until sixty days from the date an offer is made. Upon expiration of that time, withdrawal of a notice of election requires the written consent of the Company. To be effective, withdrawal of a notice of election must be accompanied by the return to the Company of any advance payment made to the shareholder. If a notice of election is withdrawn, or the Asset Sale is rescinded, or a court determines that the shareholder is not entitled to receive payment for the shareholder's shares, or the shareholder otherwise loses dissenters' rights, the shareholders will not have the right to receive payment for the shareholder's shares and will be reinstated to all rights as a shareholder as of the consummation of the Asset Sale, including any intervening preemptive rights and the right to payment of any intervening dividend or other distribution or, if any of these rights have expired or any dividend or distribution other than in cash has been completed, in lieu thereof, at the election of the Company, the fair value thereof in cash as determined by the Board of the Company as of the time of expiration or completion, but without prejudice otherwise to any corporate proceedings that may have been taken in the interim.

Submission of Stock Certificates for Notation

At the time of filing the notice of election to dissent or within one month after filing, the dissenting shareholder must submit the shareholder's stock certificate(s) representing the shareholder's dissenting shares to the Company, or to its transfer agent, which, upon receipt, will note conspicuously on the certificate(s) that a notice of election has been filed and return the certificate(s) to the shareholder who submitted them. Any shareholder with stock certificates who fails to submit them for notation will at the option of the Company, by written notice to the shareholder within forty-five days of the date of filing the notice of election to dissent, lose the shareholder's rights, unless a court, for good cause shown, otherwise directs. Upon transfer of a certificate bearing the notation, each new certificate issued will bear a similar notation together with the name of the original dissenting holder of the shares and a transferee will acquire no rights in the Company except those rights the original dissenting shareholder had at the time of transfer.

Written Offer by the Company

Within fifteen days after the expiration of the period within which shareholders may file their notices of election to dissent, or within fifteen days after the proposed Asset Sale is consummated, whichever is later (but in no case later than ninety days from the date of the Special Meeting at which the Asset Sale was approved), the Company will make a written offer by registered mail to each shareholder who has filed a notice of election to dissent to pay for the shareholder's shares at a specified price that the Company considers to be their fair value.

This offer will be accompanied by a statement setting forth the aggregate number of shares with respect to which notices of election to dissent have been received and the aggregate number of holders of these shares. If the Asset Sale has become effective, the offer must also be accompanied by (1) advance payment to each shareholder who submitted stock certificates to the Company for notation in an amount equal to 80% of the amount of the offer, or (2) as to a shareholder who has not yet submitted stock certificates, if the time period has not expired, or if the Company elects to grant the shareholder additional time, a statement that advance payment of 80% of the offer will be promptly made by the Company once certificates have been submitted and notated. If the Asset Sale has not become effective at the time of the offer, then the Company may wait until the Asset Sale becomes effective to send the advance payment or statement of advance payment to the dissenting shareholders and any offer extended without the payment or statement of payment may be conditioned on the effectiveness of the Asset Sale. Each advance payment or statement as to advance payment must advise the recipient shareholder that acceptance of payment does not constitute a waiver of any dissenters' rights. If the Asset Sale has not been consummated upon the expiration of the ninety-day period after the shareholders' authorization date, the offer may be conditioned upon the consummation of the Asset Sale. The offer made to each dissenting shareholder must be made at the same price per share to all dissenting shareholders of the same class and series of stock and will be accompanied by a balance sheet of the Company as of the latest available date, which must not be earlier than twelve months before the offer was made, and a profit and loss statement or statements for not less than a twelve-month period ended on the date of the balance sheet.

If within thirty days of the Company making the offer to the dissenting shareholders, the Company and any shareholder agree on the price to be paid for the shareholder's dissenting shares, payment for the shares must made within sixty days after the later of the offer or the effective date of the Asset Sale, upon surrender of stock certificates for the shares.

Judicial Enforcement

If the Company fails to make an offer for the dissenting shares within the fifteenday period described above, or if the Company makes the offer and any dissenting shareholder fails to agree with the Company upon the price to be paid for their shares within thirty days, the Company must within twenty days of the expiration of the applicable period, institute a special proceeding in the Suffolk County Supreme Court (the supreme court in the judicial district where the office of the Company is located) to determine the rights of dissenting shareholders and to fix the fair value of their shares. If the Company fails to timely institute the special proceeding within the twenty-day period, any dissenting shareholder may institute the proceeding within thirty days of the expiration of the twenty-day period. If a special proceeding is not then timely instituted, all dissenters' rights will be lost unless the court, for good cause shown, otherwise directs.

All dissenting shareholders, excepting those who have agreed with the Company upon the price to be paid for their shares, will be made parties to the special proceeding. The Company must serve a copy of the petition in the special proceeding upon each dissenting shareholder who is a resident of New York in the manner provided by law for the service of a summons, and upon each nonresident dissenting shareholder either by registered mail and publication, or in any other manner that is permitted by law.

If a special proceeding is instituted, the court will determine whether a dissenting shareholder is entitled to receive payment for the shareholder's shares, and the court will fix the fair value of the shares as of the close of business on the day prior to the Special Meeting at which the Asset Sale was authorized. In fixing the fair value of the shares, the court will consider the nature of the Asset Sale giving rise to the shareholder's right to receive payment for shares and its effects on the Company and its shareholders, the concepts and methods then customary in the relevant securities and financial markets for determining fair value of shares of a corporation engaging in a similar transaction under comparable circumstances and all other relevant factors. The court will determine the fair value of the shares without a jury and without referral to an appraiser or referee. Upon application by the Company or by any shareholder who is a party to the proceeding, the court may, in its discretion, permit pretrial disclosure, including disclosure of any expert's reports relating to the fair value of the shares whether or not intended for use at the trial in the proceeding.

A final order will be entered by the court against the Company for payment to the dissenting shareholders including interest from the date the Asset Sale was consummated to the date of payment. In determining the rate of interest, the court will consider all relevant factors, including the rate of interest which the Company would have had to pay to borrow money during the pendency of the proceeding. If the court finds that the refusal of any shareholder to accept the corporate offer of payment for the shareholder's shares was arbitrary, vexatious or otherwise not in good faith, no interest will be allowed to that shareholder. Each party to the special proceeding will bear its own costs and expenses, including the fees and expenses of its counsel and of any experts employed by it, provided that the court may apportion or assess all or any part of the costs, expenses and fees incurred by the Company against any or all of the dissenting shareholders who are parties to the proceeding, including any who have withdrawn their notices of election, if the court finds that their refusal to accept the Company offer was arbitrary, vexatious or otherwise not in good faith. Similarly, the court may apportion and assess all or any part of the costs, fees and expenses incurred by any or all of the dissenting shareholders who are parties to the proceeding against the Company if the court finds any of the following: (1) the fair value of the shares as determined by the court materially exceeds the amount which the Company offered to pay; (2) no offer or required advance payment was made by the Company; (3) the Company failed to institute the special proceeding within the specified period; or (4) the action of the Company in complying with its obligations as provided in Section 623 of the NYBCL was arbitrary, vexatious or otherwise not in good faith. In making any determination as provided in clause the first clause, the court may consider the dollar amount or the percentage, or both, by which the fair value of the shares as determined exceeds the Company's offer. The Company must make payment to each dissenting shareholder that is a party to the proceeding pursuant to the court order within sixty days after final determination of the special proceeding, upon surrender of the certificates for any shares represented by certificates.

Shares acquired by the Company upon the payment of the agreed value or of the amount due under the final order, may be held and disposed of as the Asset Purchase Agreement provides.

Other Considerations

No payment may be made to a dissenting shareholder at a time when the Company is insolvent or when the payment would make it insolvent. In that event, the dissenting shareholder must, at the shareholder's option: withdraw the shareholder's notice of election, which will then be deemed withdrawn with the written consent of the Company; or retain the shareholder's status as a claimant against the Company and, if the Company is liquidated, be subordinated to the rights of creditors of the Company, but have rights superior to the non-dissenting shareholders, and if the Company is not liquidated, retain the shareholder's right to be paid for the shareholder's shares, which right the Company must satisfy when the restrictions of this paragraph do not apply. The dissenting shareholder must exercise an option by written notice given to the Company within thirty days after the Company has given the shareholder written notice that payment for the shareholder's shares cannot be made because of the restrictions of this paragraph. If the dissenting shareholder fails to exercise an option, then the Company will exercise the option by written notice given to shareholder within twenty days after the expiration of the thirty-day period.

The enforcement by a shareholder of the shareholder's right to receive payment for shares excludes the enforcement by that shareholder of any other right to which the shareholder might otherwise be entitled by virtue of share ownership, except as provided in paragraph (e) of Section 623 of the NYBCL, and except that Section 623 of the NYBCL does not exclude the right of a shareholder to bring or maintain an appropriate action to obtain relief on the ground that the Asset Sale is unlawful or fraudulent as to the shareholder.

Except as otherwise expressly provided in Section 623 of the NYBCL, any notice to be given by the Company to a shareholder under Section 623 of the NYBCL shall be given in the manner provided in Section 605 of the NYBCL (Notice of meetings of shareholders).

MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE ASSET SALE

The following discussion is a general summary of the material anticipated U.S. federal income tax consequences of the Asset Sale. This discussion is based upon the Internal Revenue Code of 1986, as amended (the "Code"), its legislative history, currently applicable and proposed Treasury regulations under the Code (the "Treasury Regulations"), and published rulings and decisions, all as currently in effect as of the date of this proxy statement, and all of which are subject to change, possibly with retroactive effect. Tax considerations under state, local, and non-U.S. laws, or federal laws other than those pertaining to income tax, are not addressed in this proxy statement. The following discussion has no binding effect on the IRS or the courts.

Our shareholders will not realize any gain or loss for U.S. federal income tax purposes as a result of the Asset Sale

The proposed Asset Sale will be treated as a sale of corporate assets in exchange for cash and the assumption of certain liabilities. The adjusted tax basis in the assets being sold is less than the sales proceeds that will be received from the Buyer in which case we will incur U.S. federal and state income tax as a result of a realized gain from the Asset Sale. However, we anticipate that our tax attributes, including our U.S. federal and state net operating loss carryforwards ("NOLs"), will be available to offset a substantial portion of our U.S. federal and state income tax liabilities resulting from such gain.

In addition, in general, under Section 382 of the Code, a corporation that undergoes an "ownership change" is subject to annual limitations on its ability to use its pre-change NOLs or other tax attributes to offset future taxable income or reduce taxes. Our past issuances of stock and other changes in our stock ownership may have resulted in an ownership change within the meaning of Section 382 of the Code; accordingly, our pre-change NOLs may be subject to limitation under Section 382. However, we do not anticipate any limitation under Section 382.

The determination of our realized gain on the proposed Asset Sale and whether and to what extent our tax attributes will be available to offset the gain is highly complex and is based in part upon facts that will not be known until the completion of the Asset Sale. Therefore, it is possible that we could incur substantial U.S. federal and state income taxes as a result of the proposed Asset Sale.

PROPOSAL 1 — TO APPROVE THE ASSET SALE

This section of the proxy statement describes the material provisions of the Asset Purchase Agreement but does not purport to describe all the provisions of the Asset Purchase Agreement. The following summary is qualified in its entirety by reference to the complete text of the Asset Purchase Agreement, which is included as Annex A to this proxy statement and is incorporated into this proxy statement by reference. We urge you to read the full text of the Asset Purchase Agreement because it is the legal document that governs the Asset Sale. The Asset Purchase Agreement has been included to provide you with information regarding its terms. It is not intended to provide you with any other factual information about us. Such information can be found elsewhere in this proxy statement and in the other public filings we make with the SEC, which are available without charge at www.sec.gov.

Structure of the Asset Sale

We have agreed to sell substantially all the operating assets, and assign certain liabilities, of our clinical laboratory business (the "Business") to the Buyer for an aggregate Purchase Price of \$146 million payable in cash at closing, subject to adjustment and subject to the conditions described below, under "Conditions to Closing of the Asset Sale."

Principal Provisions of the Asset Purchase Agreement

Assets to be Sold

The assets to be purchased by the Buyer constitute substantially all the assets of the Business, including customer accounts, sales data, records, and promotional materials, customer and vendor lists, inventory and supplies, tangible personal property and equipment, reserves, security deposits, prepaid expenses, and credits related to assigned contracts.

The acquired assets also include intellectual property assets used exclusively in the Business, such as trade names and domain names, but the acquired assets do not include any patents or any intellectual property related to products manufactured or procured by the Company or its subsidiaries on behalf of the Business before the closing. In connection with the transfer of such intellectual property, the Company will agree not to sue Labcorp or its affiliates for infringement of certain intellectual property owned by the Company or its affiliates that may be used by Labcorp in its operation of the Business after closing. The covenant not to sue does not apply to activities of Labcorp outside of its operation of the Business or the infringement of intellectual property rights held by the Company or its affiliates with regard to products that the Company traditionally procured for the Business, including reagents, probes, antibodies, dyes and the GENFLEX equipment.

Liabilities to be Assumed

The Buyer will assume only those liabilities arising after the closing under leases and other contracts that constitute acquired assets as identified in the schedules to the Asset Purchase Agreement. All other liabilities will remain our obligations.

Consideration to Be Received in the Asset Sale and Purchase Price Adjustments

We will receive approximately \$146 million in exchange for substantially all of the assets of the Business. We will retain our existing cash, cash equivalents and accounts receivable, among other assets, upon consummation of the Asset Sale. The total Purchase Price to be paid upon closing is \$146 million, subject to the following adjustments:

- less the dollar value of accrued, but unused vacation time of Business employees Buyer hires at closing;
- less \$5,000,000 (the "Escrow Amount"), which shall be deposited with an escrow agent and held, subject to disbursement to cover indemnification obligations of the Company and Seller, as provided in the Asset Purchase Agreement;
- less the dollar amount of debt underlying certain liens on the assets of the Business, which is to be
 paid off at the closing;
- <u>less</u> any amounts deducted for motor vehicles not delivered at the closing;

- <u>plus</u> the amount of all reserves, security and other deposits, advances and prepaid expenses and credits that constitute acquired assets and are assignable to Buyer and related to the assigned contracts or any other acquired asset; and
- <u>plus</u> one-half (up to \$1,600,000) of the aggregate amount of the retention bonuses previously paid or to be paid by the Company to its employees engaged in the Business, in connection with the transactions contemplated by the Asset Purchase Agreement.

Further, upon obtaining shareholder approval, Buyer shall pay us a daily fee of \$75,000, payable monthly, until closing, which amount will be wholly or partly credited against the Purchase Price to be paid upon closing. Generally the entire fee is credited against the purchase price unless HSR Approval has been obtained, all other conditions to closing in the Company's control are met, and Buyer elects not to close earlier than 120 days after signing, in which case one-half of the fee payable after HSR Approval can be retained by the Company. In the event that the Asset Purchase Agreement is terminated, the Company would be obligated to return all amounts previously paid to the Company with respect to such daily fee, except in very limited circumstances.

Terms of Indemnification and Escrow

The Asset Purchase Agreement provides that at the closing, Buyer shall deposit the Escrow Amount into an escrow account for the purpose of satisfying indemnification claims, if any, made by Buyer against Seller. The Escrow Amount will be held in the escrow account for a period of fifteen months after the closing of the Asset Sale

The representations and warranties of the Company and the right of the Buyer to indemnification under the Asset Purchase Agreement generally survive the closing date of the Asset Sale for 15 months. The Company will deposit the Escrow Amount into an escrow account at the closing of the Asset Sale to secure its indemnification obligations to the Buyer with respect to breaches of the Company's representations and warranties, breaches of covenants, losses related to excluded assets or excluded liabilities and certain other matters set forth in the Asset Purchase Agreement.

Other than in the case of fraud or breach of the Fundamental Representations or breach of representations and warranties related to certain tax matters, the Seller shall not be liable to the Buyer for indemnification unless and until the aggregate amount of the indemnification obligations exceeds the Deductible, in which event the Seller shall then be liable for losses in excess of the Deductible up to an amount equal to \$584,000 (payable from the Escrow Amount), and the remainder of the losses shall be satisfied from Buyer's R&W Policy. With respect to claims by Buyer regarding certain other tax matters and breaches of Fundamental Representations, subject to certain limitations, Buyer is entitled to satisfy losses first from the Escrow Amount (up to an amount equal to the Deductible), second from its R&W Policy, and last from the Escrow Amount or, if exhausted, from Seller directly up to the amount of the Purchase Price. With respect to any claim by Buyer regarding (i) excluded liabilities, excluded assets or the excluded business specified in the Asset Purchase Agreement, or (ii) breach, default of, or failure by Seller or the Company to perform the covenants or obligations of the Asset Purchase Agreement, Buyer is entitled to satisfy losses first from Escrow Amount and then from Seller directly.

Conditions to the Closing of the Asset Sale

The parties' obligation to complete the Asset Sale is subject to certain customary conditions, including, but not limited to:

- · the expiration or termination of the waiting period under HSR ("HSR Approval"); and
- obtaining approval of the Company's shareholders to consummate the Asset Sale.

The Buyer's obligation to complete the Asset Sale is subject to certain conditions, including, but not limited to:

- the absence of a Material Adverse Effect (as defined in the Asset Purchase Agreement);
- delivery of certain lease assignments and instruments of title or transfer with respect to acquired assets; and
- certain representations and warranties (including with respect to intellectual property, title and use of
 assets, and certain legal compliance) being true and correct in all material respects as of the closing.

The assertions embodied in the representations and warranties are qualified by information in a confidential disclosure letter that we provided to the Buyer in connection with signing the Asset Purchase Agreement and are not intended to provide any other factual information about us. The disclosure letter contains information that modifies, qualifies and creates exceptions to the representations and warranties. However, if we become aware of specific material facts that contradict the representations and warranties included in the Asset Purchase Agreement, we will provide updated disclosure as necessary to make disclosures in this proxy statement not misleading.

Covenants Relating to the Conduct of Our Business

Pursuant to the Asset Purchase Agreement, the Company and the Buyer are subject to certain PreClosing Covenants regarding the conduct of our business prior to closing. These covenants include:

- not to sell or encumber any of the acquired assets;
- to comply with all laws applicable to the Business, the acquired assets and the assumed liabilities;
- preserve and maintain in effect all material permits necessary to carry on the Business;
- not to make any material change in its methods of accounting or accounting principles or practices
 used in the Business (including with respect to reserves);
- not to make or authorize any change in our charter or bylaws to the extent such amendment or change would prevent, impede or delay the shareholder approval or the consummation of the Asset Sale;
- · not to terminate or close any facility or operation used in the Business; and
- not to issue or sell any of the equity or convertible securities of Seller.

Further, pursuant to the Asset Purchase Agreement, for a period of five years after the closing date of the Asset Purchase Agreement (the "Restricted Period"), the Company will be subject to certain covenants regarding the conduct of our business. These covenants include:

- not to engage, invest in, own, manage, operate, finance, control, advise, render services to, be
 employed by any person (including direct competitors of Buyer) engaged in any business or activity
 that competes with the Business as conducted on the signing date or on the closing date of the Asset
 Purchase Agreement in the New York, New Jersey and Connecticut;
- not to use any names licensed to Buyer in any business that competes with the Business;
- not to use the trade names "Enzo Clinical Labs" or "Enzo Labs" or any other confusingly similar trade name that combines the names "Enzo" and "labs" in connection with any business of any kind;
- not to directly or indirectly induce or solicit, or attempt to cause, induce or solicit (i) any person who is or was an employee, consultant or independent contractor of the Business on, or during the 12 months prior to, the closing date of the Asset Purchase Agreement, or (ii) any person who is or was an employee, consultant or independent contractor of Buyer or its affiliates at any time during the Restricted Period with whom the Company has or has had contact (or with respect to whom the Company obtains or obtained confidential information) during the Restricted Period to (A) terminate such person's employment or consulting relationship with Buyer or its affiliates; (B) in any way interfere with the relationship between such person and Buyer or its affiliates; or (C) hire, retain, employ or otherwise engage, or attempt to hire, retain, employ or otherwise engage, such individual as an employee, independent contractor or otherwise, unless (x) such individual's employment with Buyer has been involuntarily terminated by Buyer or its affiliates or (y) such individual's employment has been voluntarily terminated by such employee not less than 12 months prior to any solicitation from the Company or any of its affiliates;
- not to solicit, induce or otherwise cause, or attempt to solicit, induce or otherwise cause, any
 customer, supplier, licensor, licensee or any prospective customer, supplier, licensor or licensee that
 has been contacted or targeted for contact by the Company on or before the closing date of the Asset
 Purchase Agreement, or any other person that had, has or is expected to have a business relationship
 with the

Company, to (A) terminate, curtail or otherwise modify its relationship with the Business or (B) engage in business with a competitor of the Business; or (ii) interfere in any way with the relationship between the Business and any of its customers, suppliers, licensors, licensees or any such prospective customers, suppliers, licensors or licensees that have been contacted or targeted for contact by the Business on or before the closing date of the Asset Purchase Agreement, or any other person engaged in a business relationship with the Business, except with respect to the excluded business specified in the Asset Purchase Agreement; and

not to make any disparaging statement, either orally or in writing, regarding the Buyer, the Business, or any products or services thereof (including its affiliates), except to protect or enforce its rights or interests under the Asset Purchase Agreement, to defend against any claim brought by the Buyer under the Asset Purchase Agreement, to make any truthful statements in any legal action or as otherwise required by applicable law, or in a manner that limits competitive speech or commercial comparisons regarding services or products of the parties.

Transition Services

Subject to the terms and conditions of the Asset Purchase Agreement, if requested by Buyer no later than five business days prior to the closing date of the Asset Sale, we will enter into a transition services agreement with Buyer to provide certain transition services reasonably necessary to allow Buyer to transition any remaining customers of the Business to Buyer or provide Buyer with access to the assigned locations specified in the Asset Purchase Agreement and not transferred to Buyer on the closing date of the Asset Sale, in each case, for up to six months following the closing of the Asset Sale.

Termination

The Asset Purchase Agreement may be terminated at any time prior to the closing date of the Asset Sale under certain circumstances, including, but are not limited to the following:

- the Asset Sale has not closed by the 210th day after the date of the Asset Purchase Agreement, subject to extension by Buyer under certain circumstances;
- the shareholder approval has not been obtained at the meeting of the Company's shareholders to approve the Asset Sale;
- a law or governmental order prohibits the closing;
- either party materially breaches its representations, warranties or covenants;
- by the Buyer, if, among other things, our board fails to recommend the Asset Sale to its shareholders;
- by the Company in order to enter into a Superior Proposal.

Breakup Fee

If the Asset Purchase Agreement is terminated under certain circumstances, including but not limited to those described below, we will be required to pay to the Buyer a Breakup Fee of \$5,000,000:

- our Board fails to recommend the Asset Sale to its shareholders or failed to reaffirm its recommendation under certain circumstances;
- the Company terminates the Asset Purchase Agreement in order to enter into a Superior Proposal, or
 otherwise enters into or consummates an alternative asset sale with respect to the Business under
 certain circumstances.

Amendment

The Asset Purchase Agreement may be amended only if such amendment is in writing and is signed by the Buyer, the Seller and the Company.

No Solicitation

The Asset Purchase Agreement contains a "no shop" restriction on the Company's ability to solicit third party proposals from, provide confidential information to, and engage in discussions and negotiations with, potential third-party bidders. The "no shop" restriction contains a "fiduciary out" provision whereby the Company may engage with a third-party bidder from whom the Company receives an unsolicited bona fide written acquisition proposal after the date of the Asset Purchase Agreement and prior to obtaining shareholder approval which our Board determines in good faith after consultation with outside legal counsel that (i) failure to take action would reasonably be expected to breach the Board's fiduciary duties under applicable law, (ii) the proposal is, or would reasonably be expected to lead to, an alternative transaction that is more favorable to the Company's shareholders from a financial point of view than the Asset Sale and (iii) the proposal is reasonably likely to be consummated on the terms proposed and to constitute a Superior Proposal, as defined in the Asset Purchase Agreement..

Thus, at any time prior to obtaining shareholder approval of the Asset Sale, in response to a Superior Proposal, our Board may, among other actions, withdraw or modify its recommendation contained in this proxy statement to approve the Asset Sale or recommend the approval of an alternative acquisition proposal, if our Board concludes in good faith (after consultation with outside counsel) that the failure to take such action would reasonably likely to cause our Board to breach its fiduciary duties under applicable law (see previous section entitled "Breakup Fee" for details related to any fees we may be obligated to pay if our Board exercises its option to withdraw or materially modify its recommendation contained in this proxy statement).

Benefit Arrangements

The Buyer shall offer employment to certain of Business employees to be specified no later than two weeks prior to closing. The Buyer has no obligation to assume sponsorship or maintain any of our current employee benefit plans or arrangements.

Voting Agreements

In connection with the entry into the Asset Purchase Agreement, on March 16, 2023, Buyer entered into separate Voting Agreements with each of the following shareholders that collectively beneficially own up to approximately 11% of the Company's outstanding shares of Common Stock: (i) Hamid Erfanian; (ii) Ian Walters; (iii) Kara Cannon; (iv) Mary Tagliaferri; (v) Patricia Eckert; and (vi) Brad Radoff. The Voting Agreements provide that each of the above shareholders shall, among other things, vote all of the Common Stock that it beneficially owns: (i) in favor of adoption and approval of the Asset Purchase Agreement and the Asset Sale; (ii) against any alternative transaction (as described in the Asset Purchase Agreement); and (iii) against any other action that is intended or could reasonably be expected to impede or interfere with or materially delay the Asset Sale or any other transactions contemplated by the Asset Purchase Agreement. Each of the above shareholders granted an irrevocable proxy to Buyer to vote its Common Stock in accordance with the foregoing.

The Voting Agreements further provide that in the event any of the above shareholders acquires any additional shares of Common Stock or other interests in or with respect to the Company, the shareholder much promptly notify the Buyer of such acquisition, and the acquired shares of Common Stock or other interests shall be subject to the provisions of the Voting Agreement. The Voting Agreements and the irrevocable proxy granted to Buyer thereunder will automatically terminate upon the occurrence of certain circumstances, including the termination of the Asset Purchase Agreement in accordance with its terms or the closing of the Asset Sale.

Further Assurances and Cooperation

Subject to the terms and conditions of the Asset Purchase Agreement, we will executive and deliver any additional documents and instruments and perform any additional acts that may be necessary or appropriate to effectuate the conveyance, transfer and assignment of the acquired assets to the Buyer and to effectuate the transactions contemplated by the Asset Purchase Agreement. We will, at the Buyer's expense, use commercially reasonable efforts to assist and cooperate with the Buyer as may be reasonably necessary to transfer and assign to Buyer the purchased intellectual property and to record and perfect the Buyer's rights in the purchased intellectual property rights. The Buyer agreed to cooperate and provide us with such information about the Buyer as is reasonably requested in connection with the disclosure required in this proxy statement.

Other Significant Provisions

We and the Buyer have agreed that each party will be entitled to an injunction or injunctions to prevent breaches or threatened breaches of the Asset Purchase Agreement and to enforce specifically the terms and provisions of the Asset Purchase Agreement.

RISKS RELATED TO THE ASSET SALE

In addition to the other information contained in this Proxy Statement, you should carefully consider the following risk factors when deciding whether to vote to approve the Asset Sale Proposal and the Adjournment Proposal. You should also consider the information in our other reports on file with the SEC. See "WHERE YOU CAN FIND MORE INFORMATION."

The Company has incurred and will continue to incur substantial expenses, including transaction-related costs, pending the Asset Sale.

Claims, liabilities and expenses from operations, such as operating costs, salaries, directors' and officers' insurance, payroll and local taxes, legal, accounting and consulting fees and office expenses will continue to be incurred by us during the pendency of the Asset Sale. Further, Enzo has incurred, and expects to continue to incur, a number of non-recurring transaction-related costs in initiating and completing the Asset Sale. Non-recurring transaction costs include, but are not limited to, fees paid to Enzo's financial, legal and accounting advisors, filing fees and printing costs. These fees and costs have been, and will continue to be, substantial. We cannot estimate what the aggregate of these expenses will be and these costs may be higher than expected. There can be no assurance of the exact amount of net cash proceeds Enzo will receive from the Asset Sale or the exact timing at which it will receive such proceeds. Therefore, it is uncertain the extent to which our financial condition and operations will benefit from or improve as a result of or after the Asset Sale.

The Asset Purchase Agreement also includes customary termination provisions for both the Company and Buyer and provides that, in connection with the termination of the Asset Purchase Agreement under specified circumstances, including termination by the Company to accept and enter into a definitive agreement with respect to an unsolicited Superior Proposal, the Company will be required to pay Buyer a termination fee of \$5,000,000, or reimbursement of Buyer's expenses of up to \$5,000,000.

Subject to the terms and conditions stated in the Asset Purchase Agreement, after shareholder approval of the Asset Sale, Buyer will be obligated to pay a fee to the Company for each day after the date of such approval until the closing of the Asset Sale. At the closing of the Asset Sale, such fee will be wholly or partially credited against the purchase price.

There can be no assurance that the Asset Purchase Agreement will close and that if it does close, the exact proceeds to be received by the Company.

While the Asset Sale is pending, it creates uncertainty about our future, which could materially and adversely affect our business, financial condition and results of operations.

While the Asset Sale is pending, it creates uncertainty about our future. As a result, our current or potential business partners may decide to delay, defer or cancel entering into new business arrangements with us pending consummation of the Asset Sale or termination of the Asset Purchase Agreement. In addition, while the Asset Sale is pending, we are subject to a number of risks, including:

- the diversion of management and employee attention from our day-to-day business, which impacts our ability to operate our business in the ordinary course and generate revenues;
- the loss of employees who may depart due to their concern about losing their jobs following the Asset
 Sale or a shift in loyalty of employees of the Company who see the Buyer as their *de facto* employer
 even before the consummation of the Asset Sale; and
- our inability to respond effectively to competitive pressures, industry developments and future opportunities.

The occurrence of any of these events individually or in combination could materially and adversely affect our business, financial condition and results of operations.

The Asset Purchase Agreement limits our ability to pursue alternatives to the Asset Sale

The Asset Purchase Agreement contains provisions that make it substantially more difficult for us to sell the Business' assets to a party other than the Buyer. Specifically, the Company agreed not to solicit any acquisition proposals until the date of closing or the proper termination of the Asset Purchase Agreement except that, subject

to our obligation to pay the Buyer a breakup fee of \$5,000,000 at any time prior to obtaining shareholder approval of the Asset Sale, in response to a superior proposal, our board may, among other actions, withdraw or materially modify its recommendation contained in this proxy statement to approve the Asset Sale or recommend the approval of an alternative acquisition proposal, if our board concludes in good faith (after consultation with outside legal and financial advisors) that the failure to take such action would reasonably be expected to cause our board to breach its fiduciary duties under applicable law.

If claims for indemnification are made by the Buyer and paid from the Escrow Amount or directly from the Company, it will have the effect of reducing the amount of the aggregate Purchase Price actually received by the Company.

The Purchase Price to be paid in connection with the Asset Sale is \$146,000,000 in cash, subject to adjustment, from which the Escrow Amount will be placed into an escrow account at the closing of the Asset Sale to secure the Company's indemnification obligations to the Buyer with respect to breaches of the Company's representations and warranties, breaches of covenants, losses related to excluded assets or excluded liabilities and certain other matters set forth in the Asset Purchase Agreement. Any claims for indemnification made by the Buyer and paid from the Escrow Amount or directly from the Company will have the effect of reducing the amount of the aggregate Purchase Price actually received by the Company. There can be no assurance that the Escrow Amount will not be used to satisfy indemnification claims or that the Escrow Agreement and/or R&W Insurance will be sufficient to satisfy all of Buyer's indemnification claims. As a result of certain claims, the Company may need to pay the Buyer out of pocket, which would have the effect of reducing the benefits received by the Company in the Asset Sale.

There can be no guarantee that the Asset Sale will be completed and, if not completed, we may have to file for bankruptcy and liquidation.

The consummation of the Asset Sale is subject to the satisfaction or waiver of various conditions, including the approval of the Asset Sale by our shareholders. We cannot guarantee that the closing conditions set forth in the Asset Purchase Agreement will be satisfied. If we are unable to satisfy the closing conditions in Buyer's favor or if other mutual closing conditions are not satisfied, Buyer will not be obligated to complete the Asset Sale. If the Asset Sale is not completed, we may be subject to a number of risks, including the following:

- we may not be able to identify an alternate transaction, or if an alternate transaction is identified, such alternate transaction may not result in an equivalent price to what is proposed in the Asset Sale;
- the trading price of our common stock may decline to the extent that the then current market price reflects a market assumption that the Asset Sale will be consummated;
- our relationships with our customers, suppliers and employees may be damaged beyond repair and the value of our assets will likely significantly decline; and

The occurrence of any of these events individually or in combination will likely materially and adversely affect our business, financial condition and results of operations, cause the market value of our common stock to significantly decline or become worthless and force us to file for bankruptcy protection, liquidate and windup our operations.

Our executive officers and directors may have interests in the Asset Sale other than, or in addition to, the interests of our shareholders generally.

Members of our Board and our executive officers may have interests in the Asset Sale that are different from, or are in addition to, the interests of our shareholders generally, including as discussed in the section above captioned "THE ASSET SALE — Interests of Certain Persons in the Asset Sale". Our Board was aware of these interests and considered them, among other matters, in approving the Asset Purchase Agreement.

We may be subject to securities litigation, which is expensive and could divert our attention.

We may be subject to securities class action litigation in connection with the Asset Sale. Securities litigation against us could result in substantial costs and divert our management's attention from closing the Asset Sale, which could harm our business and increase our expenses, which could decrease our net proceeds from the Asset Sale.

Vote Required and Board Recommendation

Approval of the proposal to approve the Asset Sale and adopt the Asset Purchase Agreement requires the affirmative vote of a majority of the votes of the outstanding shares entitled to vote thereon.

 $Assuming \ a \ quorum \ is \ present, \ abstentions \ will \ have \ the \ same \ effect \ as \ a \ vote \ ``AGAINST" \ the \ Asset \ Sale \ Proposal.$

OUR BOARD OF DIRECTORS RECOMMENDS THAT YOU VOTE "FOR" THE ASSET SALE PROPOSAL.

PROPOSAL 2 — APPROVE THE ADJOURNMENT OF THE SPECIAL MEETING, IF NECESSARY OR APPROPRIATE, TO SOLICIT ADDITIONAL PROXIES IF THERE ARE INSUFFICIENT VOTES AT THE TIME OF THE SPECIAL MEETING TO APPROVE THE ASSET SALE PROPOSAL

Adjournment of the Special Meeting

In the event that the number of shares of our Common Stock present in person or represented by proxy at the Special Meeting and voting "FOR" the Asset Sale Proposal is insufficient to approve the Asset Sale Proposal, we may move to adjourn the Special Meeting in order to enable our board to solicit additional proxies in favor of the Asset Sale Proposal. In that event, we will ask our shareholders to vote only upon the Adjournment Proposal and not on the other proposals discussed in this proxy statement. In addition, pursuant to our bylaws the Board may adjourn the Special Meeting without approval of the shareholders whether or not a quorum is present. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each shareholder of record entitled to vote at the meeting.

Vote Required and Board Recommendation

Approval of the proposal to adjourn the Special Meeting to a later date, to solicit additional proxies if there are not sufficient votes in favor of the Asset Sale Proposal requires the affirmative vote of a majority of the total votes cast at the Special Meeting by the holders of shares present in person or represented by proxy and entitled to vote on such action at the Special Meeting.

Assuming a quorum is present, abstentions will have no effect on the outcome of the vote with respect to the Adjournment Proposal.

OUR BOARD OF DIRECTORS RECOMMENDS THAT YOU VOTE "FOR" THE ADJOURNMENT PROPOSAL.

MARKET PRICE AND DIVIDEND DATA

The common stock of the Company is traded on the New York Stock Exchange (Symbol: ENZ). The following table sets forth the closing high and low price of the Company's common stock for the periods indicated as reported on the New York Stock Exchange.

2022 Fiscal Year (August 1, 2021 to July 31, 2022):

	High	Low
1 st Quarter	\$ 4.09	\$ 3.04
2 nd Quarter	\$ 3.60	\$ 3.06
3 rd Quarter	\$ 3.47	\$ 2.55
4 th Quarter	\$ 2.62	\$ 2.00

2021 Fiscal Year (August 1, 2020 to July 31, 2021):

	High	Low
1 st Quarter	\$ 2.6	1 \$ 1.82
2 nd Quarter	\$ 3.2	0 \$ 1.89
3 rd Quarter	\$ 4.0	8 \$ 2.75
4 th Quarter	\$ 3.4	0 \$ 2.86

The following table sets forth the closing sales price per share of our Common Stock, as reported on The New York Stock Exchange on March 16, 2023, the last full trading day before the public announcement of the proposed asset sale, and April 17, 2023, the latest practicable date before the filing of this proxy statement:

March 16, 2023	\$ 1.19
April 17, 2023	\$ 2.55

As of April 17, 2023, the Company had approximately 742 shareholders of record of its common stock.

The Company has not paid a cash dividend on its common stock. The Board will determine whether a dividend will be paid in the future based upon available alternative uses for such funds and the interests of the shareholders.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

Set forth below is information, as of April 5, 2023, the Record Date, concerning stock ownership of all persons known by the Company to own beneficially 5% or more of the shares of Common Stock of the Company, the named executive officers, all current directors, and all current directors, executive officers and key employees of the Company as a group, based upon the number of outstanding shares of Common Stock as of the close of business on April 5, 2023.

The percentages in the "Percent of Class" column are calculated in accordance with the rules of the SEC, under which a person may be deemed to be the beneficial owner of shares if that person has or shares the power to vote or dispose of those shares or has the right to acquire beneficial ownership of those shares within 60 days (for example, through the exercise of an option or warrant). The shares shown in the table as beneficially owned by certain individuals may include shares owned by certain members of their respective families. Because of these rules, more than one person may be deemed to be the beneficial owner of the same shares. The inclusion of the shares shown in the table is not necessarily an admission of beneficial ownership of those shares by the person indicated. Except as otherwise indicated, each of the persons named has sole voting and investment power with respect to the shares shown.

Title of Class	Name and Address of Beneficial Owner	Amount and Nature of Beneficial Ownership ⁽¹⁾	Percent of Class ⁽²⁾
Common Stock	Elazar Rabbani, Ph.D.	2,273,336(3)	4.5%
Common Stock	Hamid Erfanian	345,147 ⁽⁴⁾	*
Common Stock	Kara Cannon	285,928(5)	*
Common Stock	Patricia Eckert	36,067(6)	*
Common Stock	David Bench	10,138(7)	*
Common Stock	Bradley L. Radoff	4,663,726(8)	9.2%
Common Stock	Mary Tagliaferri, M.D.	96,633(9)	*
Common Stock	Ian B. Walters, M.D.	96,633(10)	*
Common Stock	Harbert Management Corp	5,175,913(11)	10.4%
Common Stock	Renaissance Technologies, LLC	2,813,964(12)	5.7%
Common Stock	James G. Wolf	4,942,400(13)	9.9%
Common Stock	All directors, executive officers and key employees as a group (8 persons)	7,807,606(14)	15.4%

Represents beneficial ownership of less than 1%.

- Except as otherwise noted in the footnotes to the table, all shares of Common Stock are beneficially owned and the sole investment and voting power is held by the persons named, and such persons' address is c/o Enzo Biochem, Inc., 81 Executive Blvd., Suite 3 Farmingdale, NY 11735.
- (2) For directors, executive officers and key employees, based upon 49,662,821 shares of Common Stock of the Company considered outstanding as of the close of business on April 5, 2023. Common Stock not outstanding but deemed beneficially owned by virtue of the right of an individual to acquire shares within 60 days from April 5, 2023 is treated as outstanding when determining the amount and percentage of Common Stock owned by directors and executive officers individually and as a group.
- (3) Includes (i) 348,100 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, (ii) 5,308 shares of Common Stock held in the name of Dr. Rabbani as custodian for certain of his children, (iii) 18,794 shares of Common Stock held in the name of Dr. Rabbani's wife as custodian for certain of their children, and (iv) 4,902 shares of Common Stock held in the Company's 401(k) plan.
- (4) Includes (i) 233,333 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023.
- (5) Includes (i) 237,667 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 35,661 shares of Common Stock held in the Company's 401(k) plan.
- (6) Includes (i) 16,667 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,400 shares of Common Stock held in the Company's 401(k) plan.
- (7) Mr. Bench resigned effective November 14, 2022.
- (8) Includes (i) 430,000 shares of Common Stock held by The Radoff Family Foundation of which Mr. Radoff is deemed to be the beneficial owner, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award. The address of Bradley L. Radoff is 2727 Kirby Drive Unit 29L Houston, Texas 77098.

- (9) Includes (i) 77,101 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award.
- (10) Includes (i) 77,101 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 19,532 shares of Common Stock which are issuable within 60 days from April 5, 2023 under a restricted stock unit award.
- (11) The address of Harbert Management Corp. is 2100 Third Avenue North, Suite 600, Birmingham, AL 35203. This information is based solely on a Schedule 13F filed for the period ended December 31, 2022.
- (12) The address of Renaissance Technologies, LLC is 800 Third Avenue, New York, NY 10022. This information is based solely on a Schedule 13G filed for the period ended December 31, 2022.
- (13) The address of James G. Wolf is 105, Flyway Drive, Kiawah Island, SC 29455. This information is based solely on a Schedule 13D/A filed on November 23, 2022.
- (14) Includes (i) 989,969 shares of Common Stock issuable upon the exercise of options which are exercisable within 60 days from April 5, 2023, and (ii) 58,596 shares of Common Stock which are issuable within 60 days from April 5, 2023 under restricted stock unit awards.

FUTURE SHAREHOLDER PROPOSALS

Shareholder Proposals for the 2023 Annual Meeting

Proposals of shareholders intended to be included in the Company's Proxy Statement and form of proxy for use in connection with the Company's 2023 Annual Meeting must be received by the Company's Secretary at the Company's principal executive offices at 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Secretary, no later than August 23, 2023 (120 calendar days preceding the one-year anniversary of the date the proxy statement in connection with our 2022 annual meeting of shareholders (the "2022 Annual Meeting") was first mailed to our shareholders for the 2022 Annual Meeting), and must otherwise satisfy the procedures prescribed by Rule 14a-8 under the Exchange Act. It is suggested that any such proposals be submitted by certified mail with return receipt requested.

Pursuant to Rule 14a-4 under the Exchange Act, shareholder proxies obtained by our Board in connection with our Annual Meeting will confer on the proxies and attorneys-in-fact named therein discretionary authority to vote on any matters presented at such annual meeting which were not included in the Company's Proxy Statement in connection with such annual meeting, unless notice of the matter to be presented at such annual meeting is provided to the Company's Secretary before November 6, 2023 (the 45th day preceding the one-year anniversary of the date the proxy statement in connection with our 2022 Annual Meeting was first mailed to our shareholders for the 2022 Annual Meeting).

Director Nominations

Under the By-Laws, shareholders intending to nominate one or more candidates for election to our Board at our 2023 Annual Shareholder Meeting may do so only if written notice of the intent to make such nomination(s) has been given, either by personal delivery or by United States mail, postage prepaid, to the Secretary of the Company, at the Company's principal executive offices at 81 Executive Blvd. Suite 3, Farmingdale, New York 11735, Attention: Secretary, not less than ninety (90) days nor more than one hundred twenty (120) days prior to the earlier of the date of such annual meeting or the corresponding date on which the immediately preceding year's annual meeting of shareholders was held. Such notice must contain all of the information required by the By-Laws, including, without limitation, all information that would be required in connection with such nomination(s) under the SEC's proxy rules if such nomination were the subject of a proxy solicitation and the written consent of each nominee for election to our Board named therein to serve if elected. The chairman of the meeting may refuse to acknowledge the nomination of any person not made in compliance with the By-Laws.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC under the Exchange Act. Our filings with the SEC are available to the public from commercial document retrieval services and at the web site maintained by the SEC at www.sec.gov.

The SEC allows us to "incorporate by reference" information into this proxy statement, which means that we can disclose important information to you by referring you to other documents filed separately with the SEC. The information incorporated by reference is deemed to be part of this proxy statement, except for any information superseded by information in this proxy statement or incorporated by reference subsequent to the date of this proxy statement. This proxy statement incorporates by reference the documents set forth below that we have previously filed with the SEC. These documents contain important information about us and our financial condition and are incorporated by reference into this proxy statement.

The following filings with the SEC are incorporated by reference:

- Annual Report on Form 10-K for the fiscal year ended July 31, 2022, filed with the SEC on
 October 14, 2022, as amended by Amendment No. 1 to our Annual Report on Form 10-K/A for the
 fiscal year ended July 31, 2022, filed with the SEC on November 25, 2022;
- Quarterly Reports on Form 10-Q for the quarterly periods ended October 31, 2022 and January 31, 2023, filed with the SEC on December 12, 2022 and March 20, 2023, respectively;
- Definitive Proxy Statement on <u>Schedule 14A</u>, as supplemented, filed with the SEC on December 21, 2022; and
- Current Reports on Form 8-K filed with the SEC on October 20, 2022, November 4, 2022,
 February 3, 2023, March 16, 2023 and April 5, 2023 (other than the portions of such documents not deemed to be filed).

We also incorporate by reference into this proxy statement each additional document we may file under Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act between the date of this proxy statement and the earlier of the date of the Special Meeting or the termination of the Asset Purchase Agreement. These documents include periodic reports, such as Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q, as well as Current Reports on Form 8-K (other than current reports on Form 8-K furnished pursuant to Item 2.02 or Item 7.01 of Form 8-K, including any exhibits included with such information, unless otherwise indicated therein) and proxy solicitation materials. The information provided on our website is not part of this proxy statement, and therefore is not incorporated by reference herein.

YOUR VOTE IS IMPORTANT. We hope that you will attend the Special Meeting and look forward to your presence. HOWEVER, EVEN IF YOU PLAN TO ATTEND, YOU ARE URGED TO EITHER SUBMIT A PROXY FOR YOUR SHARES ELECTRONICALLY ON THE INTERNET, BY TELEPHONE OR BY COMPLETING, SIGNING AND RETURNING THE ENCLOSED PROXY CARD. If you wish to change your vote or vote differently in person, your proxy may be revoked at any time prior to the time it is voted at the meeting. IF YOUR SHARES ARE HELD OF RECORD BY A BROKER OR OTHER NOMINEE, AND YOU WISH TO VOTE AT THE SPECIAL MEETING, YOU MUST FOLLOW THE PROCEDURES PROVIDED BY THAT RECORD HOLDER.

If you have any questions about this proxy statement, the Special Meeting or the Asset Sale or need assistance with the voting procedures, you should contact:

Enzo Biochem, Inc.

81 Executive Blvd. Suite 3, Farmingdale, New York 11735

(631)-755-5500

Attn.: Investor Relations

or



Kingsdale Advisors

+1 (888) 518-1554 (toll-free in North America)

+1 (917) 341-5244 (call collect outside of North America)

e-mail: contactus@kingsdaleadvisors.com

Questions? Need Help Voting?

Please contact our Strategic Stockholder Advisor and Proxy Solicitation Agent, Kingsdale Advisors CONTACT US:



Kingsdale Advisors

+1 (888) 518-1554 (toll-free in North America)

+1 (917) 341-5244 (call collect outside of North America)

e-mail: contactus@kingsdaleadvisors.com

ASSET PURCHASE AGREEMENT

dated as of March 16, 2023

by and among

LABORATORY CORPORATION OF AMERICA HOLDINGS,

ENZO CLINICAL LABS, INC.

and

ENZO BIOCHEM, INC.

TABLE OF CONTENTS

		Annex A Page No.
ARTICLE I	DEFINITIONS	A-1
1.1	Definitions	A-1
1.2	Other Definitional Provisions	A-16
ARTICLE II	SALE AND PURCHASE OF ACQUIRED ASSETS	A-16
2.1	Transfer of Acquired Assets	A-16
2.2	The Closing	A-16
2.3	Purchase Price	A-16
2.4	Assumption of Liabilities	A-19
2.5	Nonassignable Items	A-19
2.6	Purchase Price Allocation	A-19
2.7	Certain Acknowledgments	A-20
2.8	Closing Deliverables	A-20
2.9	Other Documentation	A-21
2.10	Escrow	A-22
ARTICLE III	REPRESENTATIONS AND WARRANTIES OF SELLER AND SELLER PARENT	A-22
3.1	Organization, Standing and Power; No Subsidiaries	A-22
3.2	Authority	A-22
3.3	No Conflict; Required Filings and Consents	A-23
3.4	International Trade and Anti-Corruption	A-23
3.5	Legal Compliance	A-24
3.6	Absence of Changes	A-26
3.7	Customers, Tests, Payors; Suppliers and Supplies	A-26
3.8	Good Title to and Use of Assets	A-27
3.9	Real Property	A-27
3.10	Material Contracts	A-28
3.11	Bankruptcy	A-30
3.12	Prepaid Expenses	A-30
3.13	Insurance	A-30
3.14	Litigation; Governmental Orders	A-31
3.15	Brokers or Finders	A-31
3.16	Disqualifying Background	A-31
3.17	Privacy and Data Security; Information Technology	A-31
3.18	Tax Matters	A-32
3.19	Labor Matters	A-33
3.20	Telephone Numbers	A-34
3.21	WARN Act	A-34
3.22	Financial Statements	A-34
3.23	Motor Vehicles	A-35
3.24	Absence of Undisclosed Liabilities	A-35
3.25	Information Supplied	A-35

Annex A-i

		Annex A Page No.
3.26	Environmental Matters	A-35
3.27	Employee Benefits	A-36
3.28	Intellectual Property	A-36
3.29	Accounts Receivable; Accounts Payable	A-37
3.30	Affiliate Transactions	A-38
3.31	No Other Representations or Warranties	A-38
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF BUYER	A-38
4.1	Organization, Standing and Power	A-38
4.2	Authority	A-38
4.3	No Conflict; Required Filings and Consents	A-38
4.4	Brokers or Finders	A-39
4.5	Litigation	A-39
4.6	Information Supplied	A-39
4.7	No Other Representations or Warranties	A-39
ARTICLE V	CONDUCT OF BUSINESS PENDING THE CLOSING	A-39
5.1	Interim Operating Covenants	A-39
5.2	No Negotiation	A-41
5.3	Governmental Approvals and Consents	A-43
5.4	Information Technology; Assistance with Customer Transition	A-44
5.5	Retention of Records and Specimens	A-45
5.6	Preparation of Proxy Statement; Stockholder Meeting	A-46
5.7	Interim Period Operations	A-47
5.8	Assigned Locations	A-47
5.9	Transfer of Warranties	A-47
5.10	Seller Activities	A-47
5.11	Notice of Developments	A-47
ARTICLE VI	CLOSING CONDITIONS	A-48
6.1	Seller's and Seller Parent's Closing Conditions	A-48
6.2	Buyer's Closing Conditions	A-48
6.3	Conditions to Obligations of All Parties to Consummate the Closing	A-49
ARTICLE VII	ADDITIONAL AGREEMENTS	A-49
	ADDITIONAL AGREEMENTS	
7.1		4 40
7.1	Covenant to Satisfy Conditions	A-49
7.2	Confidentiality Agreement	A-49
7.3	Trading Prohibitions	A-50
7.4	Public Statements and Internal Communications	A-50
7.5	Taxes	A-50
7.6	Pre-Closing and Post-Closing Information; Employee Accessibility	A-51
7.7	Bulk Sales Laws	A-52
7.8	Reconciliation	A-52
7.9	Employees	A-52
7.10	Accounts Receivable	A-54

		Annex A Page No.
7.11	Data Room	A-54
7.12	R&W Policy	A-54
7.13	Buyer Non-Solicitation	A-54
7.14	Restrictive Covenants	A-54
7.15	Payor Arrangements	A-56
7.16	Transition Services	A-56
ARTICLE VIII	TERMINATION	A-56
8.1	Termination by Mutual Consent	A-56
8.2	Termination by Seller or Buyer	A-56
8.3	Termination by Buyer	A-57
8.4	Termination by Seller	A-57
8.5	Effect of Termination and Abandonment	A-57
8.6	Termination Fee	A-58
ARTICLE IX	INDEMNIFICATION	A-59
9.1	Indemnification by the Parties	A-59
9.2	Procedure for Indemnification Claims – Third Party Claims	A-60
9.3	Procedure for Indemnification Claims – Other Claims	A-61
9.4	Nature of Payments	A-61
9.5	Survival	A-61
9.6	Limitations on Liability; Order of Recourse; Recovery from Escrow Account	A-61
9.7	Calculation of Losses	A-63
9.8	Insurance Proceeds	A-63
9.9	Exclusive Remedies	A-63
ARTICLE X	GENERAL PROVISIONS	A-64
10.1	Notices	A-64
10.2	Interpretation	A-64
10.3	Rules of Construction	A-64
10.4	Entire Agreement	A-65
10.5	Jurisdiction; Governing Law	A-65
10.6	Waiver of Jury Trial	A-65
10.7	Severability	A-65
10.8	Equitable Relief	A-65
10.9	Expenses	A-66
10.10	Assignment	A-66
10.11	No Third Party Beneficiaries	A-66
10.12	Further Assurances and Records	A-66
10.13	Amendments and Waivers	A-66
10.14	Independence of Covenants	A-66
10.15	Attorney-Client Privilege and Conflict Waiver	A-66
10.16	Counterparts	A-67

SCHEDULES

Attached

DISCLOSURE SCHEDULES

Attached

EXHIBITS

Exhibit A Form of Assignment and Assumption of Lease

Exhibit B Form of Assignment, Assumption and Bill of Sale

Exhibit C Form of Escrow Agreement

Exhibit D Form of Intellectual Property Agreement

Annex A-iv

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "Agreement"), dated as of March 16, 2023 (the "Effective Date"), is entered into by and among Laboratory Corporation of America Holdings, a Delaware corporation ("Buyer"), Enzo Clinical Labs, Inc., a New York corporation ("Seller"), and Enzo Biochem, Inc., a New York corporation ("Seller Parent"). Buyer, Seller and Seller Parent are collectively herein referred to as the "Parties" and each individually as a "Party."

RECITALS

WHEREAS, Seller Parent, indirectly through Seller, is engaged in the business of performing diagnostic tests in its clinical and anatomical pathology laboratories and provides related laboratory testing support services, in each case, including phlebotomy, accessioning, processing, electronic data interface support, supply warehousing and distribution, logistics, customer service, and consumer services to support healthcare entities, providers, patients, and other non-healthcare entities, including municipalities, employer groups, schools, unions, and trade organizations, both in the diagnosis and management of acute and chronic disease as well as general non acute screening, including drug testing and COVID-19 surveillance testing (the "Business");

WHEREAS, Seller desires, and Seller Parent desires to cause Seller, to sell, assign and transfer, and Buyer desires to purchase, assume and accept from Seller, all of Seller's right, title and interest in and to certain assets related to the Business, on the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the board of directors of Seller Parent has (i) approved and declared advisable this Agreement and the transactions contemplated hereby on the terms and subject to the conditions set forth herein, (ii) declared and determined that it is in the best interests of the stockholders of Seller Parent that Seller Parent and Seller enter into this Agreement and consummate the transactions contemplated hereby on the terms and subject to the conditions set forth herein, (iii) directed that this Agreement be submitted to a vote at a meeting of the stockholders of Seller Parent for approval and adoption, and (iv) recommended that the stockholders of Seller Parent approve and adopt this Agreement and the transactions contemplated hereby; and

WHEREAS, contemporaneously with the execution of this Agreement, and as an inducement to the willingness of Buyer to enter into this Agreement, certain stockholders of Seller Parent (the "*Principal Stockholders*") are entering into an agreement with Buyer pursuant to which, among other things, the Principal Stockholders have agreed, subject to the terms thereof, to vote all shares of common stock of Seller Parent over which each has voting control in favor of adoption of this Agreement.

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants, agreements, representations, and warranties contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and incorporating the above recitals with and into this Agreement, the Parties hereby agree as follows:

ARTICLE I DEFINITIONS

1.1 <u>Definitions</u>. When used in this Agreement, these terms have these meanings:

(a) "Acceptable Confidentiality Agreement" means a confidentiality agreement that is either (i) in effect as of the execution and delivery of this Agreement and containing provisions that require any counterparty thereto (and any of its Affiliates and Representatives named therein) that receives material non-public information of or with respect to the Business to keep such information confidential, or (ii) executed, delivered and effective after the execution and delivery of this Agreement and containing provisions that require any counterparty thereto (and any of its Affiliates and Representatives named therein) that receives material non-public information of or with respect to the Business to keep such information confidential and containing confidentiality provisions, a "non-solicitation" provision and a "standstill" provision, in each case (x) no less favorable to Seller and Seller Parent, in the aggregate and in any material respect, than those set forth in the Confidentiality Agreement, and (y) permits the sharing of information by Seller and Seller Parent to Buyer in accordance with Section 5.2 of this Agreement.

(b) "Acquired Assets" means

- (1) the assets of Seller that are primarily used in the Business and not included in the Excluded Assets, including:
- (i) all customer accounts of the Business, all customer files and records, sales data, lists of courier routes, customer requirements, and ordering patterns used by the Business;
- (ii) copies or originals of any and all past and pending documents of sales and service information, billing records, sales and promotional literature (including marketing, advertising, promotional, sales, and training materials), payor lists, inventory cost records, quality control records and procedures, machinery and equipment records, mailing lists, purchase orders, customer records (including correspondence) and sample materials, in each case, in any form or medium, that are used or held for use in the operation or conduct of the Business;
- (iii) all of Seller's customer and vendor lists, pricing and cost information and customer relationships;
- (iv) all inventory and supplies used in connection with the operation of the Business as of the Closing Date and located at any Assigned Location, but including customer supplies of the Business located at Seller's site in Farmingdale, NY, and all applicable warranties against manufacturers or vendors but excluding inventory and supplies located at patient service centers that are not Assigned Locations:
- (v) (A) all other owned tangible personal property and equipment, including owned furniture, fixtures and laboratory equipment, used in connection with the operation of the Business as of the Closing Date and located at any Assigned Location (including, for the avoidance of doubt, the owned tangible personal property and equipment located at the Manhattan STAT Laboratory), (B) the leased or rented tangible personal property and equipment that is related to an Assigned Contract, and (C) the owned tangible personal property and equipment at Seller's site in Farmingdale, NY set forth on Schedule 1.1(b)(v), except in each case for that tangible personal property and equipment reasonably needed to provide the STAT Services (as reasonably determined by the Parties) and the tangible personal property and equipment located at patient service centers that are not Assigned Locations, and, in each case, all applicable warranties against manufacturers or vendors;
- (vi) all reserves, security and other deposits, advances and prepaid expenses and credits related to the Assigned Contracts or any other Acquired Assets, including those listed on Schedule 1.1(b)(vi);
- (vii) all orders arising from the Business (excluding the Standing Orders and Hold Orders of the Business prior to the Closing except as otherwise provided in Section 1.1(b)(xii));
- (viii) all Seller's rights to telephone and facsimile numbers primarily used in the operation of the Business, including those set forth on <u>Section 3.20 of the Disclosure Schedules</u>, to the extent transferable;
 - (ix) the Assigned Contracts;
- (x) all of Seller's claims, causes of action, choses in action, rights of recovery and rights of set-off of any kind against third parties relating to the Assigned Contracts arising from and after the Closing Date, choate or inchoate, known or unknown, contingent or non-contingent, and whether or not liquidated, in each case arising out of or related to events from and after the Closing Date;
- (xi) the benefit of and all rights to enforce any covenants, warranties, representations or guarantees under the Assigned Contracts (including covenants or warranties made by any Person in connection with the Assigned Contracts or services furnished to Seller affecting the Assigned Contracts), in each case, relating to matters arising from and after the Closing Date;
 - (xii) copies of the Books and Records;

- (xiii) all Material Permits used or held for use in connection with any Assigned Location or the operation of the Business in connection with any Assigned Location, including the applicable Material Permits set forth on Section 3.5(a)(i) of the Disclosure Schedules, to the extent transferable;
- (xiv) the motor vehicles used in connection with the Business, including those listed on <u>Section 3.23 of the Disclosure Schedules</u>, and all applicable warranties against manufacturers or vendors:
 - (xv) the laboratory-related IT Assets set forth on Schedule 1.1(b)(xv);
- (xvi) all accounts receivable and other receivables relating to the Business arising from services performed on and after the Closing Date, other than those accounts receivable and other receivables arising from Add-On Testing Services;
- (xvii) all goodwill and going concern value associated with the assets described in the foregoing clauses; and
- (2) the following assets of Seller and Seller Parent (including the Affiliates of Seller Parent) (collectively, the "Acquired Intellectual Property Assets"):
- (i) the Intellectual Property exclusively used in the Business including the trade names "Enzo Labs", "Enzo Clinical Labs", "GoTestMeNow" and variations thereof and all rights to the domain names "enzoclinicallabs.com", "enzodirect.com" and "gotestmenow.com" and variations thereof. For the avoidance of doubt, the Acquired Intellectual Property Assets shall not include any patents or any Intellectual Property of any type in or to any products, including reagents, probes, antibodies, dyes, or the GENFLEX equipment that, prior to the Closing, were manufactured or otherwise procured for the Business by Seller Parent or its subsidiaries.
- (c) "Action" means any action, complaint, claim, suit, litigation, audit, investigation or other legal proceeding at law or in equity by or before, or otherwise involving, any Governmental Authority.
 - (d) "Add-On Testing Services" shall have the meaning set forth in Section 7.14(a).
 - (e) "Adverse Recommendation Change" shall have the meaning set forth in Section 5.2(d).
- (f) "Affiliate" means, with respect to any specified Person, a Person, directly or indirectly, through one or more intermediaries, controlled by, under common control with, or controlling such Person. For purposes of this definition, "control" of a Person means the possession, direct or indirect, of the power to (i) vote more than 50% of the equity interests of such Person, (ii) elect more than 50% of the members of the board of directors (or other similar governing body) of such Person, or (iii) direct or cause the direction of the management or policies of such Person, whether through the ownership of voting securities, by contract or otherwise.
 - (g) "Aggregate Ticking Fee" shall have the meaning set forth in Section 2.3(e).
- (h) "Aggregate Ticking Fee Closing Payment" shall mean an amount equal to the following percentages of the Aggregate Ticking Fee paid on the Closing Date:
 - (i) If the Closing Date is on or prior to the Final Transaction Expiration Date, (A) if the HSR Clearance Date occurred prior to the Shareholder Approval Date, 50% of the Aggregate Ticking Fee paid from the Shareholder Approval Date to the Closing Date, or (B) if the HSR Clearance Date occurred after the Shareholder Approval Date, 100% of the Aggregate Ticking Fee paid for the period from the Shareholder Approval Date to the HSR Clearance Date, and 50% of the Aggregate Ticking Fee paid from the HSR Clearance Date to the Closing Date.
 - (ii) If the Closing Date is after the Final Transaction Expiration Date, (A) if the HSR Clearance Date occurred prior to the Shareholder Approval Date, 50% of the Aggregate Ticking Fee paid from the Shareholder Approval Date to the Final Transaction Expiration Date (if any) and 100% of the Aggregate Ticking Fee paid thereafter, or (B) if the HSR Clearance Date occurred after the Shareholder Approval Date, 100% of the Aggregate Ticking Fee paid for the period from the Shareholder Approval Date to the HSR Clearance Date (or the Final Transaction Expiration Date, if earlier), 100% of the Aggregate Ticking Fee paid from the Final Transaction Expiration Date to the HSR Clearance Date (if the

HSR Clearance Date occurs after the Final Transaction Expiration Date), 50% of the Aggregate Ticking Fee paid from the HSR Clearance Date to the Final Transaction Expiration Date (if the Final Transaction Expiration Date occurs after the HSR Clearance Date), 50% of the Aggregate Ticking Fee paid from the HSR Clearance Date to the Closing Date (if the HSR Clearance Date occurs after the Final Transaction Expiration Date), 100% of the Aggregate Ticking Fee paid from the Final Transaction Expiration Date to the Closing Date (if the Final Transaction Expiration Date occurs after the HSR Clearance Date) and 100% of the Aggregate Ticking Fee paid from the Shareholder Approval Date to the Closing Date (if both the Shareholder Approval Date and the HSR Clearance Date occur after the Final Transaction Expiration Date).

- (i) "Agreement" shall have the meaning set forth in the preamble hereto.
- (j) "Alternative Transaction" shall have the meaning set forth in Section 5.2(a).
- (k) "Amended STAT Services Agreement" means, that certain STAT Services Agreement between Seller and Buyer, dated as of October 17, 2022, as amended by the STAT Services Agreement Amendment.
 - (1) "Anti-Corruption Law" shall have the meaning set forth in Section 3.4(c).
 - (m) "Anti-Kickback Statute" shall have the meaning set forth in Section 3.5(g).
 - (n) "Apportioned Obligations" shall have the meaning set forth in Section 7.5(b).
- (o) "Assigned Contracts" means, collectively, (i) each Assigned Real Property Lease and (ii) each Contract set forth on Schedule 1.1(o).
 - (p) "Assigned Locations" shall have the meaning set forth in Section 5.8.
- (q) "Assigned Real Property Leases" shall be the Real Property Leases for the Assigned Locations as such Real Property Leases may be extended, renewed or amended pursuant to the Lease Extensions.
 - (r) "Assigned Real Property List" shall have the meaning set forth in Section 5.8.
- (s) "Assignment and Assumption of Leases" means the Assignment and Assumption of Lease Agreements, in the forms set forth on Exhibit A hereto.
- (t) "Assignment, Assumption and Bill of Sale" means an Assignment, Assumption and Bill of Sale, in the form set forth on Exhibit B hereto.
- (u) "Assumed Liabilities" means the Liabilities under the Assigned Contracts arising on or after the Closing Date, but excluding any Liabilities arising on or after the Closing Date that are based on facts, circumstances, or occurrences arising prior to the Closing Date, or relating to any breach, violation or failure to perform that occurred prior to the Closing Date. In no event shall Assumed Liabilities include any Excluded Liabilities.
- (v) "BAAs" means business associate agreements, as described under HIPAA at 45 C.F.R. §§ 164.502(e) and 164.504(e).
 - (w) "Balance Sheet Date" shall have the meaning set forth in Section 3.22(a).
 - (x) "Board Recommendation" shall have the meaning set forth in Section 3.2(b).
- (y) "Books and Records" means all books and records of Seller and Seller Parent exclusively relating to the Business (excluding all patient information and other information that is not reasonably necessary to operate the Business) but specifically including copies of all records with respect to Standing Orders and Hold Orders for patients of the Business and any orders that are placed but not performed prior to the Closing.
 - (z) "Business" shall have the meaning set forth in the Recitals.
- (aa) "Business Day" means any day other than (i) a Saturday, Sunday or federal holiday or (ii) a day on which commercial banks in New York are required to or may be closed.

- (bb) "Business Employee" means each employee of Seller who spends the majority of their time working on the Business or who spends a majority of their time supporting the Business.
 - (cc) "Business Intellectual Property" shall have the meaning set forth in Section 3.28(b).
- (dd) "Business Service Provider" means any employee, contractor, subcontractor, Representative, consultant or agent of Seller or Buyer or any of their respective Affiliates and who has acted in such capacity at any time within the 12-month period immediately preceding the date of hire, recruitment, solicitation, or retention; provided, however, that independent consultants and agents who have provided professional services to Seller shall not be deemed a Business Service Provider, and no Person shall be deemed a Business Service Provider of Seller or its Affiliates except to the extent such Person's engagement pertained to the Business.
 - (ee) "Buyer" shall have the meaning set forth in the preamble hereto.
 - (ff) "Buyer Group" means Buyer and its agents and Representatives.
 - (gg) "Buyer Indemnified Parties" shall have the meaning set forth in Section 9.1(a).
 - (hh) "Cap" shall have the meaning set forth in Section 9.6(a)(ii).
 - (ii) "CLIA" shall have the meaning set forth in Section 3.5(a).
 - (jj) "Client Setup Data" shall have the meaning set forth in Section 3.7(a).
 - (kk) "Closing" shall have the meaning set forth in Section 2.3(a).
 - (ll) "Closing Date" shall have the meaning set forth in Section 2.3(a).
- (mm) "Code" means the Internal Revenue Code of 1986, as amended or supplemented from time to time.
- (nn) "Competition Laws" means the HSR Act, the Sherman Act, the Clayton Act, the Federal Trade Commission Act, and any other federal, state, local or non-United States statutes, rules, regulations, orders, decrees, administrative or judicial doctrines or other Laws, each as amended from time to time, that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization, lessening of competition or restraint of trade.
- (00) "Confidential Information" means confidential information regarding the Business or the Acquired Assets or information of Buyer that is not generally known or generally available to the public, and which shall include the following with respect to the Business, the Acquired Assets, Buyer or Seller (solely with respect to the Business and the Acquired Assets), as applicable, whether written or not: (i) information regarding operations, assets, liabilities or financial condition; (ii) information regarding bidding, quotations, price, sales, profits, merchandising, marketing and promotions (including marketing strategies and concepts), advertising campaigns, capital expenditures, costs, joint ventures, business alliances, products, services or purchasing; (iii) information regarding the terms, conditions and relationship the Business, Buyer or Seller has with Business Service Provider (other than employees), including their identities, responsibilities, qualifications, benefits, compensation and files; (iv) customer lists, databases and other information and reports related to current or prospective customers, including, in each case, information regarding their identities, contact persons and purchasing patterns; (v) information regarding current or prospective vendors, suppliers, distributors or other business partners; and (vi) forecasts, projections, budgets, strategies, and business plans. Additionally, "Confidential Information" shall include all of the terms and conditions of this Agreement. Notwithstanding the foregoing, "Confidential Information" shall not include information which (A) as of the date of disclosure is already known to the receiving party, (B) is or becomes generally known or generally available to the public through no act or fault of Seller or any of its Affiliates, (C) is lawfully disclosed to the receiving party by a third party which is not known to the receiving party to be obligated to maintain and preserve such information as confidential and proprietary, or (D) is independently developed by or for the receiving party without the benefit of any Confidential Information.
 - (pp) "Confidentiality Agreement' shall have the meaning set forth in Section 7.2(a).

- (qq) "Consents" shall mean any consents, notices, novations, approvals, authorizations, licenses, certificates, registrations, permissions, requirements, waivers, exemptions, leases, subleases, permits and agreements (or modifications or amendments thereto) required in order to transfer or assign any Acquired Asset.
- (rr) "Contract" shall mean any legally binding agreement, contract, lease (including the Real Property Leases), note, indenture, mortgage, loan, evidence of Debt, purchase or sale order, letter of credit, franchise agreement, employment agreement, license, sublicense, instrument or other written commitment or arrangement, and all amendments, supplements or modifications thereto. For purposes of this Agreement, the term "Contract" shall include any series of related Contracts.
 - (ss) "CPT" shall have the meaning set forth in Section 3.22(c).
- (tt) "Customers" means those customers with respect to the Business who ordered testing services from Seller at any time during the 24 months prior to the Closing.
 - (uu) "Daily Ticking Fee" shall have the meaning set forth in Section 2.3(e).
 - (vv) "Data Room" shall have the meaning set forth in Section 7.11.
 - (ww) "DEA" shall have the meaning set forth in Section 3.5(g).
- (xx) "Debt" means, with respect to any Person at any date, without duplication: (i) all obligations (including the principal amount thereof and, if applicable, the accreted amount thereof and the amount of accrued and unpaid interest thereon) for borrowed money or in respect of loans or advances, whether short term or long term, whether or not represented by a bond, debenture, note, or other security instrument and whether or not convertible into any other security or instrument, (ii) all obligations evidenced by bonds, debentures, notes, swap agreements or other similar instruments or debt securities, (iii) any deferred obligation for the payment of the purchase price of any property or other asset purchased (other than current accounts payable that were incurred in the Ordinary Course of Business), (iv) all obligations in respect of letters of credit and bankers' acceptances issued for the account of such Person, (v) all obligations arising from cash/book overdrafts or negative cash balances, (vi) all obligations arising from deferred compensation arrangements or employee bonuses (whether accrued or not), (vii) all obligations secured by a Lien (other than Permitted Liens) on the property or other assets of such Person (whether or not debt secured thereby has been assumed), (viii) all capital lease obligations determined in accordance with GAAP, (ix) notes and accounts payable to any Affiliates of such Person (and in the case of Seller or Seller Parent, to any of their respective stockholders) or any officers or employees of such Person, (x) all guarantees of such Person in connection with any of the foregoing, and (xi) all accrued interest, prepayment premiums or penalties related to any of the foregoing.
- (yy) "Debt Amount" means all Debt that shall be paid off at the Closing pursuant to Section 2.8(a)(iv) provided that the Debt Amount may not exceed the total amount of the Purchase Price before the exclusion of the Debt Amount.
 - (zz) "Deductible" shall have the meaning set forth in Section 9.6(a)(i).
 - (aaa) "Disclosure Schedules" shall have the meaning set forth in the preamble to Article III.
 - (bbb) "Disputed Items" shall have the meaning set forth in Section 2.3(c).
 - (ccc) "DOJ" shall have the meaning set forth in Section 5.3(a).
 - (ddd) "Draft Purchase Price Allocation" shall have the meaning set forth in Section 2.6.
 - (eee) "Effective Date" shall have the meaning set forth in the preamble hereto.
- (fff) "Employee Plan" means any plan, program, agreement, policy or arrangement, whether or not reduced to writing, and whether covering a single individual or a group of individuals, that is (i) a welfare plan within the meaning of Section 3(1) of ERISA, (ii) a pension benefit plan within the meaning of Section 3(2) of ERISA, (iii) a stock bonus, stock purchase, stock option, restricted stock, stock appreciation right, profit sharing or similar equity-based plan or agreement, or (iv) any other deferred-compensation, retirement, severance, retention, change-in-control, leave, vacation, welfare-benefit, bonus, incentive or fringe-benefit plan, program, agreement or arrangement maintained for the benefit of any Business Employee, officer, director or consultant of the Business.

- (ggg) "Environmental Law" means all Laws relating to pollution or the protection of the environment or (with respect to exposure to Hazardous Substances) protection of human health or safety, including all those relating to the use, generation, recycling, processing, labeling, production, manufacture, remediation, treatment, storage, transportation, handling, exposure to, disposal, Release or threatened Release of Hazardous Substances.
- (hhh) "ERISA" means the federal Employee Retirement Income Security Act of 1974, as amended.
- (iii) "Escrow Account" means the account into which the Escrow Amount is deposited with the Escrow Agent and held by it, subject to disbursement as provided in Section 9.6 and the Escrow Agreement.
- (jjj) "Escrow Agent" means Western Alliance Bank, an Arizona corporation, as escrow agent under the Escrow Agreement, or any successor Person appointed in accordance with the terms of the Escrow Agreement.
- (kkk) "Escrow Agreement" means the Escrow Agreement in substantially the form attached hereto as Exhibit C.
 - (111) "Escrow Amount" means \$5,000,000.
- (mmm) "Escrow Fund" means the Escrow Amount deposited with the Escrow Agent in the Escrow Account, as such amount may increase or decrease from time to time as provided in this Agreement or the Escrow Agreement.
 - (nnn) "Estimated Prepaid Amount" shall have the meaning set forth in Section 2.3(b).
- (000) "Estimated Vacation Rollover Amount" shall have the meaning set forth in Section 2.3(b).
 - (ppp) "Exchange Act" shall have the meaning set forth in Section 3.3(c).
 - (qqq) "Excluded Accounts" shall have the meaning set forth in Section 3.5(c).
 - (rrr) "Excluded Assets" means the following assets of Seller or its Affiliates:
 - (i) cash and cash equivalents;
 - (ii) fee title interest in and to any real property and leasehold interests in any real property that are not Assigned Locations;
 - (iii) all other owned and leased tangible personal property and equipment that are not Acquired Assets or are otherwise primarily used in the Excluded Business;
 - (iv) the minute books, stock records and corporate seal, as applicable, of Seller and Seller Parent;
 - (v) insurance policies and rights thereunder (except insurance benefits, including rights and proceeds, arising from or relating to the Acquired Assets prior to the Closing Date);
 - (vi) accounts or notes receivable arising out of the operation of the Business prior to the Closing, specimens processed by Seller prior to the Closing and the Add-On Testing Services (and excluding for avoidance of doubt, those accounts receivable or notes receivable arising from Standing Orders, Hold Orders and future orders arising from the Business after the Closing);
 - (vii) all rights related to and assets of any benefit plans;
 - (viii) all Contracts that are not Assigned Contracts;
 - (ix) all rights to refunds of Taxes of Seller or its Affiliates, including (A) any refunds of Taxes relating to the operation of the Business or use, ownership or operation of the Acquired Assets attributable to any period (or portion thereof) ending on or prior to the Closing, and (B) subject to Sections 7.5(a) and 7.5(b), any refunds of Taxes relating to the sale of the Acquired Assets pursuant to this Agreement;

- (x) all equity interests in any Person;
- (xi) subject to the Intellectual Property Agreement, all Intellectual Property and intangible property used in or held on behalf of the Excluded Business, including the name "Enzo" and any derivative thereof:
- (xii) any and all proprietary diagnostics platforms and products of Seller Parent or its Affiliates that are not assets of the Business, including the GENFLEX® and Ampiprobe platforms;
 - (xiii) all rights in and to Seller's LIS;
- (xiv) except as expressly set forth in the Intellectual Property Agreement, the right to use the name "Enzo" or any derivative thereof in any business other than the Business;
- (xv) all other owned or leased tangible personal property and equipment that are not Acquired Assets, including those reasonably needed to provide the services under the Amended STAT Services Agreement (the "STAT Services") (as reasonably determined by the Parties) or located at Leased Real Properties that are not Assigned Locations, including any applicable patient service centers and Seller's Farmingdale, NY site (except for the owned tangible personal property and equipment at Seller's site in Farmingdale, NY set forth on Schedule 1.1(b)(v));
- (xvi) all records, films, slides, blocks, tissue and other specimens maintained by Seller as of the Closing Date with respect to the Business;
- (xvii) all inventory and supplies located at Leased Real Properties that are not Assigned Locations, including any applicable patient service centers and Seller's Farmingdale, NY site; provided that for the avoidance of doubt customer supplies located at Seller's Farmingdale, NY site shall be Acquired Assets; and
 - (xviii) any security, claim, remedy or other right related to any of the foregoing.
- (sss) "Excluded Business" means the businesses (other than the Business) of Seller Parent or any of its Affiliates as conducted on the Effective Date, including diagnostic testing services for lab-to-lab and CROs provided by Seller Parent on the Effective Date to the extent related to research, development, marketing and selling research tools, diagnostics and therapeutics, and product and platform validation for commercial laboratory and health system customers of the Enzo Life Sciences and Enzo Therapeutics businesses of Seller Parent as provided by Seller Parent on the Effective Date. For avoidance of doubt, the Excluded Business shall not include the provision of clinical reference laboratory services, diagnostic or testing services or home testing kits.
- (ttt) "Excluded Liabilities" means any Liability of Seller or its Affiliates or otherwise arising from or related to the Business or the Acquired Assets of any kind or nature whatsoever that is not an Assumed Liability including the following Liabilities: (i) any Liabilities arising out of, related to, or in connection with Seller's or any other Person's use, ownership or operation of the Business or the Acquired Assets prior to the Closing (for the avoidance of doubt, any Liabilities arising out of, related to, or in connection with Buyer's or any other Person's use, ownership or operation of the Business or the Acquired Assets on or after the Closing shall be Liabilities of Buyer); (ii) any Liability for claims or other Actions of, or Liens imposed on the Acquired Assets or any other assets of Buyer by, creditors of Seller or its Affiliates arising as a result of the transactions contemplated by this Agreement, whether arising prior to, at or after the Closing; (iii) any Liability of Seller Parent, Seller or their Affiliates for Taxes arising on or prior to the Closing, including, (A) subject to Section 7.5(a), any Liability for Taxes arising as a result of the operation of the Business or use, ownership or operation of the Acquired Assets attributable to any period (or portion thereof) ending on or prior to the Closing, and (B) subject to Section 7.5(b), any Taxes that arise as a result of the sale of the Acquired Assets pursuant to this Agreement; (iv) any Liabilities of Seller or its Affiliates arising or incurred in connection with the negotiation, preparation, investigation and performance of this Agreement, the other Transaction Documents and the transactions contemplated hereby and thereby, including fees and expenses of counsel, accountants, consultants, advisers and others; (v) any Liabilities relating to or arising out of the Excluded Assets; (vi) any Liability arising out of Seller's or Seller Parent's obligations under or agreements with any Governmental Programs (including pursuant to any Provider Agreements) arising from the operation of the Business on or prior to the Closing Date; (vii) any Liability for laboratory testing results, any laboratory or medical malpractice claims, or any violation of or non-compliance with any Law arising from the operation of the Business

prior to the Closing Date; (viii) any Liabilities for any present or former employees, officers, directors, retirees, independent contractors or consultants of Seller or Seller Parent arising from events occurring on or prior to the Closing Date, including any Liabilities associated with any claims for wages or other benefits, bonuses, accrued vacation, workers' compensation, severance, retention, termination or other payments; (ix) any Liabilities under (A) any Contracts that are not Assigned Contracts, (B) subject to Section 2.5, Assigned Contracts to the extent arising prior to the Closing Date or (C) any Contracts (1) which, subject to Section 2.5, are not validly and effectively assigned to Buyer pursuant to this Agreement, or (2) to the extent such Liabilities arise out of or relate to a breach by Seller of such Contracts prior to the Closing; (x) any Liabilities arising out of, in respect of or in connection with the failure by Seller or any of its Affiliates to comply, in any respect, with its the certificate of incorporation, bylaws or other governing documents, in each case as amended or restated, or any Law or Governmental Order in connection with or on or prior to the Closing or any Action related thereto; (xi) any Liabilities related to infringement of third party intellectual property rights prior to the Closing; (xii) any Liabilities arising out of, or in respect of or in connection with the Enzo Direct Portal or the GoTestMeNow Portal; (xiii) any Liabilities arising out of, in respect of or in connection all records, films, slides, blocks, tissue and other specimens maintained by Seller with respect to the Business; and (xiv) any Liabilities related to matters set forth on Schedule 1.1(ttt).

(uuu) "Expense Reimbursement" shall have the meaning set forth in Section 8.6(b).

(vvv) "Export and Import Law" means (i) the Arms Export Control Act (22 U.S.C. 2778), the International Traffic in Arms Regulations (ITAR) (22 CFR Parts 120-130), the Export Control Reform Act of 2018, as amended, the Export Administration Regulations (EAR) (15 CFR Parts 730-774), the Laws administered by Customs and Border Protection (19 CFR Parts 1-199) or any other United States Laws regulating exports, imports or re-exports to or from the United States, including the export or re-export of goods, services or technical data from the United States; and (ii) any Laws of a foreign Governmental Authority regulating exports, imports or re-exports to or from such foreign country, including the export or re-export of any good, service or technical data.

(www) "FD&C Act" shall have the meaning set forth in Section 3.5(g).

(xxx) "FDA Law and Regulation" shall have the meaning set forth in Section 3.5(g).

(yyy) "Final Prepaid Amount" shall have the meaning set forth in Section 2.3(c).

(zzz) "Final Purchase Price Allocation" shall have the meaning set forth in Section 2.6.

(aaaa) "Final Transaction Expiration Date" means the date that is 120 days from the date of this Agreement.

(bbbb) "Financial Statements" shall have the meaning set forth in Section 3.22(a).

(cccc) "Foreign Official" means: (i) any officer or employee of a foreign Governmental Authority or any department, agency or instrumentality thereof (including a state-owned or state-controlled entity); (ii) any officer or employee of a public international organization; (iii) any Person acting in an official capacity for or on behalf of any such foreign Governmental Authority or department, agency or instrumentality thereof, or for or on behalf of any public international organization or any political party; or (iv) any party official or candidate of any party, excluding, in each case, any official of the government of the United States.

(dddd) "Fraud" means Delaware common law fraud in the making of the representations and warranties set forth in this Agreement.

(eeee) "FTC" shall have the meaning set forth in Section 5.3(a).

(ffff) "Fundamental Representations" shall have the meaning set forth in Section 9.5.

(gggg) "GAAP" means generally accepted accounting principles in the United States.

(hhhh) "Governmental Authority" means any United States federal, foreign, state or local government, or political subdivision thereof, or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body, and includes any contractor (e.g., a Medicare administrative contractor) acting on behalf of a Governmental Authority.

- (iiii) "Governmental Order" means any order, writ, judgment, doctrine, requirement, ruling, assessment, injunction (whether temporary, preliminary or permanent), decree, stipulation, determination or award entered by or with any Governmental Authority.
 - (jjjj) "Governmental Programs" shall have the meaning set forth in Section 3.5(d).
 - (kkkk) "Guaranteed Obligations" shall have the meaning set forth in Section 9.6(e).
- (Illl) "Hazardous Substances" means (i) any substance, material or waste listed, defined or regulated due to its deleterious or as a "chemical substance," "hazardous substance," "hazardous material" or words of similar meaning and regulatory effect under any applicable Environmental Law, and (ii) petroleum or any fraction thereof, asbestos or asbestos-containing materials, polychlorinated biphenyls, radioactive materials and per- and poly-fluoroalkyl substances.

(mmmm) "HIPAA" means, collectively, the Health Insurance Portability and Accountability Act of 1996 and its implementing regulations, as amended and supplemented by the Health Information Technology for Economic and Clinical Health Act of the American Recovery and Reinvestment Act of 2009, Pub. Law No. 111-5 and its implementing regulations, when each is effective and as each is amended from time to time.

(nnnn) "Hired Employee" shall have the meaning set forth in Section 7.9(a).

(0000) "Hold Order" means an order for Testing that has been placed with Seller for which the patient has not been drawn.

(pppp) "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and the rules and regulations promulgated thereunder.

(qqqq) "HSR Clearance Date" means the date on which the waiting period under the HSR Act shall have expired or been terminated.

- (rrrr) "Indemnitee" shall have the meaning set forth in Section 9.2(a).
- (ssss) "Indemnitor" shall have the meaning set forth in Section 9.2(a).
- (tttt) "Independent Accountant" shall have the meaning set forth in Section 2.3(c).

(uuuu) "Information Privacy and Security Laws" mean all requirements of Law concerning the Processing of Personal Information applicable to the Business, including, where so applicable, HIPAA, state data protection and breach notification laws, and state social security number protection laws, the Federal Trade Commission Act, the Gramm-Leach-Bliley Act (15 U.S.C. §§ 6801 – 6809), the Fair Credit Reporting Act and Fair and Accurate Credit Transactions Act of 2003 (15 U.S.C. §§ 1681 – 1681x), and state consumer protection laws.

(vvvv) "Insurance Policies" shall have the meaning set forth in Section 3.13.

(wwww) "Intellectual Property" means patents; trademarks and other similar designations of source or origin, trade names, corporate names, service marks, trade dress rights, internet domain names and goodwill associated with the foregoing; copyrights (including computer programs, and software, client software platforms including mobile applications therefor, intellectual property rights in information management systems, compilations, databases, moral rights and rights of attribution and integrity); trade secrets and all other confidential information, market know-how, inventions, proprietary processes and procedures, forms, catalogs, guides, manuals, formulae, models, and business methods, and registrations and applications for registration of any of the foregoing.

(xxxx) "Intellectual Property Agreement" means the Intellectual Property Agreement, in substantially the form set forth in $\underline{\text{Exhibit D}}$ hereto.

(yyyy) "Interface Configuration" shall have the meaning set forth in Section 5.4(b).

(ZZZZ) "Interim Period" shall have the meaning set forth in Section 5.1.

(aaaaa) "IOP" means an in-office phlebotomist.

(bbbbb) "IRS" means the Internal Revenue Service.

Annex A-10

(cccc) "IT Assets" means all of Seller's computer systems, including software (including programs and applications, object and source code, databases, algorithms, and documentation), hardware, databases, firmware, middleware and platforms, interfaces, systems, networks, information technology equipment, facilities, websites, infrastructure, workstations, switches, data communications lines and associated documentation used by or on behalf of Seller in connection with the products, services and operation of the Business.

(ddddd) "IT Designee" shall have the meaning set forth in Section 5.4(c).

(eeeee) "Knowledge of Seller" or words of similar import means any of (i) Hamid Erfanian; (ii) Kara Cannon; (iii) Patricia Eckert; (iv) Rich Paladino; and (v) Matthew Kupferberg (A) is, or at any time has been, actually aware of such fact or matter or (B) reasonably should have discovered or otherwise become aware of such fact or matter in the course of performing his or her duties as an officer of Seller or Seller Parent, as applicable, during such officer's term of office with Seller or Seller Parent, as applicable.

(fffff) "Laws" means all applicable foreign, national, federal, state, county, city, municipal, local, administrative and other laws, statutes, codes or treaties, common law rulings or regulations, ordinances, orders, rules and regulations.

(ggggg) "Lease Extension" has the meaning set forth in Section 5.8.

(hhhhh) "Leased Real Property" has the meaning set forth in Section 3.9(a).

(iiiii) "Leave of Absence Employees" shall have the meaning set forth in Section 7.9(a).

(jjjjj) "Liability" or "Liabilities" means, with respect to any Person, any Debt, liability or obligation of such Person of any kind, character or description, whether known or unknown, absolute or contingent, secured or unsecured, joint or several, disputed by a third party or undisputed, accrued or unaccrued, due or to become due in the future, vested or unvested, liquidated or unliquidated, executory, determined, determinable or otherwise and whether or not the same is required to be accrued on the financial statements of such Person (including that arising out of any Contract or tort, whether based on negligence, strict liability or otherwise).

(kkkkk) "Licenses" shall have the meaning set forth in Section 3.28(a).

(IllII) "Lien" means, with respect to any given assets or properties, all encumbrances, mortgages, defects of title, deeds of trust, security agreements and security interests, pledges, liens, conditional sales agreements, claims, restrictions, charges, preemptive rights, hypothecations, assignments, options, leases, subleases, encroachments, covenants, easements or rights of third parties of any kind and character arising or existing by operation of law, by judicial decree, judgment or arbitral decision, by Contract or otherwise, whether or not accrued or fixed, absolute or contingent, known or unknown, determined or determinable and whenever arising, including those evidenced by Contracts, deeds of trust, and security, conditional sale and other title retention agreements.

(mmmmm) "LIS" means laboratory information systems.

(nnnnn) "LIS Project" shall have the meaning set forth in Section 5.4(c).

(00000) "Losses" means any and all Liabilities, claims, demands, losses, damages, bonds, dues, assessments, fines, interest, penalties, Taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement), expenses (including any remediation costs) or amounts paid in settlement (in each case, including reasonable attorneys' and experts' fees and expenses).

(ppppp) "Manhattan STAT Laboratory" means the STAT laboratory operated by Seller on the Effective Date at the Assigned Location located at 44 East 67th Street, Suite 0101 (a/k/a Suite 1A), New York, NY 10022.

(qqqqq) "Material Adverse Effect" means any event, occurrence, fact, condition, change, effect, or circumstance that, individually or in the aggregate, has a material adverse effect on (i) the Business, or the results of operations, financial condition, assets, or operations of the Business or (ii) the Acquired Assets; provided, however, that in no event shall any event, occurrence, fact, condition, change, effect, or circumstance resulting from or arising out of any of the following, either alone or in combination, be taken into account when determining whether a Material Adverse Effect has occurred or may, would or could occur under the foregoing clauses: (A) (x) general

economic conditions in the United States or the world as a whole or (y) general conditions in the capital or financial markets in the United States or the world as a whole; (B) changes in applicable Laws or GAAP (or the interpretation thereof); (C) political or social conditions (whether or not pursuant to the declaration of a national emergency or war), acts of war, terrorism or sabotage in the United States or any other country or region in the world (or any escalation with respect thereto); (D) earthquakes, hurricanes, tsunamis, tornadoes, floods, wild fires or other natural disasters or calamities; (E) epidemics, pandemics or disease outbreaks and other force majeure events; (F) the negotiation, execution, announcement or pendency of the transactions contemplated by this Agreement, including the impact thereof on relationships, contractual or otherwise, with, or actual or potential loss or impairment of, customers, clients, suppliers, landlords, distributors, partners or employees, or on revenue, profitability and cash flows; (G) compliance with the terms of, or the taking of any action (or refraining to take any action) required by, this Agreement; (H) the taking of any action (or failing to take any action) at the request of Buyer; (I) any actions required to be taken under applicable Laws; or (J) any failure by the Business, or Seller or its Affiliates with respect to the Business, to meet financial forecasts, projections or estimates (but not the reasons underlying such failure unless such reasons are covered by one or the other clauses of this definition); provided, in the case of each of clauses (A), (B), (C), (D) and (E), that such event, occurrence, fact, condition, change, effect, or circumstance may be taken into account when determining whether a Material Adverse Effect has occurred to the extent that such event, occurrence, fact, condition, change, effect, or circumstance has a disproportionate impact on the Business relative to other companies in the industry and geographic location in which the Business operates.

(rrrrr) "Material Contracts" shall have the meaning set forth in Section 3.10(e).

(sssss) "Material Customers" shall have the meaning set forth in Section 3.7(a).

(ttttt) "Material Payor" shall have the meaning set forth in Section 3.7(a).

(uuuuu) "Material Permits" shall have the meaning set forth in Section 3.5(a).

(vvvvv) "Material Vendors" shall have the meaning set forth in Section 3.7(b).

(wwww) "Medicare and Medicaid Programs" shall have the meaning set forth in

Section 3.5(d)

(xxxxx) "Nonassignable Items" shall have the meaning set forth in Section 2.5.

(yyyyy) "Nonsolicit Period" shall have the meaning set forth in Section 7.13.

(ZZZZZ) "NY Guidelines" shall have the meaning set forth in Section 3.7(c).

(aaaaaa) "NYSE" shall have the meaning set forth in Section 3.3(c).

(bbbbbb) "OFAC" means the Office of Foreign Assets Control.

(ccccc) "Offered Employee" shall have the meaning set forth in Section 7.9(a).

(dddddd) "Ordinary Course of Business" means the ordinary and usual course of normal dayto-day operations of the Business, as conducted by Seller through the Effective Date materially consistent with past custom and practice.

(eeeeee) "OSHA" shall have the meaning set forth in Section 3.5(g).

(ffffff) "Outside Date" shall have the meaning set forth in Section 8.2(a).

(gggggg) "Parent Plan" means each plan, program, policy, agreement or other arrangement covering current or former employees, directors or consultants, that is (i) an employee welfare plan within the meaning of Section 3(1) of ERISA (whether or not subject to ERISA), (ii) an employee pension benefit plan within the meaning of Section 3(2) of ERISA (whether or not subject to ERISA), including any plan which is a "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA), (iii) a stock option, stock purchase, stock appreciation right or other stock-based agreement, program or plan, (iv) an individual severance, retention, transaction bonus or other similar Contract, or (v) a bonus, incentive, deferred compensation, profit-sharing, retirement, post-retirement, vacation, severance or termination pay, benefit or fringe-benefit plan, program, policy, agreement or other arrangement, in each case that is sponsored, maintained or contributed to by Seller, Seller Parent

or any of its or their subsidiaries or to which Seller, Seller Parent or any of its or their subsidiaries contributes or is obligated to contribute to, and in each case providing benefits to any Business Employee, other than any plan, program, policy, agreement or arrangement mandated by applicable Law that is sponsored, maintained or administered by a Governmental Authority, including any workers' compensation plan, program or arrangement.

(hhhhhh) "Party" or "Parties" shall have the meanings set forth in the preamble hereto.

(iiiiii) "Payoff Letter" shall have the meaning set forth in Section 2.8(a)(iv).

(jjjjjj) "Payor Claims" shall have the meaning set forth in Section 3.5(d).

(kkkkk) "PCI DSS" means the Payment Card Industry Data Security Standard.

(IllIII) "*Permit*" means the franchises, licenses, permits, easements, variances, orders, certificates, registrations, accreditations, qualifications, filings, waivers, exemptions, clearance, Consents, approvals, identification numbers and other authorizations required by Law or any Governmental Authority to be held by a Person for a particular purpose.

(mmmmmm) "Permitted Lien" means (i) liens for Taxes not yet due and payable or being contested in good faith by appropriate procedures; (ii) mechanics', carriers', workmen's, repairmen's or other like liens arising or incurred in the Ordinary Course of Business for amounts not yet due and payable; (iii) easements, rights of way, zoning ordinances and other similar restrictions imposed by Law which, to the Knowledge of Seller, are not currently violated by the use of the Leased Real Property; (iv) the Liens set forth on Schedule 1.1(mmmmmm); (v) with respect to Intellectual Property, non-exclusive licenses entered into in the Ordinary Course of Business with customers and vendors of the Business, and (vi) liens arising under original purchase price conditional sales contracts and equipment leases with third parties entered into in the Ordinary Course of Business; provided that Seller has made available to Buyer the underlying sales contract or equipment lease

(nnnnnn) "Person" means any individual, partnership, limited liability company, corporation, estate, trust, organization, unincorporated association, joint venture, business or other legal entity, and any government or any governmental agency or political subdivision thereof.

(00000) "Personal Information" means, collectively, any information that (i) identifies (or in combination with other information may identify), relates to, describes, is capable of being associated with, or can be reasonably linked, directly or indirectly, to a natural person or household; (ii) is governed, regulated, or protected by one or more Information Privacy and Security Laws concerning the Processing of information relating to an identified or identifiable natural person, including, without limitation, any "protected health information" (as defined by HIPAA); or (iii) Seller or any of its Affiliates receive on behalf of Seller's customers that is governed, regulated or protected by Information Privacy and Security Laws.

(pppppp) "PII Consents" has the meaning set forth in Section 3.17(a).

(qqqqqq) "Post-Closing Adjustment Objection Notice" shall have the meaning set forth in

Section 2.3(c).

(rrrrrr) "Post-Closing Adjustment Statement" shall have the meaning set forth in

Section 2.3(c).

(ssssss) "Post-Closing Tax Period" shall have the meaning set forth in Section 7.5(b).

(tttttt) "Prepaid Amount" shall have the meaning set forth in Section 2.3(b).

(uuuuuu) "Principal Stockholders" shall have the meaning set forth in the Recitals.

(vvvvvv) "*Privacy Policies*" means each past or present external, published privacy policy or notice of (A) Seller, including any such policy or notice relating to: (i) the privacy and security of the Personal Information of Customers; (ii) the Processing of any Personal Information; and (iii) any Seller employee Personal Information, and (B) Seller Parent, including any such policy or notice, but in each case solely to the extent relating to Seller's activities with respect to (A)(i) or (ii) as set forth in this <u>Section 1.1(vvvvvv</u>).

(wwwww) "Private Programs" shall have the meaning set forth in Section 3.5(d).

(xxxxxx) "Processe," "Processed," "Processes," or "Processing" means any operation or set of operations performed on Personal Information, whether or not by automatic means, such as receipt, collection, monitoring, maintenance, creation, recording, organization, structuring, storage, adaptation or alteration, retrieval, consultation, use, processing, analysis, transfer, transmission, disclosure, dissemination or otherwise making available, combination, erasure, or destruction.

(yyyyyy) "Provider Agreements" shall have the meaning set forth in Section 3.5(d).

(ZZZZZZ) "Proxy Statement" shall have the meaning set forth in Section 3.25.

(aaaaaaa) "Purchase Price" shall have the meaning set forth in Section 2.3(a).

(bbbbbbb) "R&W Policy" means a Buyer-Side Representation and Warranty Insurance Policy issued in respect of the transactions contemplated by this Agreement, for the benefit of Buyer as the named insured.

(cccccc) "Real Property Leases" has the meaning set forth in Section 3.9(a).

(ddddddd) "Referring Physician" shall have the meaning set forth in Section 3.5(e).

(eeeeeee) "Related Parties" shall have the meaning set forth in Section 8.6(f).

(ffffff) "Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, migrating, dumping, disposing, or other release into the environment.

(ggggggg) "Release Date" shall have the meaning set forth in Section 9.6(b).

(hhhhhhh) "*Representative*" means, with respect to any Person, any director, officer, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.

(iiiiiii) "Restraint" shall have the meaning set forth in Section 6.3(a).

(jjjjjjj) "Restricted Period" shall have the meaning set forth in Section 7.14(a).

(kkkkkk) "Retention Bonus Amount" means an amount equal to the lesser of (x) one-half of the aggregate amount of the retention bonuses previously paid or to be paid by Seller to Business Employees, in each case, in connection with the transactions contemplated by this Agreement and (y) \$1,600,000.

(IllIIII) "Sanctions" means all applicable economic sanctions regulations, embargoes, and related measures enforced by (i) OFAC, including the Foreign Assets Control Regulations (31 CFR Parts 500-598) and lists administered by OFAC, including but not limited to the list of Specially Designated Nationals and Blocked Persons, the List of Foreign Sanctions Evaders, or the Sectoral Sanctions Identifications List; (ii) the U.S. Department of Commerce, including the Denied Persons List, the Entity List, or the Unverified List; (iii) the U.S. Department of State, including the Debarred List; (iv) the United Nations Security Council; (v) Her Majesty's Treasury; the (vi) European Union and as adopted by its member states; or (vii) any other applicable sanctions authority.

(mmmmmmm) "Second Request" shall have the meaning set forth in Section 5.3(b).

(nnnnnnn) "Security Risk Assessments" shall have the meaning set forth in Section 3.17(e).

(0000000) "Seller" shall have the meaning set forth in the preamble hereto.

(ppppppp) "Seller Counsel" shall have the meaning set forth in Section 10.15.

(qqqqqqq) "Seller Indemnified Parties" shall have the meaning set forth in Section 9.1(b).

(rrrrrr) "Seller Parent" shall have the meaning set forth in the preamble hereto.

(sssssss) "Seller Plans" shall have the meaning set forth in Section 7.9(d).

(ttttttt) "Seller Termination Fee" shall have the meaning set forth in Section 8.6(a).

(uuuuuuu) "Shareholder Approval Date" shall mean the date on which the Stockholder Approval is obtained.

(vvvvvvv) "Specified Circumstance" means (i) the condition set forth in Section 6.3(b) is not satisfied and has not been waived by both Seller and Buyer; or (ii) as a result of a Restraint under any applicable Competition Law, the condition set forth in Section 6.3(a) is not satisfied and has not been waived by both Seller and Buyer.

(wwwwww) "Standing Order" means a test request for a specific patient requiring repetitive testing performed on a regular basis for a specified period of time to monitor a condition or disease.

(xxxxxxx) "Stark Law" shall have the meaning set forth in Section 3.5(a).

(yyyyyy) "STAT Services" shall have the meaning set forth in Section 1.1(rrr)(xv).

(zzzzzzz) "STAT Services Agreement Amendment" means that certain Amendment No. 1, dated as of the Effective Date, to that certain STAT Services Agreement between Seller and Buyer, dated as of October 17, 2022.

(aaaaaaaa) "Stockholder Approval" shall have the meaning set forth in Section 3.2(b).

(bbbbbbbb) "Stockholder Meeting" shall have the meaning set forth in Section 5.6(a).

(ccccccc) "Straddle Period" shall have the meaning set forth in Section 7.5(b).

(dddddddd) "Superior Proposal" shall have the meaning set forth in Section 5.2(b).

(eeeeeee) "Tax" means any assessment of any kind or nature imposed by any Governmental Authority, any duties including customs duties, any federal, state, local or non-U.S. income, gross receipts, business and occupancy, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), unclaimed property, escheatment, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated or other taxes or similar assessments, charges or fees, including the health insurance providers fee under the Patient Protection and Affordable Care Act (PPACA) Section 9010, whether computed on a separate or consolidated, unitary or combined basis or in any other manner, including any interest, penalty, deficiency or addition thereto, whether disputed or not and including any obligation to indemnify or otherwise assume or succeed to the liability for Taxes of any other Person, whether as a result of (i) being a member of an affiliated, consolidated, combined or unitary group for any period, (ii) any express or implied obligation to indemnify any other Person, (iii) a Contract with any other Person or (iv) any other reason.

(ffffffff) "Tax Returns" means any return (including any information return), report, statement, declaration, estimate, schedule, notice, notification, form, election, certificate, voucher or other document or information (including any amendments thereto) with respect to Taxes.

(ggggggg) "Testing" means all laboratory testing services.

(hhhhhhh) "*Transaction Documents*" means this Agreement, the Assignment, Assumption and Bill of Sale, the Assignment and Assumption of Leases, the Escrow Agreement, the STAT Services Agreement Amendment, the Transition Services Agreement (if any), the Intellectual Property Agreement, and all other documents to be executed by one or more of the Parties, as identified in this Agreement, in connection with the consummation of the transactions contemplated hereby.

(iiiiiiii) "Transaction Taxes" shall have the meaning set forth in Section 7.5(a).

(jjjjjjj) "Transition Services Agreement" shall have the meaning set forth in Section 7.16.

(kkkkkkk) "Unresolved Claims" shall have the meaning set forth in Section 9.6(b).

(IllIllII) "Unresolved Claims Amount" shall have the meaning set forth in Section 9.6(b)

(mmmmmmm) "Vacation Rollover Amount" shall the dollar amount equal to the value of any accrued, but unused vacation time credited to any Hired Employee under any benefit plan maintained by Buyer and/or its Affiliates providing similar benefits to Hired Employee after the Effective Date pursuant to Section 7.9(d). The Vacation Rollover Amount shall be calculated as follows: The sum of the products of (i) the number of unused vacation hours, up to ten, accrued by each Hired Employee, multiplied by (ii) each respective Hired Employee's regular hourly rate of pay. For employees classified as exempt, "regular hourly rate of pay" shall be calculated as follows: base salary divided by 52 divided by 40.

(nnnnnnn) "WARN" means the Workers Adjustment and Retraining Notification Act of 1988, as amended.

(00000000) "WARN Laws" shall have the meaning set forth in Section 7.9(e).

1.2 Other Definitional Provisions. Unless the context clearly requires otherwise, references to the plural include the singular, to the singular include the plural, to the part include the whole and to the whole include the part. References to "Articles," "Sections," "Subsections," "Exhibits" and "Schedules" are to Articles, Sections, Subsections, Exhibits and Schedules of this Agreement unless otherwise specifically provided. All references to statutes and related regulations shall include any amendments of same and any successor statute and regulations promulgated thereunder. A reference to any document, agreement or instrument shall include such document, agreement or instrument as amended, modified or supplemented from time to time. All words used in this Agreement will be construed to be of such gender or gender neutral as the context requires. A reference to any Person includes its permitted successors and permitted assigns. For purposes of this Agreement, (a) the words "include," "includes" and "including" are not limiting and shall be deemed to be followed by the words "without limitation" unless context requires otherwise; (b) the word "or" is not exclusive; and (c) the words "herein," "hereof," "hereby," "hereto," "hereunder" and words of like import shall refer to this Agreement as a whole and not to any particular section or subdivision of this Agreement.

ARTICLE II SALE AND PURCHASE OF ACQUIRED ASSETS

- 2.1 Transfer of Acquired Assets. Subject to Section 2.2, and on the terms and subject to the conditions set forth in this Agreement, Seller (and solely with respect to the Acquired Intellectual Property Assets, Seller Parent and its Affiliates, if applicable) shall irrevocably sell, convey, assign, transfer, and deliver to Buyer, and Buyer shall purchase, accept and acquire from Seller (and solely with respect to the Acquired Intellectual Property Assets, Seller Parent and its Affiliates, if applicable), all of Seller's (and solely with respect to the Acquired Intellectual Property Assets, Seller Parent's and its Affiliates', if applicable) right, title and interest in, to and under the Acquired Assets, free and clear of all Liens, except for Permitted Liens.
- 2.2 The Closing. Subject to the terms and conditions set forth in this Agreement, the closing of the transactions contemplated by this Agreement, namely the purchase and acquisition from Seller of the Acquired Assets, and the assumption by Buyer of the Assumed Liabilities (the "Closing"), shall take place electronically through the exchange of documents and signatures (a) on the first Monday that is a Business Day following the later of (i) the second Business Day following the date on which the last of the conditions to the Closing set forth in Article VI are satisfied or waived (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions at such time) and (ii) 120 days from the Effective Date or (b) at such other time, date or place as the Parties may agree (the "Closing Date"). At the option of the Parties, documents to be delivered at the Closing may be delivered by electronic transmission. For purposes of this Agreement, the Closing shall be deemed effective as of 12:01 a.m. ET time on the Closing Date.

2.3 Purchase Price.

(a) At the Closing, as payment for the Acquired Assets, Buyer shall pay to Seller \$146,000,000 minus (i) the Estimated Vacation Rollover Amount plus (ii) the Estimated Prepaid Amount minus (iii) the Escrow Amount minus (iv) the Debt Amount minus (v) the Aggregate Ticking Fee Closing Paymentplus (vi) the Retention Bonus Amount and (vii) minus any amounts deducted for motor vehicles as provided in Section 2.8(a)(ix) below (such collective amount, as may be adjusted by this Section 2.3, the "Purchase Price"), by wire transfer of immediately available funds in lawful currency of the United States of America to the account designated by Seller in writing at least four Business Days prior to the Closing. At the Closing, Buyer shall (x) deposit the Escrow

Amount in accordance with <u>Section 2.10</u> and (y) deliver or cause to be delivered on behalf of Seller or Seller Parent the amount payable to each holder of Debt identified in the Payoff Letters in order to fully discharge such Debt and terminate all applicable obligations and Liabilities of Seller or Seller Parent related thereto, as specified in the Payoff Letters and in accordance with this Agreement.

(b) No later than five Business Days prior to the Closing Date, Seller shall calculate (i) the amount of all reserves, security and other deposits, advances and prepaid expenses and credits that are assignable to Buyer and related to the Assigned Contracts or any other Acquired Asset (the "Prepaid Amount"), (ii) the Vacation Rollover Amount and (iii) the Retention Bonus Amount with respect to the applicable Business Employees, and provide to Buyer a statement setting forth each such calculation in reasonable detail with reasonable supporting documentation, including with respect to the Retention Bonus Amount the names and amounts payable to each individual employee and the date or anticipated date of the payment of such amounts to such individual employees. The Prepaid Amount set forth on such statement provided prior to the Closing (the "Estimated Prepaid Amount") and the Vacation Rollover Amount set forth on such statement provided prior to the Closing (the "Estimated Vacation Rollover Amount") shall, in each case, be used to calculate the Purchase Price to be paid by Buyer to Seller at the Closing pursuant to Section 2.3(a).

(c) As soon as reasonably practicable after the Closing Date, but in any event no later than 90 days thereafter, Buyer will calculate the Prepaid Amount and the Vacation Rollover Amount and provide Seller with a statement (the "Post-Closing Adjustment Statement") setting forth each such calculation in reasonable detail with reasonable supporting documentation. The Post-Closing Adjustment Statement will be deemed to be accepted by Seller and shall be conclusive for purposes of determining the Prepaid Amount and the Vacation Rollover Amount, unless Seller delivers to Buyer, within 30 days following receipt of the Post-Closing Adjustment Statement from Buyer, a notice (each, a "Post-Closing Adjustment Objection Notice") setting forth Seller's objections to such Post-Closing Adjustment Statement in reasonable detail. If Seller and Buyer fail to resolve the issues raised in the Post-Closing Adjustment Objection Notice within 30 days of Buyer's receipt of the Post-Closing Adjustment Objection Notice, Seller and Buyer shall submit the items under dispute in the Post-Closing Adjustment Statement (the "Disputed Items") to Ernst & Young LLP (or if such firm is unable to serve, another nationally recognized independent firm jointly chosen by Buyer and Seller) (the "Independent Accountant'), for resolution in accordance with the dispute resolution provisions set forth in Section 2.3(d) and the relevant terms of this Agreement. Within ten Business Days following the final determination of the Prepaid Amount and the Vacation Rollover Amount, in each case, pursuant to this Section 2.3(c) (such amounts as so finally determined pursuant to this Section 2.3(c), the "Final Prepaid Amount" and the "Final Vacation Rollover Amount"), (w) if the difference of (1) the Final Prepaid Amount minus (2) the Final Vacation Rollover Amount is greater than (x) the difference of (1) the Estimated Prepaid Amount minus (2) the Estimated Vacation Rollover Amount, Buyer shall pay such difference to Seller, and (y) if the difference of (1) the Final Prepaid Amount minus (2) the Final Vacation Rollover Amount is less than (z) the difference of (1) the Estimated Prepaid Amount minus (2) the Estimated Vacation Rollover Amount, Seller and Seller Parent shall pay such difference to Buyer. Any payments required to be made pursuant to this Section 2.3(c) shall be made by wire transfer of immediately available funds in lawful currency of the United States of America to the account designated by Seller or Buyer, as applicable, in writing at least four Business Days prior to the date of such payment.

(d) If any Disputed Item is referred to the Independent Accountant for resolution pursuant to this Section 2.3, then the Independent Accountant shall determine only with respect to such Disputed Item submitted whether such Disputed Item requires adjustment and the amount of any such required adjustment. Seller and Buyer shall be entitled to submit to the Independent Accountant such documents and other information to support their respective calculations of each Disputed Item. Seller and Buyer shall instruct the Independent Accountant to, and the Independent Accountant shall, (i) make its determination (A) based solely on such documentation submitted by Seller and Buyer in accordance with the guidelines and procedures set forth in this Agreement; (B) not on the basis of an independent review; and (C) within the range of values assigned to such Disputed Item by Seller and Buyer; (ii) complete its determination within 30 days of submission to the Independent Accountant; and (iii) notify both Seller and Buyer in writing of its final determination, which shall state in reasonable detail the findings of fact on which it is based. Any finding by the Independent Accountant shall be final, non-appealable and binding on Seller and Buyer and accompanied by a certificate from the Independent Accountant certifying that such determination was reached in accordance with the provisions of this Section 2.3. Seller and Buyer shall each bear 50% of the fees and costs of the Independent Accountant for such determination.

(e) Beginning on the first Business Day after the Shareholder Approval Date, and again every day thereafter until the earlier of (i) the Closing Date and (ii) termination of this Agreement pursuant to Article VIII, Buyer shall be obligated to pay to Seller an amount equal to seventy-five thousand dollars (\$75,000) per day during such period (the "Daily Ticking Fee"). On the last day of each month prior to the Closing or termination of this Agreement pursuant to Article VIII (and again on the date this Agreement is terminated, if applicable), Buyer shall deposit by wire transfer of immediately available funds in lawful currency of the United States of America to the account designated by Seller in writing an amount equal to the sum of all Daily Ticking Fee accrued during the period between such date and the date of the immediately preceding deposit of the Daily Ticking Fee; provided, however, that the aggregate amount of Buyer's liability hereunder with respect to the Daily Ticking Fee shall not exceed seven million dollars (\$7,000,000). The total amount of the accumulated Daily Ticking Fee owed hereunder by Buyer shall be the "Aggregate Ticking Fee"; provided, that for the avoidance of doubt, in no event shall the Aggregate Ticking Fee exceed seven million dollars (\$7,000,000). If the Closing occurs, the Aggregate Ticking Fee Closing Payment shall be credited against the Purchase Price as set forth in Section 2.3(a) without any interest earned thereon. Upon the termination of this Agreement the Aggregate Ticking Fee will be applied as set forth in Section 2.3(f).

(f) Seller shall retain the Aggregate Ticking Fee as liquidated damages if (x) Buyer, on the one hand, or Seller or Seller Parent, on the other hand, has terminated this Agreement pursuant to Section 8.2(a), (y) as of 11:59 p.m. (New York City time) on the Outside Date a Specified Circumstance exists and (z) each of the conditions set forth in Section 6.3 (other than the conditions set forth in Section 6.3(a) as a result of a Restraint under any applicable Competition Law and Section 6.3(b)) and Section 6.2 (other than any requirement of receipt of a certificate regarding compliance with any such condition), in each case other than with respect to a Specified Circumstance, has been satisfied or waived (together the circumstances set forth in clauses (x), (y) and (z), a "Retention Event"); provided, that if this Agreement is terminated pursuant to Section 8.2(a) and (x) either Seller or Seller Parent is then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement, or if any representations or warranties of Seller or Seller Parent have become, after the Effective Date, and at that time remain, untrue, in each case such that it would result in the failure of any of the conditions set forth in Section 6.2 or Section 6.3 or (y) Seller or Seller Parent has failed to perform in any material respect any of its respective obligations under this Agreement such that Buyer's conditions to close would not have been met, notwithstanding a Retention Event, Seller shall return the Aggregate Ticking Fee without interest to Buyer within two Business Days after termination of this Agreement in immediately available funds in lawful currency of the United States of America. If this Agreement is terminated pursuant to Sections 8.1, 8.2 (except a termination under Section 8.2(a) in connection with a Retention Event as limited by the proviso at the end of the preceding sentence) or 8.3, then Seller shall return the Aggregate Ticking Fee without interest to Buyer within two Business Days after termination of this Agreement in immediately available funds in lawful currency of the United States of America. Each of the Parties further acknowledges that the payment of the amounts specified in Section 2.3(e) and this Section 2.3(f) is not a penalty, but, in each case, is liquidated damages in a reasonable amount that will compensate Seller in the circumstances in which such amounts are permitted to be retained by Seller for the efforts and resources expended and the opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement, which amount would otherwise be impossible to calculate with precision.

(g) Notwithstanding anything to the contrary set forth in this Agreement, but subject to Section 10.8, each of the Parties acknowledges and agrees that in the event the Aggregate Ticking Fee previously paid by Buyer to Seller is retained by Seller, retention of the Aggregate Ticking Fee pursuant to Section 2.3(f) shall constitute the sole and exclusive monetary remedy of Seller and Seller Parent, and their Related Parties, for all Losses and damages in respect of this Agreement, the transactions contemplated by this Agreement, any breach of any covenant or agreement or otherwise in respect of this Agreement, or the failure to consummate the transactions contemplated hereby, and upon retention of the Aggregate Ticking Fee pursuant to Section 2.3(f). Buyer, or its Related Parties, shall have no further liability or obligation relating to or arising out of this Agreement or the transactions contemplated hereby or thereby (except that the Parties will remain subject to the Confidentiality Agreement and any covenants or obligations that expressly survive the termination of this Agreement, and nothing herein shall limit any Party's remedies with respect thereto), and none of Seller, Seller Parent or their Related Parties shall seek to recover any other damages or seek any other remedy, whether based on a claim at law or in equity, in contract, tort or otherwise, with respect to any Losses suffered in connection with this Agreement or the transactions contemplated hereby (except with respect to the Confidentiality Agreement and any covenants or obligations that

expressly survive the termination of this Agreement and any remedies available to the any Party with respect thereto). For the avoidance of doubt, this <u>Section 2.3(g)</u> applies only in the event of a valid termination under <u>Article VIII</u>. Notwithstanding the foregoing, nothing in this <u>Section 2.3(g)</u> shall relieve any Party or its Related Parties from liability for damages arising from Fraud by any other Party or its Related Parties.

- (h) While Seller may pursue a grant of specific performance or other equitable relief under Section 10.8, subject to Section 8.5, under no circumstances shall Seller be permitted or entitled to receive both a grant of specific performance or other equitable relief and the Aggregate Ticking Fee, in connection with the termination of this Agreement (except with respect to the Confidentiality Agreement and any covenants or obligations that expressly survive the termination of this Agreement).
- 2.4 Assumption of Liabilities. Subject to the terms and conditions of this Agreement, at and as of the Closing, Buyer shall assume the Assumed Liabilities, in accordance with this Section 2.4. Buyer shall not assume, and Seller and Seller Parent, as applicable, shall retain, the Excluded Liabilities. Seller shall be responsible for and shall pay when due all of the Excluded Liabilities. Other than the Assumed Liabilities, nothing contained in this Agreement shall be construed as an agreement by Buyer to assume any Liability or to perform any obligation of Seller or Seller Parent or their Related Parties. For the avoidance of doubt, all pre-paid expenses, rent, equipment rent, utilities and similar items pursuant to the Assigned Contracts (which are not paid under any other Transaction Document), any similar expenses or amounts paid, and goods paid for by Seller but not delivered prior to the Closing whether payable before or after the Closing, shall be apportioned between the period beginning before and ending the day prior to the Closing and the period beginning on the Closing. Seller shall be liable for all amounts apportioned to periods prior to the Closing and Buyer shall be liable for all amounts apportioned to periods on and after the Closing. Buyer and Seller shall settle all prorations within 90 Business Days following the Closing Date. Notwithstanding anything in this Agreement to the contrary, if the documents described in Section 2.8(a)(vi) are not delivered to Buyer on or prior to the date that is at least 30 days before the Closing Date, then Seller shall pay any rent due for the Assigned Locations for the first rent period following the Closing Date, consistent with prior practice, and Buyer shall reimburse Seller therefor promptly following receipt of the evidence of such payment from Seller.
- 2.5 Nonassignable Items. Anything in this Agreement to the contrary notwithstanding, this Agreement shall not constitute an agreement to sell, convey, assign, sublease or transfer any Acquired Assets if an attempted sale, conveyance, assignment, sublease or transfer thereof, without the Consent of any Person, would constitute a breach of, or in any way affect, the rights of Seller with respect to such Acquired Assets ("Nonassignable Items"). Seller shall use its commercially reasonable efforts to obtain, and Buyer shall use its commercially reasonable efforts to cooperate in seeking, all Consents and to resolve all impracticalities of sale, conveyance, assignment, sublease or transfer necessary to convey to Buyer all Nonassignable Items as quickly as practicable on and after the Effective Date, and without any change to the material terms of such items, and Seller shall continue such efforts as may be required for a period of six months after the Closing Date, to facilitate the full and expeditious transfer to Buyer of legal title to the Acquired Assets. Prior to the obtaining of any such Consent, Seller and Buyer shall cooperate with each other in any reasonable and lawful arrangements designed to provide to Buyer the benefits of use of any Nonassignable Item for its term, and to the extent that Buyer receives such benefits, it will assume the Liabilities of Seller thereunder to the extent that Buyer would have been responsible therefor if such Consent had been obtained. Once Consent is obtained, Seller shall promptly assign such Nonassignable Item to Buyer, and Buyer shall assume the Liabilities thereunder; provided that nothing contained in this Section 2.5 or elsewhere in this Agreement shall be deemed to constitute an agreement to exclude from the Acquired Assets the benefits, nor from the Assumed Liabilities the Liabilities associated with such benefits, under any Assigned Contract as to which Consent may be necessary.
- 2.6 Purchase Price Allocation. Buyer shall prepare an allocation of the Purchase Price, and any Liabilities of Seller or Seller Parent treated as assumed by Buyer for Tax purposes among the Acquired Assets in accordance with Section 1060 of the Code (and any similar provision of state, local, or non-U.S. Law, as appropriate), for approval by Seller (the "Draft Purchase Price Allocation"), which approval shall not be unreasonably withheld, conditioned or delayed, and which, upon such approval, shall be binding upon Buyer and Seller as the "Final Purchase Price Allocation." Seller shall provide such information as Buyer may reasonably request to assist Buyer with preparing the Draft Purchase Price Allocation, including the net book value (to the extent reasonably determinable) for purposes of GAAP as of the Closing of any tangible assets that are Acquired Assets and a description of such item, original cost and date of acquisition (in each case, to the extent reasonably determinable).

to assist Buyer with preparing the Draft Purchase Price Allocation. The Draft Purchase Price Allocation shall be prepared in a manner consistent with Schedule 2.6. Buyer shall deliver the Draft Purchase Price Allocation to Seller within 30 days after the Parties agree to the Post-Closing Adjustment Statement pursuant to Section 2.3(c), and Seller shall inform Buyer in writing within 30 days of receiving the Draft Purchase Price Allocation if Seller does not approve the Draft Purchase Price Allocation or any portion thereof. If Seller does not provide written notice to Buyer that Seller objects to the Draft Purchase Price Allocation within the 30-day period, then the Draft Purchase Price Allocation shall become the Final Purchase Price Allocation. If Seller provides Buyer a written objection to the Draft Purchase Price Allocation within such 30-day period (together with a statement explaining Seller's particular objections to the Draft Purchase Price Allocation along with Seller's reasons for any objections), then Buyer and Seller shall cooperate in good faith for a period of 30 days to resolve their differences. If Buyer and Seller resolve their differences within such 30-day period, the Draft Purchase Price Allocation, as amended to reflect any changes agreed upon by Buyer and Seller, shall become the Final Purchase Price Allocation. If Buyer and Seller are unable to resolve their differences within such 30-day period, Buyer and Seller and their respective Affiliates shall be entitled to allocate the Purchase Price to the Acquired Assets in such manner as each independently determines is reasonable, and there shall be no Final Purchase Price Allocation. If a Final Purchase Price Allocation is established, Buyer and Seller and their respective Affiliates shall report and file all Tax Returns which each of them is required by Law to file and report (including, but not limited to, Internal Revenue Service Form 8594) in all respects and for all purposes consistent with the Final Purchase Price Allocation. In the event that (a) the Final Prepaid Amount differs from the Estimated Prepaid Amount or (b) the Final Vacation Rollover Amount differs from the Estimated Vacation Rollover Amount, the Parties and their Affiliates shall update the IRS Form 8594 in a manner consistent with Schedule 2.6. If a Final Purchase Price Allocation is established, neither Buyer, Seller, nor any of their respective Affiliates, shall take any position on any Tax Return, audit or otherwise that is inconsistent with such Final Purchase Price Allocation, if any, unless required to do so by applicable Law. In the event that the Final Purchase Price Allocation is disputed by any Governmental Authority, the Party receiving notice of the dispute shall promptly notify the other Party in writing of such notice and resolution of the dispute.

2.7 Certain Acknowledgments. No part of this Agreement shall be construed to induce, encourage, solicit or reimburse the referral of any patients or business, including any patient or business funded in whole or in part by federal or state government programs (i.e., Medicare, Medicaid, TRICARE, etc.). The Parties acknowledge and agree that there is no requirement under this Agreement or any other agreement between the Parties to refer patients to Buyer or its Affiliates. None of Seller, any of its Affiliates or any of their respective officers, directors or employees shall, at any time, directly or indirectly, require, request or otherwise influence the decision of any physician that practices as an employee or contractor of Seller or any of its Affiliates to order any services from Buyer or any of its Affiliates. No payment made under this Agreement is in return for the referral of patients or business, including those paid in whole or in part by federal or state government programs. Seller and Seller Parent each agrees and covenants that none of the proceeds of the transactions contemplated by this Agreement will be paid, or otherwise provided, directly or indirectly, to any physician or their immediate family members (as defined by the Stark Law) who are affiliated with Seller or its Affiliates and who have referred or are in a position to refer Testing to the Business.

2.8 Closing Deliverables.

- (a) At the Closing, Seller and/or Seller Parent, as applicable, shall deliver to Buyer the following:
 - (i) the Assignment, Assumption and Bill of Sale, with respect to the Acquired Assets and Assumed Liabilities;
 - (ii) a certificate of a duly authorized officer of Seller, dated as of the Closing Date, certifying that the conditions set forth in <u>Sections 6.2(a)</u>, <u>6.2(b)</u> and <u>6.2(e)</u> have been satisfied;
 - (iii) (A) a properly completed IRS Form W-9 executed by Seller and (B) if applicable, any certificate, affidavit or other documentation required by applicable Law to establish that no withholding is required under applicable state, local or non-U.S. Tax Laws in connection with Buyer's purchase of the Acquired Assets:
 - (iv) payoff letters duly executed by each holder of Debt underlying any Lien, including those Liens which are set forth on Schedule 2.8(a)(iv), (other than a Permitted Lien) on any Acquired Asset (each, a "Payoff Letter"), pursuant to which letters such holders shall agree that upon payment of

the amount specified in such Payoff Letter (A) all outstanding obligations of Seller or Seller Parent arising under or related to the applicable Debt shall be repaid, discharged and extinguished in full; (B) all Liens (other than Permitted Liens) in favor of such holder in connection therewith shall be released, including by filing termination statements with respect to all UCC financing statements;(v) the Escrow Agreement, duly executed by Seller;

- (vi) assignments of leases in form and substance of the Assignment and Assumption of Leases, duly executed by Seller (or its applicable Affiliates) and the applicable lessor and applicable Lease Extensions, duly executed by Seller and the applicable lessor, each in form and substance reasonably satisfactory to Buyer, including with respect to compliance with applicable regulatory requirements, collectively, for 80% (rounded up to the nearest whole number) of the Assigned Locations set forth on Schedule 5.8, including each of the Assigned Locations described as "required" on Schedule 5.8, and all consents or waivers and completion of the landlord disclosure letter in the form provided by Buyer from the applicable lessor required in connection herewith, each in form and substance reasonably satisfactory to Buyer, accompanied by all properly completed IRS Form W-9s to the extent reasonably obtainable by Seller;
- (vii) if requested by Buyer pursuant to $\underline{\text{Section 7.16}}$, the Transition Service Agreement, duly executed by Seller and Seller Parent;
- (viii) the Intellectual Property Agreement, duly executed by Seller and Seller Parent and any other Affiliate of Seller that owns any Acquired Intellectual Property Assets;
- (ix) any vehicle title, registration documents or bills of sale or other payoff documents necessary to effect the transfer of titles to those Acquired Assets comprised of the motor vehicles listed in Section 3.23 of the Disclosure Schedules, free and clear of all Liens except for Permitted Liens; provided that at least 30 days prior to Closing Seller shall identify by written notice to Buyer the motor vehicles for which transfer of title will not be made at Closing (which shall not comprise more than 1/4th of the number of the motor vehicles listed on Section 3.23 of the Disclosure Schedules), in which event such identified number of motor vehicles need not be delivered at Closing, and Buyer shall be entitled to deduct the value of such vehicles from the Purchase Price at Closing (based on Kelly Blue Book sale prices), and if title to any of those identified motor vehicles is transferred to Buyer within 30 days following Closing, Buyer shall promptly pay to Seller the amount so deducted for such motor vehicles; and
- (x) such other customary instruments of transfer, assumption, filings or documents, in form and substance reasonably satisfactory to Buyer, as may be required to give effect to this Agreement.
 - (b) At the Closing, Buyer shall deliver to Seller the following:
 - (i) the Purchase Price;
 - (ii) the Assignment, Assumption and Bill of Sale, duly executed by Buyer;
- (iii) a certificate of a duly authorized officer of Buyer, dated as of the Closing Date, certifying that the conditions set forth in Sections 6.1(a) and 6.1(b) have been satisfied;
- (iv) a copy of the bound commitment in respect of the R&W Policy and evidence that the R&W Policy is in full force and effect;
 - (v) the Escrow Agreement, duly executed by Buyer and the Escrow Agent
 - (vi) the Intellectual Property Agreement, duly executed by Buyer; and
- (vii) such other customary instruments of transfer, assumption, filings or documents, in form and substance reasonably satisfactory to Seller, as may be required to give effect to this Agreement.
- 2.9 Other Documentation. With respect to Assigned Locations for which Assignment and Assumptions of Leases and Lease Extensions were not delivered at the Closing, as soon as reasonably practicable, and in any event within 60 days following the Closing, subject to the terms of Section 2.5 and Section 2.8(a)(vi), Seller shall deliver to Buyer assignments of leases in form and substance of the Assignment and Assumption of Leases, duly executed

by Seller (or its applicable Affiliates) and any applicable Lease Extensions in form and substance reasonably satisfactory to Buyer including with respect to compliance with applicable regulatory requirements, for the remainder of the Assigned Locations and all consents or waivers and landlord disclosure letters from the applicable lessor of such Assigned Locations required in connection herewith, each in form and substance reasonably satisfactory to Buyer, and all properly completed IRS Form W-9s to the extent reasonably obtainable by Seller in connection herewith.

2.10 Escrow. Subject to the terms and conditions of this Agreement, at the Closing, Buyer shall deposit, or cause to be deposited, the Escrow Amount by wire transfer of immediately available funds into an account designated by the Escrow Agent. The Escrow Fund shall be held, managed and distributed by the Escrow Agent in accordance with the terms of this Agreement and the Escrow Agreement to satisfy claims, if any, made by Buyer or any other Buyer Indemnified Party against Seller pursuant to Article IX. Each of Buyer and Seller agree to take all actions reasonably necessary to cause any payment to be made pursuant to the Escrow Agreement, including by delivering executed written instructions to the Escrow Agent and directing the Escrow Agent to make such payments.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF SELLER AND SELLER PARENT

Except as set forth in the schedules delivered by Seller and Seller Parent to Buyer prior to the execution of this Agreement, which schedules shall reference the specific section of this Article III to which such schedules relate (collectively, the "Disclosure Schedules") (provided, that any fact or condition disclosed in any section or subsection of the Disclosure Schedules shall be deemed to be an exception to another section of the Disclosure Schedules to the extent the relevance of such exception to such other sections is readily apparent on the face of such disclosure without reference to any matters referenced therein), each of Seller and Seller Parent, as applicable, represents and warrants to Buyer:

3.1 Organization, Standing and Power; No Subsidiaries. Each of Seller and Seller Parent is a corporation, duly organized, validly existing, and in good standing under the Laws of the State of New York, and has the requisite power and authority to own, lease and operate its properties and assets, including the Acquired Assets, and to carry on the Business as now being conducted. True, correct and complete copies of the certificate of incorporation and bylaws, as amended through the Effective Date, of Seller and Seller Parent have been delivered to Buyer. Seller does not have any subsidiaries.

3.2 Authority.

(a) Each of Seller and Seller Parent has all requisite power and authority to execute and deliver the Transaction Documents to which it is, or will be, a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby, subject to receipt of Stockholder Approval. The execution and delivery by each of Seller and Seller Parent of this Agreement and the other Transaction Documents to which it is, or will be, a party, the performance by each of Seller and Seller Parent of its obligations hereunder and thereunder and the consummation by each of Seller and Seller Parent of the transactions contemplated hereby and thereby, have been duly authorized by all necessary corporate action, and no other action, or approval on the part of Seller or Seller Parent or the stockholders of Seller Parent is necessary to authorize this Agreement or the other Transaction Documents or to consummate the transactions contemplated hereby or thereby, except the Stockholder Approval. This Agreement has been duly executed and delivered by each of Seller and Seller Parent and, assuming the due authorization, execution and delivery thereof by Buyer, constitutes the legal, valid and binding obligation of Seller and Seller Parent, enforceable against Seller and Seller Parent in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and similar Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law). When each other Transaction Document to which Seller and Seller Parent are or will be a party has been duly executed and delivered by Seller and Seller Parent, as applicable (assuming due authorization, execution and delivery by each other party thereto), each such Transaction Document will constitute a legal, valid and binding obligation of Seller and Seller Parent, as applicable, enforceable against Seller and Seller Parent in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally or by general equity principles (regardless of whether enforcement is sought in a proceeding in equity or at law).

(b) The affirmative vote of the holders of a majority of the outstanding shares of Seller Parent's common stock entitled to vote to approve the transactions contemplated by this Agreement (the "Stockholder Approval") is the only vote or consent of the holders of any class or series of Seller's or Seller Parent's capital stock that is necessary in connection with the consummation of the transactions contemplated by this Agreement, and no other corporate or stockholder proceedings are necessary to approve this Agreement or to consummate the transactions contemplated hereby. Prior to the execution of this Agreement, at a meeting duly called and held, the board of directors of Seller Parent duly, validly and adopted resolutions (i) approving and declaring advisable this Agreement and the transactions contemplated hereby on the terms and subject to the conditions set forth herein, (ii) declaring and determining that it is in the best interests of the stockholders of Seller Parent that Seller Parent and Seller enter into this Agreement and consummate the transactions contemplated hereby on the terms and subject to the conditions set forth herein, (iii) directing that this Agreement be submitted to a vote at a meeting of the stockholders of Seller Parent for approval and adoption, and (iv) recommending that the stockholders of Seller Parent approve and adopt this Agreement ((i), (ii), (iii) and (iv) being collectively referred to herein as the "Board Recommendation"), which resolutions, as of the date of this Agreement, have not been rescinded, modified or withdrawn in any way.

3.3 No Conflict; Required Filings and Consents.

- (a) Neither Seller nor Seller Parent is in default under or in violation or breach of any provision of its certificate of incorporation, bylaws or other governing documents, in each case as amended or restated. Seller is not in material default under or in material violation of any provision of any Assigned Contract to which it is a party or Assigned Contract by which it or any of its properties or assets is bound and which is material to the Business, the transactions contemplated hereby, or Buyer's unencumbered possession or use of the Acquired Assets, and except as disclosed on Section 3.3(a) of the Disclosure Schedules, neither Seller nor Seller Parent has received any written notice that Seller is in default or violation thereunder.
- (b) The execution and delivery of the Transaction Documents by each of Seller and Seller Parent do not, and the performance of the Transaction Documents by Seller and Seller Parent and the consummation of the transactions contemplated hereby or thereby will not (with or without notice or lapse of time): (i) conflict with or violate the certificate of incorporation, bylaws or other governing documents, in each case as amended or restated, of Seller or Seller Parent, (ii) conflict with or violate, in any material respect, any Law or Governmental Order in effect as of the Effective Date and applicable to Seller, Seller Parent, the Business, the Assumed Liabilities or the Acquired Assets, or (iii) except as set forth in Section 3.3(b) of the Disclosure Schedules, conflict with, result in any violation or breach of, or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration, first refusal, first offer or cancellation of, or require payment under, or require any notice or consent under, any of the Assigned Contracts or Assigned Real Property Leases, or result in the creation or imposition of any Lien (other than Permitted Liens) on any of the Acquired Assets.
- (c) Except (i) as disclosed on Section 3.3(c) of the Disclosure Schedules, (ii) for the filing with the SEC of the preliminary proxy statement, the Proxy Statement and any related filings under Section 14 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), (iii) the Stockholder Approval, (iv) filings, permits, clearances, authorizations, consents, orders and approvals as may be required pursuant to the rules and regulations of the New York Stock Exchange ("NYSE") and (v) the pre-merger notifications requirements under the HSR Act, the execution and delivery of the Transaction Documents by Seller and Seller Parent does not, and the performance of the Transaction Documents by Seller and Seller Parent does not, and the transactions contemplated hereby and thereby will not require Seller or Seller Parent obtain any Consent, Permit or order of, make any filing with or notification to, or procure other action by, any Person based on any Laws.
- **3.4** <u>International Trade and Anti-Corruption</u>. During the past five years, and except as set forth on <u>Section 3.4 of the Disclosure Schedules</u>:
- (a) The Business has not violated any Export and Import Law or Sanctions, or made a voluntary disclosure with respect to any violation of any Export and Import Law or Sanctions. The Business is and has prepared and timely applied for all import and export licenses required under any Export and Import Law or Sanctions for the conduct of its business. Neither the Business nor, to the Knowledge of Seller, any employee or agent of the Business: (i) is or has been a Person with whom transactions are prohibited or limited under any Law including applicable Sanctions; or (ii) has violated or made a voluntary disclosure regarding any such Law.

No product, service or financing provided by the Business has been, directly or indirectly, provided to, sold to or performed for or on behalf of a national of, or Persons located in, the Crimea Region of Ukraine, Cuba, Iran, North Korea, Syria or any other country, region or Person against which the United States maintains, or has maintained at the time of relevant activity, Sanctions.

(b) No Sanctions-related, export-related or import-related Law, investigation or inquiry is or has been pending or, to the Knowledge of Seller, threatened against the Business or any employee or agent of the Business (in such Person's capacity as an employee or agent of the Business) by or before (or, in the case of a threatened matter, that would come before) any Governmental Authority.

(c) Neither the Business nor any of the Business's employees have, and, to the Knowledge of Seller, no agent of the Business or any other Person associated with or acting for or on behalf of the Business or the Acquired Assets has: (i) directly or indirectly violated any provision of the Foreign Corrupt Practices Act of 1977, 15 U.S.C. §§ 78dd-1, et seq., the UK Bribery Act of 2010 or any other applicable anti-corruption or antibribery law or similar Law (each, an "Anti-Corruption Law"); (ii) directly or indirectly made, offered, promised or authorized any unlawful payment, loan or transfer of anything of value, including any reward, bribe, payoff, influence payment, kickback, rebate, contribution, gift, entertainment, advantage or benefit of any kind, to or for the benefit of any Person, including a Foreign Official, for the purpose of: (A) influencing any act or decision of such Person; (B) inducing such Person to do or omit to do any act in violation of a lawful duty; (C) obtaining or retaining business for or with any Person; (D) expediting or securing the performance of official acts of a routine nature; or (E) otherwise securing any improper advantage; (iii) established or maintained any unlawful fund of corporate monies or other properties; (iv) created or caused the creation of any false or inaccurate books and records of the Business related to any of the foregoing; or (v) been the subject of any actual or, to the Knowledge of Seller, threatened (in writing) investigation, allegation, inquiry or enforcement proceeding by any Governmental Authority regarding any offense or alleged offense under any Anti-Corruption Law, and no event has occurred and no condition or circumstance exists that will or could reasonably be expected to give rise to any such investigation, allegation, inquiry or proceeding.

3.5 Legal Compliance; Licenses; Permits and Registration.

(a) Section 3.5(a)(i) of the Disclosure Schedules contains a true, correct and complete list of all material Permits (including, 510(k) clearances, Medicare, Medicaid and other provider numbers, state laboratory licenses, employment Permits, CLIA and DEA certifications and other Permits, but excluding any Permits required under Environmental Laws, which are covered by Section 3.26(b)), necessary for the operation and conduct of the Business, use of the Assigned Locations (and any facilities at the Assigned Locations) and ownership, lease, operation and use of the Acquired Assets (the "Material Permits"). Except as would not be material to the Business or the Acquired Assets taken as a whole, each Business Employee that is required to have a professional permit or licensure for the performance of his/her duties as an employee of Seller or Seller Parent is duly licensed and registered by the applicable Governmental Authority if required under applicable Law, and is in good standing to engage in the licensed activity, and said permit or registration has not been suspended, revoked or restricted in any manner. Each Material Permit is valid, subsisting, and in full force and effect, and the execution and delivery by Seller and Seller Parent of this Agreement and the other Transaction Documents, the performance of this Agreement and the other Transaction Documents by Seller and Seller Parent and the consummation of the transactions contemplated hereby and thereby, shall not in any material respect (with or without notice or lapse of time) (1) conflict with or violate, or breach the terms of, or constitute a default (or any event that with notice or lapse of time or both would become a default) under, or give others any rights of termination or amendment under, any Material Permit, or (2) except as set forth on Section 3.5(a)(ii) of the Disclosure Schedules, require the consent, approval or act of, waiver from, notice to, or the making of any filing with, any Governmental Authority in connection with the transactions contemplated hereby. There is no Action pending or, to the Knowledge of Seller, threatened against Seller, Seller Parent or any of their respective, directors, officers, members, managers or employees that has or would reasonably be expected to result in the revocation, termination, suspension, or material limitation or restriction of any Material Permits or the imposition of any fine, penalty or other sanctions for violation of any Law relating to any Material Permit. Seller and Seller Parent are in material compliance with the terms of all Material Permits and with all requirements, standards and procedures of the Governmental Authorities that issued them related thereto. Seller has made available to Buyer true, complete and correct copies of all material, non-routine (i) surveys, reports, and subpoenas related to any and all certification, licensure or other governmental investigations, inspections, inquiries and audits and summaries of all proficiency test results related to the Business, for the three year period prior to the Effective Date, (ii) written inquiries, notices, and

subpoenas related to utilization, reimbursement or other audits or investigations by any Governmental Authority or other Person related to the Business for the three year period prior to the Effective Date, (iii) filings related to the Business pursuant to Section 1877 of the Social Security Act and the regulations promulgated thereunder (the "Stark Law"), including any voluntary self-disclosures, as well as pursuant to any similar counterpart state Laws, for the three-year period prior to the Effective Date, and (iv) all filings made by Seller related to the Business pursuant to the Clinical Laboratory Improvement Amendments, 42 U.S.C 263a, and all regulations promulgated thereunder ("CLIA") for the three year period prior to the Effective Date, and all of the materials referenced by clauses (i) through (iv) are described on Section 3.5(a)(iii) of the Disclosure Schedules.

- (b) Except as disclosed in <u>Section 3.5(b) of the Disclosure Schedules</u>, neither Seller nor Seller Parent is in breach or violation of, or default under, nor has been at any time during the past five years in breach or violation of, or default under, nor has been given written notice or been charged with any breach or violation of, any Law in any material respect (including under any Governmental Program or Private Program), applicable to the Business, the operation thereof and the ownership or use of Acquired Assets or the Assumed Liabilities;
- (c) During the past three years, neither Seller, Seller Parent, nor any directors, officers, or to the Knowledge of Seller, employees of Seller or Seller Parent have directly or indirectly, overtly or covertly, in violation of any Law in connection with the Business (i) made, or agreed to make, any contribution, gift, bribe, rebate, payoff, influence payment, kickback or other payment or transfer of value to any Person (including, in the case of an individual, any family members of such Person and in the case of an entity, any Affiliates of such entity), regardless of form, whether in money, property or services or other remuneration, including (A) to obtain favorable treatment in securing business, (B) to pay for favorable treatment for business secured, or (C) to obtain special concessions or pay for special concessions already obtained for or in respect of Seller, or (ii) established or maintained any fund or asset that has not been recorded in the books and records of Seller. Other than as set forth on Section 3.5(c) of the Disclosure Schedules, there are no physicians, practice groups or other health care providers who refer testing to Seller and receive distributions or other compensation based in whole or in part on the volume or value of such testing referred by the physician (collectively, the physicians, practice groups and other health care providers on Section 3.5(c) of the Disclosure Schedules, the "Excluded Accounts").
- (d) Except as disclosed in Section 3.5(d)(i) of the Disclosure Schedules, Seller (i) is certified for participation and reimbursement under Titles XVIII and XIX of the Social Security Act (the "Medicare and Medicaid Programs") and the TRICARE Program (the Medicare and Medicaid Programs, the TRICARE Program, and such other "Federal health care programs" (as defined in 42 U.S.C. § 1320a-7b(f)), are referred to collectively as the "Governmental Programs"), (ii) currently participates in each Governmental Program in which it is enrolled pursuant to a provider agreement, if required by the applicable Governmental Program (the "Provider Agreements") and receives payments from private, non-governmental programs (including any private insurance program) (such private, non-governmental programs are referred to collectively as "Private Programs"), (iii) is, to the Knowledge of Seller, in good standing with the Governmental Programs and material Private Programs, and (iv) has, to the Knowledge of Seller, no outstanding overpayments or refunds due to Governmental Programs or Private Programs, or asserted by a Governmental Authority or Private Program, individually in excess of \$150,000, or in the aggregate in excess of \$500,000 except as disclosed in Section 3.5(d)(i) of the Disclosure Schedules. Except as disclosed in Section 3.5(d)(i)(w) of the Disclosure Schedules, during the past three years Seller has timely filed all material claims and reports required to be filed prior to the Effective Date with respect to the Governmental Programs and Private Programs, all fiscal intermediaries and/or carriers, and other insurance carriers ("Payor Claims"). All Payor Claims that have been filed during the past three years have been complete, accurate and in compliance with applicable Laws in all material respects. Except as set forth in Section 3.5(d)(i)(x) of the Disclosure Schedules, Seller has paid or caused to be paid all known and undisputed refunds and overpayments that have become due pursuant to Payor Claims. Other than any refund, overpayment, discount or adjustment that occurs in the Ordinary Course of Business, Seller has not during the past three years claimed or received reimbursements from Governmental Programs or Private Programs in excess of the amounts permitted by applicable Laws, and has no Liability under any Governmental Program or Private Program. Except as disclosed in Section 3.5(d)(i)(y) of the Disclosure Schedules, there are no pending or, to Knowledge of Seller, threatened proceedings, investigations, or audits relating to Seller's participation in any Governmental Program or Private Program, other than in the Ordinary Course of Business. Neither Seller nor Seller Parent is subject to, nor has Seller or Seller Parent been subject to in the last twelve (12) months, any material non-routine pre-payment utilization review activities from Governmental Programs. Neither Seller nor Seller Parent has been excluded from participation in any

Governmental Program or any material Private Program. During the past three years, all billing practices of Seller and Seller Parent with respect to all Governmental Programs and Private Programs related to the Business have been in material compliance with all Laws applicable to Seller and Seller Parent.

- (e) Section 3.5(e) of the Disclosure Schedules lists all financial relationships (whether or not memorialized in writing), including any joint venture, partnership, co-ownership or other arrangement involving any ownership or investment interest, that Seller has had in the last five years, with any individual that, to the Knowledge of Seller, is a physician who refers or has referred to Seller or Seller Parent for laboratory testing (a "Referring Physician"), or family member of a Referring Physician, by the Business, and all such relationships are in compliance in all material respects with the Anti-Kickback Statute (as defined below) and the Stark Law. For purposes of this Section 3.5(e) and Section 3.7(c), the term "financial relationship" has the meaning set forth in 42 U.S.C. § 1395nn and the regulations promulgated thereunder.
- (f) Except as set forth in <u>Section 3.5(f) of the Disclosure Schedules</u>, neither Seller nor Seller Parent nor, to the Knowledge of Seller, any director, officer, employee or agent of Seller or Seller Parent:
 - (i) has been convicted of or charged with any violation of any Laws related to any Governmental Program;
 - (ii) has been convicted of or charged with any violation of Laws related to fraud, theft, embezzlement, breach of fiduciary responsibility, financial misconduct, obstruction of an investigation, or controlled substances; or
 - (iii) is excluded, suspended or debarred from participation, or is otherwise ineligible to participate, in any Governmental Program.
- (g) In connection with the operation of the Business, Seller and Seller Parent, including with respect to their respective systems and operations and Material Contracts are not now, and have not during the past three years been, in default or violation of, (i) the Federal Health Care Program Anti-Kickback Statute, 42 U.S.C. § 1320a-7b(b), and all regulations promulgated thereunder (known as the "Anti-Kickback Statute"), (ii) the Stark Law, and all regulations promulgated thereunder, (iii) the Federal False Claim Act, 31 U.S.C. § 3729, and all regulations promulgated thereunder, (iv) the Occupational Safety and Health Act, and all regulations promulgated thereunder that apply to the Business (known as "OSHA"), (v) the Federal Food, Drug and Cosmetic Act (the "FD&C Act"), 21 U.S.C. § 301 et seq., and all regulations promulgated by the United States Food and Drug Administration thereunder (together with the FD&C Act, the "FDA Law and Regulation"), (vi) CLIA, (vii) applicable Laws of the United States Drug Enforcement Administration ("DEA"), (viii) state anti-kickback, fee-splitting and patient brokering Laws, and (ix) state Laws governing self-referral and the licensure and operation of clinical laboratories, in each case, as applicable to the Business; and
- (h) To the Knowledge of Seller, there is no civil, criminal or administrative action, suit, demand, claim, complaint, hearing, investigation, demand letter, warning letter, proceeding or request for information pending against Seller associated with the Acquired Assets, where Seller or Seller Parent has material Liability (whether actual or contingent), for failure to comply with FDA Law and Regulation.
- 3.6 Absence of Changes. In the 12-month period prior to the date of this Agreement, (a) Seller and Seller Parent have operated the Business in the Ordinary Course of Business, (b) except as disclosed on Section 3.6 of the Disclosure Schedules, other than in the Ordinary Course of Business, none of Seller or Seller Parent has taken any action that would have required the consent of Buyer under Section 5.1 of this Agreement had such action or failure to act occurred between the date of the Agreement and the Closing and (c) no event, occurrence, fact, condition, change, effect or circumstance has occurred which, individually or in the aggregate, has had or would reasonably be expected to have, a Material Adverse Effect.

3.7 Customers, Tests, Payors; Suppliers and Supplies.

(a) Section 3.7(a) of the Disclosure Schedules sets forth (1) a list of the top customers of the Business (which customers represent 80% of the sales of the Business) ("Material Customers"), and (2) a list of the top ten payors of the Business (by volume of sales from such payors) and an indication of whether they are in network or out-of-network, in each case for the 2021 and 2022 fiscal years, and for the six-month period ended January 31, 2023 ("Material Payor"). Section 3.7(a)(i) of the Disclosure Schedules accurately sets forth, with

respect to each Material Customer, the information set forth in Annex 3.7(a)(i) with all fields accurately completed (such information, excluding the monthly test code order volume, the "Client Setup Data"). Neither Seller nor Seller Parent has received any written or, to the Knowledge of Seller, other notice from any Material Customer or Material Payor to the effect that it will, and to the Knowledge of Seller, no Material Customer or Material Payor intends to, cease doing business with or materially diminish the amount or terms (whether related to payment, price or otherwise) of the business that it is currently doing with Seller or Seller Parent with respect to the Business or, after the Closing Date, will be doing with Buyer (whether as a result of the consummation of the transactions contemplated hereby or otherwise). Section 3.7(a)(ii) of the Disclosure Schedules accurately sets forth, with respect to each such Material Customer the client and patient pricing information of the Business in the form set forth on Annex 3.7(a)(ii). Section 3.7(a)(iii) of the Disclosure Schedules accurately sets forth the name of each customer of the Business as of the Effective Date and the percentage of Seller's total revenue of the Business for the 12-month period ended December 31, 2022, attributable to each such customer. At the Effective Date and periodically thereafter as set forth in Section 5.4(a), Section 3.7(a)(iv) of the Disclosure Schedules accurately sets forth the name of each customer by practice of the Business as of the Effective Date and the open Standing Orders and Hold Orders, attributable to each such practice, as of each such date (including to reflect any new customer accounts gained since the previous date of delivery)

(b) Section 3.7(b) of the Disclosure Schedules sets forth a list of the top 10 vendors of the Business (by volume of revenue paid to such suppliers for the 2021 and 2022 fiscal years, and for the one-month period ended January 31, 2022) ("Material Vendors"). Since January 1, 2022, Seller has not received any notice from, and to the Knowledge of Seller, no Material Vendor to the effect that any such Material Vendor intends to cease doing business with or materially diminish the amount or terms (whether related to payment, price or otherwise) of the business that it is currently doing with Seller or Seller Parent with respect to the Business or, after the Closing Date, will be doing with Buyer (whether as a result of the consummation of the transactions contemplated hereby or otherwise).

(c) Neither Seller nor Seller Parent provides, and has not for the past three years, provided, to its customers of the Business any supplies other than those supplies listed in Section 3.7(c)(i) of the Disclosure Schedules, none of which are excluded supplies per the NY Guidelines, and provides such supplies in compliance in all material respects with Law and with applicable guidance set forth by an applicable Governmental Authority, including the guidelines attached hereto (the "NY Guidelines"). Other than as specifically set forth in Section 3.7(c)(ii) of the Disclosure Schedules, Seller and Seller Parent each does not provide, and has not for the past three years provided, any services, items, supplies, remuneration or any other benefit to any customer or potential customer of the Business and there are no other financial relationships with any Customer or potential customer of the Business, including any consulting arrangement or similar engagement.

3.8 Good Title to and Use of Assets.

(a) Seller (and not its Affiliates) has sole and exclusive, good and marketable title to, or, in the case of property held by lease under Contract, an exclusive and valid leasehold interest in, and right to use, all of the tangible Acquired Assets, free and clear of any Liens other than Permitted Liens, and, subject to obtaining the Consents set forth in Section 3.3(c) of the Disclosure Schedules, each of Seller and Seller Parent has the authority and right to, and at the Closing, shall, sell, assign, transfer and deliver to Buyer title to the Acquired Assets free and clear of all Liens, other than Permitted Liens.

(b) The Acquired Assets are (to the extent applicable) in good operating condition (ordinary wear and tear excepted) and are adequate for the uses to which they are being put, and none of such tangible Acquired Assets is in need of maintenance or repairs except for ordinary, routine maintenance and repairs. Except for the Excluded Assets and Intellectual Property (the sufficiency of which is addressed solely with respect to Section 3.28 below) the Acquired Assets comprise all of the assets, properties, Contracts and rights utilized by Seller and Seller Parent in connection with the conduct of the Business and are sufficient for Buyer to operate the Business following the Closing in substantially the manner as operated by Seller and Seller Parent prior to the Closing.

3.9 Real Property.

(a) Section 3.9(a)(x) of the Disclosure Schedules sets forth a true, complete and correct list (with addresses) of the locations at which the Business is operated subject to a lease, sublease or other occupancy agreement, including any and all STAT laboratory locations and patient service centers used in the Business and

any other locations where Testing with respect to the Business is performed, and for each location listed indicates whether, to the Knowledge of Seller, it is directly or indirectly owned by a physician or by an immediate family member of a physician (collectively, the premises at such locations at which the Business is operated, the "Leased Real Property"). Neither Seller nor Seller Parent owns any real property at which the Business is operated. Except as set forth on Section 3.9(a)(y) of the Disclosure Schedules, Seller (or its applicable Affiliate) has an exclusive right to lease, sublease, or occupy, and enjoys peaceful and undisturbed possession of the Leased Real Property, and, subject to the terms of the lease (or sublease, or occupancy agreement, as applicable) (collectively, the "Real Property Leases") demising such Leased Real Property, Seller (or its applicable Affiliate) has the exclusive right to assign such right. Seller has delivered to Buyer a true and complete copy of each Real Property Lease. Except as set forth on Section 3.9(a)(z) of the Disclosure Schedules, with respect to each of the Assigned Real Property Leases: (1) Seller (or its applicable Affiliate) has a valid and enforceable leasehold interest in each Assigned Location; (2) none of Seller or any of its Affiliates (as applicable) have received written notice of any existing defaults thereunder by Seller or its Affiliate (as applicable) that remains outstanding nor, to the Knowledge of Seller, are there any existing defaults by the lessor thereof; (3) to the Knowledge of Seller, no event has occurred which (with notice, lapse of time or both) would constitute a breach or default thereunder by any of Seller or its Affiliates (as applicable) or, to the Knowledge of Seller, any other party thereto; (4) Seller or its Affiliates (as applicable) has not subleased or otherwise granted to any person rights to use, enjoy or occupy any portion of the Assigned Locations; (5) to the Knowledge of Seller, there is no pending or threatened (in writing) condemnation or similar proceeding affecting the Assigned Locations and, to the Knowledge of Seller, the Assigned Locations are not subject to any rights of first refusal, first offer, purchase options or rights of occupancy except as set forth in the applicable Assigned Real Property Lease; and (6) no party to any Assigned Real Property Lease has exercised any termination or cancellation rights with respect thereto and no party has given notice of any significant dispute with respect to any Real Property Lease.

- (b) Except as set forth on Section 3.9(b) of the Disclosure Schedules, all buildings, structures, fixtures and other improvements located on the Assigned Locations for which Seller or any of its Affiliates (as applicable) is responsible are in good operating condition and repair and are adequate for its intended uses and purposes in the Ordinary Course of Business and other uses currently conducted by Seller or any of its Affiliates (as applicable) at such location.
- (c) Seller does not own, and has never owned, any fee interest in any real property used by it in connection with the Business.

3.10 Material Contracts.

- (a) <u>Section 3.10(a)</u> of the <u>Disclosure Schedules</u> sets forth a true, complete and correct list (subsections of the <u>Disclosure Schedules</u> are numbered to correspond to the subsections of this <u>Section 3.10(a)</u>) of all of the following Contracts to the extent such Contracts are exclusively used in the Business, or will otherwise be Assigned Contracts, and to which Seller or any of its Affiliates is a party:
 - (i) all Contracts that relate to equipment or personal property located at the Assigned Locations, the purchase of sale or inventory, accounts receivable, raw materials, commodities, supplies, goods, products, or for the furnishing or receipt of services, in each case, the performance of which will extend over a period of more than six months from the date of such Contract or that involve payments to or from Seller anticipated to be in excess of \$50,000 in any one calendar year;
 - (ii) all Contracts with customers, payors or vendors that involve payments to or from Seller or its Affiliate anticipated to be in excess of \$50,000 in any one calendar year;
 - (iii) all Contracts with other service providers related to the collection of laboratory tests and forwarding such results for Testing, that involve payments to or from Seller anticipated to be in excess of \$50,000 in any one calendar year;
 - (iv) any Contracts with any vendor that provides services relating to billing, coding and/or reimbursement;
 - (v) all Contracts with IOPs (each of which agreements describes the services being provided by the IOP), that involve payments to or from Seller or its Affiliate anticipated to be in excess of \$50,000 in any one calendar year;

(vi) all Contracts pursuant to which Seller or its Affiliate (x) licenses any Intellectual Property to another Person and (y) licenses any Intellectual Property from another Person, except for agreements pertaining to commercially available, mass-market software, that involve payments to or from Seller anticipated to be in excess of \$50,000 in any one calendar year, licenses that are merely appurtenant to Contracts for purchased or leased equipment to permit the intended use of such equipment, or licenses granted to customers, contractors, consultants, or vendors in the Ordinary Course of Business:

(vii) the Real Property Leases;

- (viii) any Contracts (pending or executed) relating to the acquisition or disposition of (A) any business of Seller (whether by merger, consolidation or other business combination, sale of securities, sale of assets or otherwise), (B) any material asset of the Business other than in the Ordinary Course of Business, or (C) a series or group of related transactions or events of the type specified in clauses (A) and (B) above;
- (ix) any Contract that is a partnership, limited liability company, or joint venture agreement with another Person;
- (x) any material Contract (or group of related Contracts) under which Seller has permitted any Acquired Asset to become subject to any Lien, except Permitted Liens;
- (xi) any Contracts under which Seller or its Affiliate is, or may become, obligated to incur any material severance pay or special compensation obligations which would become payable, increased or accelerated by reason of the Transaction Documents or the transactions contemplated hereby or thereby;
- (xii) any collective bargaining agreement, memorandum of understanding, side letter, neutrality, or card check agreement, profit sharing, defined compensation, severance or other plan or arrangement for the benefit of Seller's current or former directors, officers and employees;
- (xiii) any Contracts contemplating annual payments in excess of \$100,000 per year providing for the employment or consultancy with an individual on a full-time, part-time, consulting or other basis to any officer, director, employee or consultant (other than an Employee Plan), or any Contract pursuant to which Seller has an obligation to pay a transaction-related bonus or severance to any individual or any other payments that would become payable, directly or indirectly, by reason of execution of the Transaction Documents or the consummation of the transactions contemplated hereby and thereby:
- (xiv) any material agency, dealer, distributor, sales representative, marketing or other similar agreement;
- (xv) any Contract which contains an option, grants any right of first refusal, right of first negotiation, or right of first offer in favor of any Person;
- (xvi) except as may be entered into in the Ordinary Course of Business, any warranty Contract with respect to products sold or indemnity agreements with any supplier or customer to the Business under which Seller or its Affiliate is obligated to indemnify such supplier or customer against product liability claims;
 - (xvii) any Contract with any Governmental Authority;
- (xviii) all Contracts subjecting the Business to non-competition or other similar material restrictions on its ability to conduct the Business anywhere in the world, containing any material exclusivity, "most favored customer," or other similar restrictions on any party thereto, or pursuant to which the Business has the benefit of material non-competition, confidentiality or other similar restrictions by another Person; and
- (xix) all other Contracts (or group of related Contracts) included in the Assigned Contracts that involve payments to or from Seller or its Affiliate anticipated to be in excess of \$50,000 in any one calendar year or the performance of which involves consideration in excess of \$50,000 over the life of such Contract.
- (b) Except as set forth in Section 3.10(b) of the Disclosure Schedules, none of the Material Contracts materially limit Seller or any assignee thereof in any way from competing anywhere or from soliciting any Person as a customer, client, payor or employee, or require Seller to refer any Testing to any third party. Each

Material Contract is legal, valid and binding, in compliance with all applicable Laws, in full force and effect, and enforceable against Seller in accordance with its terms, and, to the Knowledge of Seller, the other parties thereto, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity). Seller is not in material breach or default (with or without notice or lapse of time, or both) under any Material Contract and, to the Knowledge of Seller, no other party to any of the Material Contracts has materially breached or defaulted thereunder or has repudiated any provision thereof. To the Knowledge of Seller, no event has occurred or circumstance exists that (with or without notice or lapse of time, or both) would result in a material breach of, or give Seller or other Person the right to declare a material default or exercise any remedy under, or accelerate the maturity or performance of or payment under, or cancel, terminate or modify, any Material Contract. As of the Closing, Seller will have made all required payments under leases included in the Material Contracts, unless validly waived or otherwise satisfied. Seller has delivered to Buyer true, accurate and complete copies of all written Material Contracts (or, with respect to oral Material Contracts, a written summary), together with all amendments or modifications thereto.

- (c) Except as set forth on Section 3.10(c) of the Disclosure Schedules, Seller has not, and to the Knowledge of Seller, no party to any Material Contract has, exercised any termination rights with respect thereto
- (d) Section 3.10(d) of the Disclosure Schedules lists any Contract pursuant to which Seller and/or any of its Affiliates provides or receives material goods or services (including licenses) that are related to both the Business and any other business operated by Seller or any of its Affiliates.
- (e) As used herein, "*Material Contracts*" means all of the Contracts of the type required to be listed in <u>Section 3.10(a) of the Disclosure Schedules</u>, whether or not listed in the Disclosure Schedules.
- **3.11** <u>Bankruptcy</u>. Seller (a) is not in receivership or dissolution, (b) has not made any assignment for the benefit of creditors, (c) has not admitted in writing its inability to pay its Debts as they mature, (d) has not been adjudicated as bankrupt, or (e) has not filed a petition in voluntary bankruptcy, a petition or answer seeking reorganization, or an arrangement with creditors under federal bankruptcy Law or any other similar Law of the United States or any state thereof, nor has any such petition been filed against Seller.
- 3.12 <u>Prepaid Expenses</u>. <u>Section 3.12 of the Disclosure Schedules</u> sets forth a true, complete and correct list of all reserves, security and other deposits, advances and prepaid expenses and credits related to the Business. Each reserve, security or other deposit, advance or prepaid expense or credit listed on <u>Section 3.12 of the Disclosure Schedules</u> assignable to Buyer in connection with the transactions contemplated by this Agreement without the consent or approval of any other Person.
- 3.13 Insurance. Section 3.13 of the Disclosure Schedules sets forth a true, complete and correct list and description (showing type, number, insurer name and date of expiration) of all current insurance policies or binders, including policies by which the Business, any of the Acquired Assets, Assumed Liabilities or the Business Employees are insured, and any fire, liability, product liability, umbrella liability, real and personal property, workers' compensation, vehicular, fiduciary liability and other casualty and property insurance maintained by Seller or its Affiliates relating to the Business, the Acquired Assets or the Assumed Liabilities (collectively, the "Insurance Policies"). The description includes for each Insurance Policy the type of policy, policy number, name of insurer and date of expiration. Seller has made available to Buyer true, accurate and complete copies of all Insurance Policies, in each case, as amended or otherwise modified and in effect. Each Insurance Policy is in full force and effect on the date of this Agreement and all premiums due on such Insurance Policies have been paid. There are no claims related to the Business, the Acquired Assets or the Assumed Liabilities pending under any such Insurance Policies as to which coverage has been denied or disputed and during the past five years no insurer has denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Insurance Policy or, to Knowledge of Seller, has threatened in writing to cancel any Insurance Policy with respect to the Business. Neither Seller nor its Affiliates are in default, in any material respect, with respect to its obligations under any insurance policy relating to the Business maintained by it, and Seller has not been denied insurance coverage or been subject to any gaps in insurance coverage with respect to the Business, Acquired Assets or Assumed Liabilities in the last two years. Neither Seller nor any of its Affiliates have received any written notice of cancellation of, premium increase with respect to, or alteration of coverage under, any of such Insurance Policies. Neither Seller nor any of its Affiliates have any selfinsurance or co-insurance programs relating to the Business, the Acquired Assets or the Assumed Liabilities.

3.14 Litigation; Governmental Orders.

- (a) Except as disclosed on Section 3.14(a)(i) of the Disclosure Schedules, there is no Action to which Seller, Seller Parent or any Business Employee or any director or officer of Seller or Seller Parent (in their capacity as such) is, or in the last five years has been, a party (either as plaintiff or defendant) or, to the Knowledge of Seller, threatened, in each case, which materially and adversely affects the Business, the Acquired Assets or the Assumed Liabilities or in any manner challenges or seeks the recission of, or seeks to prevent, enjoin, alter, materially delay the consummation of, or otherwise relates to, the Transaction Documents or the transactions contemplated hereby or thereby. Except as disclosed on Section 3.14(a)(ii) of the Disclosure Schedules, there is no Action which Seller presently intends to initiate. Except as described on Section 3.14(a) (iii) of the Disclosure Schedules, Seller is not subject to any arbitration proceedings under collective bargaining agreements.
- (b) Except as disclosed on <u>Section 3.14(b) of the Disclosure Schedules</u>, no Governmental Order is, or in the last five years has been, issued or, to the Knowledge of Seller, is threatened which is applicable to Seller, the Acquired Assets or the Business, nor any unsatisfied judgments or awards against, relating to or affecting the Acquired Assets or the Business.
- **3.15 Brokers or Finders.** No agent, finder, broker, investment banker, financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or agent's commission or any other similar commission or fee in connection with any of the transactions contemplated by the Transaction Documents, based upon arrangements made by or on behalf of Seller or Seller Parent or any of their Affiliates, for which Buyer or any of its Affiliates could have any liability or responsibility whatsoever.
- 3.16 <u>Disqualifying Background</u>. Neither Seller nor Seller Parent has ever been debarred, suspended or otherwise excluded from participating in any state or federally funded healthcare program.

3.17 Privacy and Data Security; Information Technology.

- (a) Seller has been during the past six years and is in material compliance with Information Privacy and Security Laws. In addition, Seller has been during the past three years and is in material compliance with (i) the terms of all Contracts to which it is a party setting forth Personal Information privacy, security, or breach notification requirements or restrictions (including provisions that impose conditions or restrictions on the receipt, collection, monitoring, maintenance, creation, transmission, processing, use, analysis, disclosure, storage, transfer, or disposal and security of Personal Information); (ii) the Privacy Policies; (iii) each approval, consent, ratification, permission, waiver, order or authorization received by Seller from any Governmental Authority relating to the privacy or security of Personal Information received, collected, created, or maintained by Seller; (iv) to the extent applicable to Seller, PCI DSS; and (v) the terms of all consents and authorizations obtained by Seller or provided to Seller from a Customer and that apply to Seller's receipt, use, processing or disclosure of Personal Information in connection with the Business (collectively as set forth in subsection (v), "PII Consents"). Seller has entered into all agreements it is required to enter into by Information Privacy and Security Laws in connection with the collection, receipt, access, transfer, processing, use and disclosure of Personal Information, including any BAAs required by HIPAA where Seller acts as the "covered entity" or the upstream "business associate" under such BAA (as each term is defined under HIPAA at 45 C.F.R. § 160.103) and any data processing agreements or standard contractual clauses governing the international transfer of Personal Information required by the EU General Data Protection Regulation as applicable to Seller.
- (b) Seller has established, and is and has been during the past three years in material compliance with, a reasonable written information security program that: (i) implements reasonable and appropriate administrative, technical and physical safeguards that protect the security, confidentiality, and integrity of all IT Assets and all Personal Information in the Acquired Assets; (ii) is reasonably designed to prevent unauthorized access, use or disclosure of Personal Information in the IT Assets and Acquired Assets; (iii) materially complies with Information Privacy and Security Laws and industry standards related to the industry in which the Business operates; (iv) includes reasonable policies and procedures that apply to Seller with respect to privacy, data protection, processing, security and the collection and use of Personal Information in the IT Assets and Acquired Assets gathered or accessed in the operation of the Business; (v) reasonably identifies threats to the confidentiality or security of Personal Information in the Acquired Assets and intrusions into IT Assets that Process Personal Information; and (vi) to the extent applicable to Seller, complies with PCI DSS. Seller takes and have at all times in the past three years taken, in connection with the Business, all reasonable steps to require that any third parties

acting on behalf of Seller that Process Personal Information in the Acquired Assets provide similar reasonable and appropriate safeguards, in each case, in compliance with applicable Information Privacy and Security Laws and consistent with industry standards related to the industry in which the third parties operate.

- (c) The Business has not suffered or incurred: (i) during the past six years, a "breach of unsecured protected health information" or "security incident," as such terms are defined by HIPAA, or (ii) during the past six years, a "breach" or "security breach" (or similar terms such as "breach of security of the system"), as defined by any Information Privacy and Security Laws other than HIPAA, in the case of (i) and (ii) involving any Personal Information owned, transmitted, used, stored, received, or controlled by or on behalf of Seller. During the past six years, there have been no Actions conducted or, to the Knowledge of Seller, asserted, and to the Knowledge of Seller no Actions are pending, against Seller by any Person (including any Governmental Authority) alleging violation of Information Privacy and Security Laws, and to the Knowledge of Seller, no such Actions have been threatened. During the past six years, Seller has not received any written notices from the U.S. Department of Health and Human Services Office for Civil Rights, Department of Justice, Federal Trade Commission, or the Attorney General of any state, or any equivalent foreign Governmental Authority, relating to any such violations. Seller is not or has not during the past three years been required under any PII Consent to notify any Person, including any Governmental Authority, of the loss, or unauthorized or unlawful access, use or disclosure, of any Personal Information of such Person.
- (d) Business Employees who have access to Personal Information have received training with respect to compliance with Information Privacy and Security Laws as required under Information Privacy and Security Laws.
- (e) Seller has performed security risk assessments that materially meet (as applicable): (i) the standards set forth at 45 C.F.R. § 164.308(a)(1)(ii)(A)-(B) and an assessment as required by 45 C.F.R. 164.306(d)(3), taking into account the factors set forth in 45 C.F.R. 164.306(a), (b), and (c) and created and maintained documentation of the foregoing assessments in accordance with 45 C.F.R. 164.316; and (ii) all other Information Privacy and Security Laws (collectively, the "Security Risk Assessments"), and Seller has implemented reasonable security measures necessary to mitigate or remediate, to reasonable and appropriate levels in accordance with Information Privacy and Security Laws and prevailing industry standards in which the Business operates, all risks and vulnerabilities identified in every Security Risk Assessment as being of "high," "critical" or any similar level of severity.
- (f) Seller's (i) collection, storage, processing, transfer, sharing and destruction of Personal Information in connection with the transactions contemplated by this Agreement and (ii) execution, delivery and performance of this Agreement and the Transaction Documents, and the consummation of the transactions contemplated hereby and thereby, materially comply with Seller's Privacy Policies and materially comply in all respects with Information Privacy and Security Laws.
- (g) The IT Assets are maintained and in good working condition to effectively perform all computing, information technology, and data processing operations reasonably necessary for the operations of the Business. There are no substantial alterations, modifications, or updates to the IT Assets intended or required currently or, to the Knowledge of Seller, that will be required in the near future for the operations of the Business. Seller has taken reasonable steps and implemented reasonable safeguards to: (i) protect the IT Assets from any disabling codes or instructions, spyware, Trojan horses, worms, trap doors, backdoors, Easter eggs, logic bombs, time bombs, cancelbots, viruses, or other software or programming routines that permit or cause (or are intended to permit or cause) unauthorized access to, or disruption, impairment, modification, recordation, misuse, transmission, disablement, or destruction of, the IT Assets, (ii) maintain the integrity and continued operations of the IT Assets, and (iii) provide for the remote site back up of data and information material to IT Assets (including such data and information that is stored on magnetic or optical media in the Ordinary Course of Business) in a reasonable attempt to avoid disruption or interruption to the Business. To the Knowledge of Seller, there has been no failure, breakdown, or continued substandard performance of any IT Assets that has caused a material disruption or interruption to the operation of the Business.

3.18 Tax Matters.

(a) Seller and Seller Parent have filed all income and other material Tax Returns that they were required to file with respect to the Business or that relate to the Acquired Assets or the Assumed Liabilities, and all such Tax Returns are true, correct and complete in all material respects. Seller and Seller Parent have paid all

Taxes with respect to the Business or that relate to the Acquired Assets or the Assumed Liabilities due and payable or claimed by any Governmental Authority to be due and payable (whether or not shown on any Tax Return) by them.

- (b) All amounts required to be withheld by Seller or Seller Parent with respect to amounts paid or owing to any Person (including from employees for income Taxes and social security and other payroll Taxes) and all sales and use Taxes, value-added Taxes and ad valorem Taxes (including personal property taxes) required to be collected by Seller or Seller Parent, in each case with respect to the Business or that relate to the Acquired Assets or the Assumed Liabilities, have been collected or withheld and paid or remitted to the applicable Governmental Authorities, and all Tax Returns required with respect thereto (including IRS Forms W-2 and 1099) have been properly completed and timely filed with the applicable Governmental Authority.
- (c) None of the Acquired Assets are subject to any Liens for Taxes, other than Liens for Taxes not yet due and payable.
- (d) No transaction contemplated by this Agreement is subject to withholding under Section 1445 of the Code.
- (e) None of the Acquired Assets are (i) required to be treated as being owned by another Person pursuant to the so-called "safe harbor lease" provisions of former Section 168(f)(8) of the Internal Revenue Code of 1954, (ii) subject to Section 168(g)(1)(A) of the Code, (iii) subject to a disqualified leaseback or long-term agreement as defined in Section 467 of the Code, or (iv) "tax-exempt use property" within the meaning of Section 168(h) of the Code.

3.19 Labor Matters.

- (a) Neither Seller nor Seller Parent is, or has been in the past three years, a party to any collective bargaining or other similar labor agreement with respect to any Business Employees, nor, to the Knowledge of Seller, are there, nor have there been in the past three years, any organizational efforts, such as union organizing campaigns, or labor organization certifications or recognition, with respect to the formation of a new collective bargaining unit presently being made or, to the Knowledge of Seller, threatened involving the Business Employees. There are, and in the past three years have been, no strikes, slowdowns, lockouts, pickets, work stoppages or material labor disputes pending or, to the Knowledge of Seller threatened against Seller or Seller Parent with respect to the Business.
- (b) There are, and in the past three years have been no unfair labor practice charges or complaints, or other grievances, against Seller or Seller Parent threatened or pending before the National Labor Relations Board, the Equal Employment Opportunity Commission, the Department of Labor or any other Governmental Authority. Neither Seller nor Seller Parent is a party to or otherwise bound by any consent decree with or citation from any Governmental Authority relating to employees or employee practices with respect to the Business.
- (c) Within the past four years, Seller and Seller Parent have each been in material compliance with all applicable Laws pertaining to employees, employment matters, employment practices and termination of employment, to the extent related to current Business Employees, including all such applicable Laws and decrees and orders of any court or Governmental Authority applicable to the Business relating to terms and conditions of employment, wages, salary, overtime, hours, breaks, eligibility for and payment of overtime compensation, worker classification (including classification of employees as "exempt" or "nonexempt" and classification of independent contractors), child labor, immigration, equal opportunity, employment discrimination and harassment, retaliation, employment eligibility, immigration, disability rights, fair labor practices, unemployment insurance, affirmative action, leaves of absence (including paid sick leave), civil rights, occupational safety and health, workers' compensation, and the collection and payment of withholding of Social Security Taxes and similar Taxes, except as would not reasonably be likely to result in any material Liability to Buyer. Seller and Seller Parent operate and have operated the Business for the four years preceding the date of this Agreement in compliance with all applicable wage and hour, worker classification Laws, Each Business Employee is (i) a United States citizen, (ii) a United States national, (iii) a lawful permanent resident of the United States, or (iv) an alien authorized to work in the United States either specifically for Seller or for any United States employer.
- (d) Neither Seller nor Seller Parent is a party to a settlement agreement, tolling agreement, non-disparagement agreement, confidentiality agreement, non-disclosure agreement, separation agreement, resolution agreement or any contract or provision similar to any of the foregoing, in the prior six years with a

current or former officer, Business Employee or independent contractor of the Business that releases and settles specific allegations asserted by such former officer, Business Employee or independent contractor relating to discrimination (including age, gender, race, religion or any other legally protected category), sexual harassment or sexual misconduct by any officer or executive level employee of Seller or Seller Parent; provided, that for the sake of clarity, agreements which include releases of general applicability shall not be considered allegations relating to discrimination, harassment, or sexual misconduct for purposes of this Agreement. To the Knowledge of Seller, in the last three years, no allegations, complaints, charges or claims (formal or informal) of discrimination (including age, gender, race, religion or any other legally protected category), sexual harassment, sexual misconduct, or similar behavior have been made or threatened against any officer, director, manager, or supervisory-level employee of Seller or Seller Parent or any executive level employee of Seller or Seller Parent that have not been resolved without further actual or potential Liability to Buyer.

- (e) Seller and Seller Parent have complied in all material respects with all applicable Laws with respect to the COVID-19 Pandemic and has not received any written notice asserting any material employment-related liability with respect to the COVID-19 Pandemic.
- (f) Section 3.19(f)(i) of the Disclosure Schedules lists each of the Business Employees, including whether such employee works primarily or exclusively for the Business and such employee's: (i) employee ID number, (ii) employing entity, (iii) job title or position, (iv) location (city, state), (v) address, (vi) salary or other rate of base pay, if applicable, (vii) other compensation targets or arrangements currently in place for 2023 (including bonus target, incentive compensation target, deferred compensation, and commissions arrangements), (viii) date of hire, (ix) status as an exempt or non-exempt employee under the Fair Labor Standards Act and any similar wage and hour law, (x) status as a full-or part-time employee, (xi) status as an active or inactive employee (and, if on leave, the nature of the leave and the expected return date, if known) and (xii) severance entitlement under existing agreements, if any. Section 3.19(f)(ii) of the Disclosure Schedules lists all individual consultants currently hired by Seller or Seller Parent to perform services for the Business, the services provided by each individual, their rate of pay, their location, the date when they began their assignment with Seller or Seller Parent, and the estimated completion date of their services, if any.
- **3.20 Telephone Numbers.** <u>Section 3.20 of the Disclosure Schedules</u> sets forth a true, complete and correct list of telephone and facsimile numbers used exclusively in the Business to contact the Assigned Locations.
- **3.21 WARN Act**. Except as set forth on <u>Section 3.21 of the Disclosure Schedules</u>, within the past three years, Seller has not implemented any relocation, plant closing, mass layoff of employees, or taken any action that, either alone or in the aggregate that will require Buyer or any of its Affiliates to give notice or otherwise comply with WARN or any similar foreign, state or local Law.

3.22 Financial Statements.

- (a) Section 3.22 of the Disclosure Schedules sets forth the following financial statements of the Business: (i) the unaudited balance sheets of the Business as of December 31, 2022 and 2021 and the related unaudited statements of income for each of the 12-month periods then ended, (ii) the unaudited balance sheet of the Business as of January 31, 2023 (the "Balance Sheet Date") and the related unaudited statement of income for the one-month period then ended, and (iii) the audited consolidated balance sheets of Seller Parent as of July 31, 2022, and 2021 and the related audited consolidated statements of operations, income, stockholders' equity and cash flows for each of the 12-month periods then ended and the 12-month period ended July 31, 2022 (collectively, the "Financial Statements"). The Financial Statements (including any notes thereto) have been prepared from and in accordance with the books and records of Seller in accordance with GAAP applied on a consistent basis throughout all the periods involved, and fairly present, in all material respects, the financial condition and results of operations of the Business and changes in financial condition for the periods specified. The Financial Statements are consistent with the books of account and minute books of Seller, are complete and correct in all material respects, have been maintained in accordance with applicable Laws, and accurately and fairly reflect, in reasonable detail, the activities of Seller with respect to the Business.
- (b) Seller Parent maintains a standard system of accounting established and administered in accordance with GAAP and a system of internal controls that are effective in providing assurance regarding the reliability of financial reporting and the preparation of financial statements in accordance with GAAP, including policies and procedures that (i) require the maintenance of records that in reasonable detail accurately and fairly

reflect the transactions and dispositions of Seller's assets and properties, (ii) provide assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that receipts and expenditures of Seller and the Business are being made only in accordance with appropriate authorizations of management and the board of directors of Seller, and (iii) provide assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of Seller's assets and properties.

- (c) All final data that Seller has provided to Buyer with respect to historical test utilization, Current Procedural Terminology ("*CPT*") codes, payor CPT detail, requisition volumes, rental payments, cash collections, and full time employees is true, accurate, and complete in all material respects.
- 3.23 Motor Vehicles. Section 3.23 of the Disclosure Schedules sets forth a true, complete and correct list of all the motor vehicles that are primarily used or held for use in connection with the Business, including the manufacturer, model and year of each vehicle, and all such motor vehicles are owned (and not leased) by Seller and there are no outstanding loans on any such motor vehicles. Neither Seller nor Seller Parent is party to any lease agreement, installment contract or other Contract with respect to any motor vehicles primarily used or held for use in connection with the Business.
- 3.24 Absence of Undisclosed Liabilities. Neither Seller nor Seller Parent has material obligations or Liabilities of any nature with respect to the Business other than: (a) those reflected on or adequately reserved against in the Financial Statements in accordance with GAAP; and (b) those incurred in the Ordinary Course of Business since the Balance Sheet Date that are not material in amount or significance or did not arise in connection with a breach of Contract or Permit, breach of warranty, tort or infringement or violation of Law (without giving effect to the proviso in the definition thereof).
- 3.25 Information Supplied. None of the information supplied or to be supplied by Seller or Seller Parent for inclusion or incorporation by reference in the proxy statement relating to the Stockholder Meeting (such proxy statement, as amended or supplemented from time to time, the "Proxy Statement") will, at the date it is first mailed to Seller Parent's stockholders or at the time of the Stockholder Meeting or, if amended or supplemented, at the time of any amendment or supplement thereto, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. The Proxy Statement will comply as to form in all material respects with the requirements of the Exchange Act and the rules and regulations thereunder, except that no representation or warranty is made by Seller or Seller Parent with respect to statements made or incorporated by reference therein based solely on information supplied in writing by or on behalf of Buyer specifically for inclusion in any of the foregoing documents and is approved by Buyer for inclusion or incorporation by reference therein.

3.26 Environmental Matters.

- (a) Except as set forth on Section 3.26(a) of the Disclosure Schedules, each of Seller and Seller Parent (with respect to the Business) is and has been, for the past five years, in compliance in all material respects with all Environmental Laws and has no material liability under Environmental Laws or with respect to Hazardous Substances.
- (b) (i) Except as set forth on Section 3.26(b)(i) of the Disclosure Schedules, each of Seller and Seller Parent (with respect to the Business) has timely applied for, obtained and maintains all Permits required under Environmental Laws for the operation of the Business as currently operated; (ii) except as set forth on Section 3.26(b)(ii) of the Disclosure Schedules, Seller has not received any written notice proposing to revoke, suspend or materially modify the terms of any such Permits; and (iii) Section 3.26(b)(iii) of the Disclosure Schedules sets forth a true and complete list of all such Permits, all of which are valid and in full force and effect.
- (c) Except as set forth on Section 3.26(c) of the Disclosure Schedules, in the past five years, or earlier for matters that remain unresolved, neither Seller nor Seller Parent (with respect to the Business) has received or, to the Knowledge of Seller, been threatened with any request for information from a Governmental Authority, or any notice, order, summons, complaint, demand or letter from any Person, and there is no Action pending, or to the Knowledge of Seller, threatened, alleging, in either case, that Seller (with respect to the Business) is or has been in material violation of or subject to material Liability under Environmental Laws or with respect to Hazardous Substances.

- (d) Except as set forth on Section 3.26(d) of the Disclosure Schedules, there has been no Release of Hazardous Substances at, on, under or from any real property currently owned, operated, or leased by Seller or Seller Parent (with respect to the Business), and to Seller's Knowledge, there has been no Release of Hazardous Substances at, on, under or from any real property formerly owned, operated, or leased by Seller or Seller Parent (with respect to the Business), in each case that has resulted in or would reasonably be expected to result in material Liability for Seller or Seller Parent under any Environmental Law or with respect to Hazardous Substances.
- (e) Neither Seller nor Seller Parent (with respect to the Business) has arranged, by contract, agreement or otherwise, for the transportation, treatment or disposal of Hazardous Substances at any location except as would not reasonably be expected to result in material Liability pursuant to Environmental Laws.
- (f) Seller has made available all environmental assessments, reports, compliance audits prepared within the last five years, and all other material documents in Seller's or Seller Parent's possession or under their control related to the environmental condition of any real property currently or formerly owned, operated or leased by Seller or Seller Parent (with respect to the Business) or Seller's and Seller Parent's (with respect to the Business) compliance with Environmental Laws.

3.27 Employee Benefits.

- (a) Section 3.27(a) of the Disclosure Schedules sets forth a true and complete list, as of the date of this Agreement, of each material Parent Plan. With respect to each material Parent Plan, Seller or Seller Parent has made available to Buyer true and complete copies (to the extent applicable) of (i) the plan document or a written description thereof (or, if appropriate, a form thereof), including any amendments thereto, (ii) the most recent IRS determination or opinion letter received by Seller, (iii) the most recent summary plan description, and (iv) each insurance or group annuity contract or other funding vehicle.
- (b) Each Parent Plan complies in all material respects with applicable Laws and the terms thereof. Each Parent Plan intended to be "qualified" within the meaning of Section 401(a) of the Code has received a favorable determination letter from the IRS or is entitled to rely upon a favorable opinion issued by the IRS, and there are no existing circumstances or any events that have occurred that could reasonably be expected to cause the loss of any such qualification status of any such Parent Plan.
- (c) No Parent Plan is subject to Section 302 or Title IV of ERISA or Section 412 of the Code. No Parent Plan provides benefits or coverage in the nature of health, life or disability insurance or similar benefits following retirement or other termination of employment, other than coverage or benefits (i) required to be provided under Part 6 of Title I of ERISA or Section 4980(B)(f) of the Code, or any other applicable Law, or (ii) the full cost of which is borne by the employee or former employee (or any of their beneficiaries).
- (d) Except as set forth on Section 3.27(d) of the Disclosure Schedules, neither the execution of this Agreement nor the consummation of the transactions contemplated by the Transaction Documents will, either alone or in combination with another event, (i) accelerate the time of payment or vesting, or materially increase the amount of compensation due to any Business Employee under any Parent Plan, or (ii) result in any Business Employee who is a "disqualified individual" receiving any "excess parachute payment" (each such term as defined in Section 280G of the Code). No Parent Plan provides for the gross-up or reimbursement of Taxes for any Business Employee under Sections 4999 or 409A of the Code, or otherwise.

3.28 Intellectual Property.

- (a) Section 3.28(a) of the Disclosure Schedules sets forth a true and complete list of all registered or material Intellectual Property used in the conduct of the Business, separated by (i) Intellectual Property that is owned by Seller or Seller Parent or its subsidiaries, and (ii) Intellectual Property that is licensed to Seller or Seller Parent, excluding where the Intellectual Property is incorporated into personal property, such as laboratory or clinical equipment, kits, or assays, and used pursuant to a written license (collectively "Licenses").
- (b) Except as set forth in Section 3.28(b) of the Disclosure Schedules, Seller or Seller Parent owns or has valid licenses to use (which licenses are set forth in Section 3.28(a) of the Disclosure Schedules or do not need to be listed based on the exclusion set forth in Section 3.28(a) all material Intellectual Property used in

the conduct of the Business (the "Business Intellectual Property"), free and clear of all Liens other than Permitted Liens. To Seller's Knowledge, neither Seller's nor Seller Parent's conduct of the Business infringes on the Intellectual Property of any Person.

- (c) Except as set forth in Section 3.28(c) of the Disclosure Schedules or with respect to any Excluded Assets, neither Seller nor Seller Parent pays or receives any sales-related (either on revenue or a perunit basis) royalty to or from anyone with respect to any Business Intellectual Property, nor has Seller or Seller Parent licensed or sublicensed anyone to use any Business Intellectual Property for use in a business that may reasonably be considered competitive to the Business.
- (d) The Business Intellectual Property constitutes, and the Acquired Assets (along with the rights granted under the Intellectual Property Agreement and rights to Intellectual Property incorporated into personal property excluded from the scheduling obligations of Section 3.28(a) of the Disclosure Schedules pursuant to Section 3.28(b)) include, all of the Intellectual Property necessary or material to conduct the Business. To Seller's Knowledge, none of the Business Intellectual Property is owned by any other Person without a valid and enforceable right of Seller or Seller Parent to use and possess such Business Intellectual Property. None of the Business Intellectual Property is subject to any Lien, other than Permitted Liens.
- (e) Except as set forth in Section 3.28(e) of the Disclosure Schedules or in the Intellectual Property Agreement, (i) all rights of the Business in and to the material Business Intellectual Property will be unaffected by the transactions contemplated by the Transaction Documents, and (ii) through the Acquired Assets and the Intellectual Property Agreement, Buyer will have rights immediately after the Closing that are in all material respects similar to the rights exercised by the Business prior to Closing.
- (f) Except as set forth in Section 3.28(f) of the Disclosure Schedules, neither Seller nor Seller Parent has given nor received any written notice of any pending conflict with, or infringement of the rights of others with respect to any Business Intellectual Property, and to the Knowledge of Seller, no Person has infringed, misappropriated or otherwise violated, and no Person is currently infringing, misappropriating or otherwise violating, any Business Intellectual Property.
- (g) All trade secrets, confidential information or know-how owned by or purported to be owned by Seller or Seller Parent and exclusively used in the Business have been maintained in confidence in accordance with protection procedures substantially similar to those customarily used by comparable companies in the same industry as Seller and Seller Parent to protect rights of like importance. All of Seller's or Seller Parent's employees or consultants who have contributed to or participated in the conception or development of any material Business Intellectual Property or material Intellectual Property exclusively used in the Business that are purported to be owned by Seller or Seller Parent have executed and delivered to Seller or Seller Parent an agreement assigning all proprietary rights to Seller or Seller Parent and restricting such Person's rights to use or disclose such proprietary information.
- (h) Seller has provided to Buyer copies of all standard form agreements used by Seller and Seller Parent in the last three years that contain assignments or licenses of Intellectual Property or otherwise relate to Business Intellectual Property.
- (i) <u>Section 3.28(i)</u> of the <u>Disclosure Schedules</u> lists all open source computer code contained or used in any product or service of the Business and describes (i) the applicable software name and version number, (ii) the licensor, (iii) the license under which such code was obtained, (iv) whether such code was modified by or for the Business and (v) whether such code was distributed by or for the Business. No product or service of the Business is subject to any Contract that would require Seller to divulge to any Person any source code owned or purported to be owned by Seller or Seller Parent that is part of any product or service of the Business.
- 3.29 Accounts Receivable; Accounts Payable All accounts and notes receivable reflected on the Financial Statements and all accounts and notes receivable arising subsequent to the Financial Statements and prior to the Closing Date, have arisen or will arise in the Ordinary Course of Business out of bona fide sales and deliveries of goods, performance of services or other business transactions, and, to the Knowledge of Seller, represent or will represent legal, valid, binding and enforceable obligations to Seller or Seller Parent, as applicable. Except as set forth in Section 3.29 of the Disclosure Schedules, all accounts payable related to (a) the rent, utilities and other services of the Business provided at the Assigned Locations, (b) Business motor vehicle licenses and registrations, and (c) all obligations related to the Business's IT Assets (including hardware and software) are current within 30 days of the due date.

- 3.30 Affiliate Transactions. Except as set forth in Section 3.30(a) of the Disclosure Schedules, no manager, member, officer, director, or Affiliate of Seller or Seller Parent or, to the Knowledge of Seller, any immediate family member (as defined in Item 404 of the Exchange Act (17 CFR § 229.4040)) of any such individual or any entity in which any such Person or individual owns any controlling beneficial interest, is a party or has been a party to any Contract, commitment or transaction with Seller or Seller Parent involving the Business, the Acquired Assets or the Assumed Liabilities, or has any interest in any Assumed Liability or Acquired Asset or in any other property used by Seller or Seller Parent in the Business. Section 3.30(b) of the Disclosure Schedules lists all intercompany transactions between or among Seller, Seller Parent and their Affiliates involving the Business (including any transactions involving the Business, on one hand, and the Excluded Business, on the other hand).
- 3.31 No Other Representations or Warranties Except for the representations and warranties made by Seller and Seller Parent in this Article III, neither Seller nor Seller Parent makes any other express or implied representation or warranty with respect to Seller, Seller Parent or their respective Affiliates, businesses (including the Business), operations, properties, assets, liabilities, condition (financial or otherwise) or prospects, or any estimates, projections, forecasts and other forward-looking information or business and strategic plan information, notwithstanding the delivery or disclosure to Buyer or any of its Affiliates or Representatives of any documentation, forecasts or other information (in any form or through any medium) with respect to any one or more of the foregoing, and such other representations and warranties are hereby disclaimed. Except for the representations and warranties expressly set forth in Article IV, Seller and Seller Parent each hereby acknowledges that Buyer has not made and is not making any other express or implied representation or warranty with respect to Buyer, its Affiliates or their respective businesses or operations. Notwithstanding the foregoing, nothing in this Agreement limits any claim, right or remedy that a Person may have against another Person for Fraud.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to Seller:

- **4.1** Organization, Standing and Power. Buyer is a corporation duly organized, validly existing, and in good standing under the Laws of the State of Delaware, has the requisite power and authority to own, lease and operate its properties and assets and to carry on its business as now being conducted.
- 4.2 Authority. Buyer has all requisite corporate power and authority to execute and deliver the Transaction Documents to which it is, or will be, a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery by Buyer of this Agreement and the other Transaction Documents to which Buyer is, or will be, a party, the performance by Buyer of its obligations hereunder and thereunder and the consummation by Buyer of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action, and no other action on the part of Buyer are necessary to authorize this Agreement or the other Transaction Documents or to consummate the transactions contemplated hereby or thereby. This Agreement has been duly executed and delivered by Buyer and, assuming the due authorization, execution and delivery thereof by Seller and Seller Parent, constitutes the legal, valid and binding obligation of Buyer, enforceable in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium, and similar Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law). When each other Transaction Document to which Buyer is a party is entered into, such Transaction Document will be duly executed and delivered by Buyer. When each other Transaction Document to which Buyer is or will be a party has been duly executed and delivered by Buyer (assuming due authorization, execution and delivery by each other party thereto), each such Transaction Document will constitute a legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally or by general equity principles (regardless of whether enforcement is sought in a proceeding in equity or at law).

4.3 No Conflict; Required Filings and Consents.

(a) The execution and delivery of this Agreement and the other Transaction Documents by Buyer does not, and the performance of this Agreement and the Transaction Documents by Buyer will not (i) conflict with or violate the certificate of incorporation or bylaws of Buyer, (ii) conflict with or violate any Laws in effect as of the Effective Date and applicable to Buyer or by which any of its properties are bound, or (iii) result in any breach of or

constitute a default (or an event that with notice or lapse of time or both would become a default) under, or required payment under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Lien on, any of the properties or assets of Buyer, in each case of clauses (i) through (iii) above, with such exceptions as would not reasonably be expected to prevent Buyer from consummating the transactions contemplated by this Agreement or the other Transaction Documents.

- (b) The execution and delivery of this Agreement and the other Transaction Documents by Buyer does not, and the performance of this Agreement and the other Transaction Documents by Buyer and the consummation of the transactions contemplated hereby and thereby will not require Buyer to obtain any Consent, Permit or order of, or to make any filing with or notification to, or procure other action by, any Person based on any Laws, except as required under the HSR Act and for any such consent, permit or order, or filing or notification which failure to obtain or make would not reasonably be expected to prevent Buyer from consummating the transactions contemplated by this Agreement.
- **4.4** <u>Brokers or Finders</u>. No agent, broker, investment banker, financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or any other similar commission or fee in connection with any of the transactions contemplated by this Agreement, based upon arrangements made by or on behalf of Buyer, for which Seller or any of its Affiliates could have any liability or responsibility whatsoever.
- **4.5** <u>Litigation</u>. There is no Action or Governmental Order pending, or to the knowledge of Buyer, threatened, against or by Buyer that would reasonably be expected to affect Buyer's ability to consummate the transactions contemplated by this Agreement or perform its obligations under the Transaction Documents to which it is a party.
- 4.6 Information Supplied. None of the information supplied or to be supplied by Buyer for inclusion or incorporation by reference in the Proxy Statement will, at the date it is first mailed and at the time of the Stockholder Meeting, and if either is amended or supplemented, at the time of any amendment or supplement thereto, to Seller Parent's stockholders or at the time of the Stockholder Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- 4.7 No Other Representations or Warranties. Except as expressly set forth in this Article IV or as may be set forth in any other Transaction Document, Buyer makes no representation or warranty, express or implied, at law or in equity, with respect to Buyer, its Affiliates, its businesses or financial condition or any of its assets, liabilities or operations or any other matter, and any such other representations or warranties are hereby disclaimed. Buyer acknowledges that (a) it has received access to certain books and records, facilities, equipment, Contracts and other assets of Seller and Seller Parent related to the Business, and (b) that Buyer and its Affiliates and representatives have had an opportunity to meet with the management of Seller and Seller Parent to discuss the Business. Buyer has conducted its own independent investigation, review, analysis and evaluation of the Business to assist Buyer in making an informed decision with respect to the purchase of the Acquired Assets and the Business, the assumption of the Assumed Liabilities and the execution of the Transaction Documents and, with respect to the transactions contemplated by this Agreement, has relied on the results of such investigation and the representations and warranties of Seller and Seller Parent in Article III and as set forth in any other Transaction Document. Except for the representations and warranties expressly set forth in Article III and as set forth in any other Transaction Document, Buyer hereby acknowledges that neither Seller nor Seller Parent has made or is making any other express or implied representation or warranty with respect to Seller, Seller Parent or their respective Affiliates, business (including the Business) or operations. Notwithstanding the foregoing, nothing in this Section 4.7 or elsewhere in this Agreement limits any claim, right or remedy that a Person may have against another Person for Fraud. Buyer specifically disclaims that it is relying upon or has relied upon any other representations or warranties that may have been made by Seller, Seller Parent or any other Person, and acknowledges and agrees that Seller and Seller Parent has specifically disclaimed and does hereby specifically disclaim any such other representation or warranty.

ARTICLE V CONDUCT OF BUSINESS PENDING THE CLOSING

5.1 <u>Interim Operating Covenants</u>. During the period beginning on the Effective Date and ending on the earlier of the Closing Date, or the termination of this Agreement (such period, the "*Interim Period*"), except (a) as provided for in this Agreement, (b) as required by applicable Law or GAAP, or (c) to the extent that Buyer shall

otherwise consent in writing (which consent shall not be unreasonably withheld, conditioned or delayed), each of Seller and Seller Parent shall (x) conduct the Business in the Ordinary Course of Business (which shall include conducting the Business in accordance with any Material Contract to which Seller is a party as of the Effective Date, and which has been made available to Buyer) and (y) use commercially reasonable efforts to maintain and preserve intact the current organization, business and franchise of the Business and to preserve the rights, franchises, goodwill and relationships of the employees, customers, suppliers, regulators and others having business relationships with Seller and Seller Parent with respect to the Business. Nothing contained in this Section 5.1 shall impose any restriction on, or impose obligations of Seller or Seller Parent with respect to, the Excluded Business (including, for the avoidance of doubt, selling or continuing to operate any segment of the Excluded Business, subject to the other terms and conditions of this Agreement and the Amended STAT Services Agreement and the Transition Services Agreement (if any)) or the Excluded Assets (but only to the extent such Excluded Assets do not relate to the Business or the performance of services under, and solely for the duration of, the Amended STAT Services Agreement or, if later, the duration of the Transition Services Agreement (if any)). In addition, and subject to the foregoing, solely with respect to the Business (including the Excluded Assets that relate to the Business or the performance of services under the Amended STAT Services Agreement) during the Interim Period, without the prior written consent of Buyer (which consent shall not be unreasonably withheld, conditioned or delayed), Seller and Seller Parent shall:

- (i) not sell, lease, transfer, license, pledge, dispose of, assign, or otherwise encumber or subject to any Lien (other than any Permitted Lien) any of the Acquired Assets or any rights thereto, other than inventory in the Ordinary Course of Business and Liens (other than Permitted Liens) with respect to Debt that will be paid off at the Closing pursuant to Section 2.8(a)(iv), provided that Seller Parent shall promptly notify Buyer of any Debt that is incurred after the Effective Date and that is secured by Liens on the Acquired Assets;
- (ii) comply with all Laws applicable to the Business, the Acquired Assets and the Assumed Liabilities;
- (iii) keep in full force and effect the Insurance Policies (or replacement insurance materially comparable in amount and scope to insurance currently carried by Seller and Seller Parent with respect to the Business);
- (iv) preserve and maintain in effect all material Permits necessary to carry on the Business as currently conducted or for the ownership and use of the Acquired Assets and Assigned Locations;
- (v) not cancel, compromise, waive, or release any material right or claim, other than in the Ordinary Course of Business;
- (vi) not grant any license or sublicense of any rights under or with respect to any Business Intellectual Property other than in the Ordinary Course of Business;(vii) not make or authorize any change in any of Seller Parent's charter or bylaws to the extent such amendment or change would prevent, impede or delay the Stockholder Approval, the consummation of the transactions contemplated hereby or otherwise adversely affect the Acquired Assets or the Assumed Liabilities;
- (viii) not issue, sell, or otherwise dispose of any of the equity securities of Seller, or grant any options, warrants, or other rights to purchase or obtain (including upon conversion, exchange, or exercise) any of Seller's equity securities;
- (ix) not make any increase in the base compensation or target bonus of any of the Business Employees, except in the Ordinary Course of Business or as may be required by any Law or Contract;
- (x) other than with respect to the Lease Extensions, not amend, change, supplement, waive or terminate any Material Contract or any other Contract that is an Assigned Contract, except for renewals in the Ordinary Course of Business;
- (xi) not to enter into, amend, change, supplement, waive or terminate any Material Contract or any other Contract that is an Assigned Contract (A) that would reasonably be expected to have an adverse effect on the Business or any of the Acquired Assets or the Assumed Liabilities, or (B) pursuant

to which any Person (other than Buyer and its Affiliates) is first granted, or that otherwise first subjects Buyer or any of its Affiliates to, (1) any covenants or provisions restricting competition or prohibiting any of them from freely engaging in the Business or otherwise restricting the conduct of the Business or use of the Acquired Assets in any market, geographic area or other jurisdiction, (2) any "most favored nation" or "best pricing" terms or any type of exclusivity, special discount, right of first refusal, first notice or first negotiation with respect to any of the Acquired Assets, or (3) any license, covenant not to sue, immunity or other similar right with respect to or under any of the Acquired Assets;

- (xii) enforce any and all non-solicitation agreements or other agreements containing restrictive covenants with respect to each Business Employee; *provided* that neither Seller Parent nor Seller shall enforce any such provisions against Buyer or any Business Employee with respect to the transactions expressly contemplated by this Agreement;
- (xiii) not make any material change in its methods of accounting principles or practices used in the Business (including with respect to reserves);
- (xiv) not commence, negotiate, settle, pay, discharge or satisfy any Action (A) primarily relating to the Business or any of the Acquired Assets or the Assumed Liabilities or (B) which may impact the Business or any of the Acquired Assets or the Assumed Liabilities;
- (xv) not acquire by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization, in each case if such transaction has an adverse effect on the Business or any of the Acquired Assets or the Assumed Liabilities;(xvi) not terminate or close any facility or operation used in the Business, including any of the Assigned Locations;
- (xvii) settle or compromise any Action in respect of material Taxes; enter into any Contract in respect of material Taxes with any Governmental Authority; or amend any Tax Return, in each case, with respect to the Acquired Assets or the Business and to the extent such action would reasonably be expected to result in any increase in the Liability for Taxes of Buyer or its Affiliates;
- (xviii) not take or omit to take any action which has, individually or in the aggregate, a Material Adverse Effect;
 - (xix) not acquire any real property; and
 - (xx) not agree, whether in writing or otherwise, to do any of the foregoing.

5.2 No Negotiation.

(a) Except as permitted by this Section 5.2, during the Interim Period, each of Seller and Seller Parent shall not, and each of Seller and Seller Parent shall instruct and cause its directors, officers, employees and other Representatives not to (i), directly or indirectly, solicit, initiate, knowingly encourage or entertain any inquiries or proposals from, discuss or negotiate with, provide any Confidential Information to, any other Person (other than Buyer) regarding any merger, consolidation, business combination or similar transaction involving Seller or Seller Parent that would include all or any part of the Business or the sale, directly or indirectly, of any of the Acquired Assets or any portion of the Business (any of the above, an "Alternative Transaction") or any proposal or offer that would reasonably be expected to lead to an Alternative Transaction, (ii) engage in, participate in or otherwise continue any discussions or negotiations regarding, or furnish to any Person any nonpublic information with respect to, or take any other action to knowingly facilitate the making of, any proposal or offer that constitutes, or would reasonably be expected to lead to, any Alternative Transaction, or (iii) enter into or agree to enter into any letter of intent, memorandum of understanding, agreement in principle, merger agreement, acquisition agreement or other similar agreement relating to any Alternative Transaction, other than any Acceptable Confidentiality Agreement. Seller Parent and Seller shall, and Seller Parent and Seller shall cause its and their respective Representatives to, immediately cease and cause to be terminated any discussions and negotiations with any person conducted heretofore with respect to any Alternative Transaction, or proposal or offer that would reasonably be expected to lead to any Alternative Transaction. Seller Parent will promptly inform the individuals and entities referred to in the preceding sentence of the obligations undertaken in this Section 5.2. Seller Parent will promptly request from each Person that has executed a confidentiality agreement in connection with its consideration of making an Alternative

Transaction within the past two years to return or destroy (as provided in the terms of such confidentiality agreement) all confidential information concerning the Business and promptly terminate all physical and electronic data access previously granted to such person. Seller Parent shall promptly (but in no event later than 24 hours after any directors or executive officers of Seller Parent become aware of the occurrence of any of the events set forth in clauses (A)-(C) below and, in any event, within 72 hours) notify Buyer if (A) any Alternative Transaction or any indication of interest, proposal or offer with respect to an Alternative Transaction is received by, (B) any non-public information is requested in connection with any Alternative Transaction from, or (C) any discussions or negotiation with respect to an Alternative Transaction are sought to be initiated or continued with, it, or any of its Representatives, indicating, in connection with such notice, the name of such person and the material terms and conditions of any proposals or offers (including, if applicable, copies of any written requests, proposals or offers, including proposed agreements) and thereafter shall keep Buyer reasonably informed, on a reasonably current basis, of the status and material terms of any such proposals or offers (including any amendments thereto) and the status of any such discussions or negotiations.

(b) Notwithstanding anything to the contrary in Section 5.2(a), at any time prior to receipt of the Stockholder Approval, Seller or Seller Parent may furnish information to, and negotiate or otherwise engage in discussions with, any Person who delivers a bona fide written inquiry, proposal or offer from any Person or group (other than Buyer or its Affiliates) relating to, in a single transaction or series of related transactions, an Alternative Transaction, that was not solicited after the date of this Agreement, if and so long as (i) the board of directors of Seller Parent determines in good faith after consultation with its outside legal counsel that (x) the failure to take such action would reasonably be expected to breach the directors' fiduciary duties under applicable Law, (y) such a proposal is, or would reasonably be expected to lead to, an Alternative Transaction that is more favorable from a financial point of view to Seller Parent's stockholders than the transactions contemplated by this Agreement and (z) is reasonably likely to be consummated on the terms proposed, taking into account all legal, regulatory, financial, timing, financing and other aspects of such proposal and of this Agreement (a "Superior Proposal"), (ii) there has been no material breach of Section 5.2(a) in connection with such Alternative Transaction, (iii) Seller Parent has provided at least 24 hours prior written notice to Buyer of its intention to furnish information to, and negotiate or otherwise engage in discussions with, any Person and (iv) Seller Parent has provided Buyer with a copy of written inquiry, proposal or offer; provided, however, that Seller Parent (x) will not take any such action without first entering into an Acceptable Confidentiality Agreement with such Person notwithstanding Section 5.2(a), and (y) will concurrently (and in any event within 24 hours) provide to Buyer any non-public information concerning the Business provided to such other Person which was not previously provided to Buyer.

(c) Seller Parent agrees that it will keep Buyer reasonably informed, on a prompt basis, of the status and material terms of any such proposals or offers and the current basis of the status and details (including any material developments) in respect of any such discussions or negotiations and that it will deliver to Buyer a summary of any material changes to any such proposals or offers and all nonpublic information being furnished to such Person.

(d) Except as set forth in this Section 5.2(d), neither the board of directors of Seller Parent nor any committee thereof shall (i) withdraw, modify or qualify, or publicly propose to withdraw, modify or qualify, the Board Recommendation, (ii) recommend the approval or adoption of, or approve or adopt, declare advisable, or publicly propose to recommend, approve, adopt or declare advisable, any Alternative Transaction, (iii) publicly take a neutral position or no position with respect to an Alternative Transaction at any time beyond ten Business Days after receipt of such Alternative Transaction by Seller Parent or any public announcement by the party that made the Alternative Transaction proposal, (iv) fail to publicly reaffirm the Board Recommendation within five Business Days of a reasonable written request by Buyer to make such public reaffirmation or (v) make any public statement in connection with the Stockholders Meeting that is inconsistent with the Board Recommendation (any action described in this clause being referred to as an "Adverse Recommendation Change"). Notwithstanding anything in this Agreement to the contrary, at any time prior to the receipt of the Stockholder Approval (and in no event after the receipt of the Stockholder Approval), the board of directors of Seller Parent may effect an Adverse Recommendation Change and, subject to compliance with this Section 5.2 and Section 8.4, terminate this Agreement if and only if the board of directors of Seller Parent receives an unsolicited Alternative Transaction that the board has determined to be a Superior Proposal and further determines in good faith after consultation with its outside legal counsel that such action is reasonably likely to be required in order for the directors to comply with their fiduciary duties under applicable Law. Prior to effecting such Adverse Recommendation Change, Seller Parent shall provide Buyer with five (5) Business Days' prior written notice advising Buyer of its intention to effect an Adverse Recommendation

Change, and specifying in reasonable detail the material terms and conditions of, and the identity of any Person proposing, such Superior Proposal, and Seller Parent shall, during such time and if requested by Buyer, engage in good faith negotiations with Buyer to amend this Agreement (x) such that such Superior Proposal would no longer constitute a Superior Proposal or (y) in a manner that obviates the need to effect an Adverse Recommendation Change, as applicable. Notwithstanding any Adverse Recommendation Change, if this Agreement is not otherwise terminated by any Party in accordance with the terms hereof, this Agreement shall be submitted to Seller Parent stockholders at the Stockholders Meeting for the purpose of voting on adopting this Agreement, and nothing contained herein shall be deemed to relieve Seller Parent of such obligation. From the date of this Agreement until the Closing Date, Seller Parent shall not terminate, amend, modify or waive any provision of any "standstill" or similar agreement to which Seller Parent is a party and shall use its reasonable best efforts to enforce, to the fullest extent permitted under applicable Law, the provisions of any such agreement, including by seeking injunctions to prevent any breaches of such agreements and to enforce specifically the terms and provisions thereof, except, in each case, to the extent the board of directors of Seller Parent determines in good faith, after consultation with outside counsel, that taking such action would result in a violation of the fiduciary duties of the board of directors of Seller Parent under applicable Law.

(e) Nothing contained in this Section 5.2 or elsewhere in this Agreement shall prohibit Seller Parent from (i) taking and disclosing to its stockholders a position contemplated by Rule 14d-9, Rule 14e-2(a) or Item 1012(a) of Regulation M-A promulgated under the Exchange Act or (ii) making any disclosure to its stockholders if the board of directors of Seller Parent determines in good faith (after consultation with and receiving advice of its outside counsel) that failure to do so would result in a violation of its fiduciary duties under applicable Law; provided, however, that any disclosure made pursuant to clause (i) or (ii) of this Section 5.2(e) (other than any disclosure made in connection with an Adverse Recommendation Change permitted by Section 5.2(d)) shall be deemed to constitute an Adverse Recommendation Change unless the board of directors of Seller Parent shall expressly publicly reaffirm the Board Recommendations in such disclosure. For avoidance of doubt, it shall not constitute an Adverse Recommendation Change for the board of directors of Seller Parent to make a "stop, look and listen" communication pursuant to Rule 14d-9(f) under the Exchange Act.

5.3 Governmental Approvals and Consents.

(a) Seller Parent and Buyer shall, to the extent required by any applicable Law and, if so required, in no event later than ten Business Days following the Effective Date, file or cause their respective ultimate parents to file, with the Federal Trade Commission ("FTC") and the Department of Justice ('DOJ') the notification and report form required for the transactions contemplated by this Agreement and any supplemental information requested in connection therewith pursuant to the HSR Act. Seller Parent and Buyer shall furnish to the other such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission which is necessary under the HSR Act. Seller Parent and Buyer shall keep each other apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC and the DOJ and shall comply promptly with any such inquiry or request. Seller Parent and Buyer shall use their commercially reasonable efforts to obtain any clearance required under the HSR Act for the transactions contemplated by this Agreement.

(b) The parties shall each cooperate reasonably with one another in connection with resolving any inquiry or investigation by the Governmental Authority relating to the transactions contemplated hereby. Without limiting the foregoing, each party shall (i) promptly inform the other party of any written or oral communication received from the Governmental Authority relating to the transactions contemplated hereby (and if in writing, furnish the other party with a copy of such communication); (ii) respond as promptly as practicable to any request from the Governmental Authority for information, documents or other materials in connection with the transactions contemplated hereby; (iii) provide to the other party, and permit the other party to review and comment in advance of submission, considering in good faith the views of the other party, and incorporate the other party's reasonable comments, all proposed correspondence, filings, and written communications to the Governmental Authority with respect to the transactions contemplated hereby, provided that such material (A) may be redacted as necessary (1) to comply with contractual arrangements, (2) to address legal privilege concerns or (3) to remove references concerning the valuation of the parties or (B) designated as "outside counsel only," which materials and the information contained therein shall be given only to outside counsel and previously-agreed outside economic consultants of the recipient and will not be disclosed by such outside counsel or outside economic consultants to employees, officers, or directors of the recipient without the advance written consent of the party providing such

materials; and (iv) not participate in any substantive meeting, communication, or discussion with the Governmental Authority in respect of investigation or inquiry concerning the transactions contemplated hereby unless it consults with the other party in advance and, except as prohibited by applicable Law or the FTC or DOJ, gives the other party the opportunity to attend and participate therein. Neither Buyer nor Seller Parent shall stay, toll, or extend any applicable waiting period under the HSR Act without the advance written agreement of other Party, and Buyer shall not pull or refile any filing made under the HSR Act without advance notice to Seller Parent. Each of Buyer and Seller shall use commercially reasonable efforts to certify compliance with an additional information or documentary material issued by a Governmental Entity under 15 U.S.C. Sect. 18a(e) and in conjunction with the transactions contemplated by this Agreement (a "Second Request"), pursuant to the HSR Act as soon as practicable after receipt of such Second Request.

- (c) Notwithstanding anything to the contrary contained in this Agreement, nothing in this Agreement will require or obligate Buyer or any of its Affiliates to (and in no event shall any representation, warranty or covenant of Buyer contained in this Agreement be breached or deemed breached as a result of the failure of Buyer to take any of the following actions): (i) agree to or otherwise become subject to any limitations on (A) the right of Buyer effectively to control or operate its business (including the Business after the Closing) or assets (including the Acquired Assets) or (B) the right of Buyer to exercise its full rights of ownership of its business (including the Business after the Closing) or assets (including the Acquired Assets); (ii) agree or be required to sell or otherwise dispose of, hold separate (through the establishment of a trust or otherwise), or divest itself of all or any portion of the business, assets or operations of Buyer or any of its Affiliates; or (iii) otherwise take any steps to avoid or eliminate any impediment that may be asserted under any Competition Law. In regard to any Governmental Authority, neither Seller Parent nor Seller shall, without Buyer's prior written consent in Buyer's sole discretion, discuss or commit to any divestiture transaction, or discuss or commit to alter any of their businesses or commercial practices in any way, or otherwise discuss, take or commit to take any action that limits Buyer's freedom of action with respect to, or Buyer's ability to retain any of the businesses, product or service lines or assets of, the Business or otherwise limits Buyer's ability to receive the full benefits of this Agreement.
- (d) Buyer shall lead all communications, strategy and efforts to obtain all necessary clearances, actions or non-actions and consents from Governmental Authorities under Competition Laws in connection with the transactions contemplated hereby and any litigation matters relating to the Competition Laws, subject to good faith consultations with Seller Parent and the inclusion of Seller Parent at meetings with Governmental Authorities with respect to any discussion related to the transactions contemplated hereby under the Competition Laws.
- (e) Seller Parent and Buyer shall each pay 50% of any filing fees payable by under the HSR Act.
- (f) Each Party hereto shall, as promptly as possible, use its commercially reasonable efforts to obtain, or cause to be obtained, all other Consents and orders from all Governmental Authorities and make, or cause to be made, all filings required by Law to be made by it to consummate the transactions contemplated hereby that may be or become necessary for its execution and delivery of this Agreement and the performance of its obligations pursuant to this Agreement and the other Transaction Documents. Each Party shall cooperate with each other Party and its Affiliates in promptly seeking to obtain all such other Consents and orders that the other is required by Law to make. The Parties hereto shall not willfully take any action that will have the effect of delaying, impairing or impeding the receipt of any required Consents and orders that the other is required by Law to make.

5.4 Information Technology; Assistance with Customer Transition.

(a) At the Effective Date and monthly thereafter until the Closing Date, Seller shall deliver to Buyer, consistent with applicable Law, the Client Setup Data in the form of Annex 3.7(a)(i) with all fields accurately updated and completed for all customers of the Business as of each such date (including to reflect any new customer accounts gained since the previous date of delivery). Seller shall, until the Closing Date, update on a monthly basis the information provided in Section 3.7(a)(iv) of the Disclosure Schedules with all fields accurately updated and completed as of each such date (including to reflect any new customer accounts gained since the previous date of delivery). During the Interim Period, consistent with applicable Law, Seller and Seller Parent shall reasonably cooperate with Buyer, provide Buyer reasonable access to, and shall make its employees reasonably available for, general informational and client pre-planning meetings as well as arranging for introductory visits with any and all Customers that Buyer reasonably requests as soon as reasonably practicable after the Effective Date, and Seller and Seller Parent shall promptly provide information and access, consistent with applicable Law, reasonably requested by Buyer for the express and limited purpose of facilitating Buyer's integration activities with respect to the Business

from and after the Effective Date, including facilitating the technical development of customer interfaces for Buyer's LIS (as defined below) as promptly as practicable on and after the Effective Date, deactivating existing lines as may be mutually agreed upon and when reasonably necessary, facilitation of Buyer dropping off supplies for Buyer's use from and after the Closing, and performing such other activities as are reasonably necessary in order to transition the Business from Seller to Buyer effective from and after the Closing. Buyer shall use commercially reasonable efforts to seek any such reasonable access contemplated in this Section 5.4 in a manner that does not unreasonably interfere with the day-to-day operation of the business of Seller, Seller Parent or their Affiliates.

- (b) Without limiting the foregoing, promptly after the Effective Date, subject to Competition Laws and the terms of any "clean team" agreement among the Parties governing the sharing of competitively sensitive information, Seller Parent and Seller shall reasonably cooperate as reasonably requested by Buyer in connection with the building, configuration, completion and testing of the customer interfaces (as may be mutually agreed upon) with Buyer's LIS that will allow the ordering of tests and reporting of test results (the "Interface Configuration") after the Closing. Buyer shall use commercially reasonable efforts to complete such Interface Configuration in a manner as to not unreasonably interfere with the day-to-day operations of Seller, Seller Parent or their Affiliates. The Parties acknowledge that the process of building, configuring, completing and testing these interfaces is an iterative, collaborative process and agree to reasonably cooperate in completing the interfaces as promptly as reasonably practicable. If Buyer believes that Seller is not timely performing its obligations under this Section 5.4(b). Buyer shall promptly notify Seller of the same in a reasonably detailed written notice.
- (c) The Parties hereby agree that Kurt Wanner and Nicole Carlson, on behalf of Buyer, and Richard Paladino, on behalf of Seller (each such individual, an "IT Designee") will serve as project leads and primary contact persons with respect to completion of the Interface Configuration (the "LIS Project") and all other integration matters related to the IT Assets. The IT Designees will meet (in person, via teleconference or video conference) no less than weekly (unless otherwise reasonably requested by Buyer) to discuss the LIS Project and reasonably cooperate and provide status updates to Buyer with respect to the completion of the LIS Project or any amendments or modifications to the Interface Configuration on a weekly basis until such LIS Project is complete, or upon such other timeframes as mutually agreed to by the Parties. Seller's IT Designee shall serve as a point of contact and coordination for all IT Asset-related questions and issues relating to Buyer's integration activities with respect to the Business from and after the Effective Date, including with respect to any related activities as are reasonably necessary in order to transition the Business from Seller to Buyer effective from and after the Closing. The IT Designee shall have the authority to, and Seller shall cause the IT Designee to, provide (i) Buyer Group with access to the Assigned Locations, Customers, vendors and information that is reasonably requested by Buyer in connection with the transition of the Business from Seller to Buyer effective from and after the Closing and (ii) vendors of Buyer Group with access to the Assigned Locations within five Business Days following Buyer's request therefor, for the express and limited purpose of completion of the LIS Project, provided that this shall be subject to applicable Law.
- (d) During the Interim Period, Seller shall promptly conduct reasonable knowledge transfer and training sessions for Buyer's employees with respect to Seller's compendium and Customer connectivity solutions known to and used by Seller in connection with the Business.
- (e) During the Interim Period, Seller shall provide Buyer reasonable access to, and shall make its employees reasonably available for, general informational meetings, and Seller shall provide information reasonably requested to facilitate Buyer's integration activities including facilitation of Buyer dropping off supplies, and performing such other related activities as are reasonably necessary in order to transition the Acquired Assets and Assumed Liabilities from Seller to Buyer, effective from and after the Closing. Buyer shall use commercially reasonable efforts to complete such integration activities in a manner as to not unreasonably interfere with the day-to-day operations of Seller, Seller Parent or their Affiliates.
- (f) During the Interim Period, Seller shall use commercially reasonable efforts to clear title to, and obtain any documents and other records required to transfer title to, all motor vehicles included in the Acquired Assets such that all such motor vehicles can be transferred to Buyer, free and clear of all Liens other than Permitted Liens, on the Closing Date.
- 5.5 <u>Retention of Records and Specimens</u> Seller shall engage a third-party to ensure that all records, films, slides, blocks, tissue and other specimens required to be maintained by Seller with respect to the Business are retained and available in the manner and for the timeframe required by applicable Laws. Seller shall inform Buyer as

to where such records and specimens are maintained prior to the Closing Date (and if any change to such location or arrangement occurs following the Closing Date) and shall ensure that such third-party provides access to Buyer to such records and specimens within one week of a valid request or as otherwise required by applicable Laws.

5.6 Preparation of Proxy Statement; Stockholder Meeting.

- (a) Seller Parent shall, in consultation with Buyer, prepare and file with the SEC as promptly as practicable, and in any event within 20 Business Days after the Effective Date, a preliminary version of the Proxy Statement to be sent to Seller Parent's stockholders in connection with the Stockholder Meeting. Seller Parent shall not file the Proxy Statement (or any preliminary version thereof) with the SEC without first providing Buyer and its counsel a reasonable opportunity to review and comment thereon, and Seller Parent shall give reasonable and good faith consideration to all additions, deletions or changes suggested by Buyer or its counsel; provided that no disclosure relating specifically to non-public information about Buyer or its Affiliates may be made without Buyer's prior approval. Seller Parent shall include in the Proxy Statement the Board Recommendation. If, prior to the expiration of the ten (10) day waiting period provided in Rule 14a-6 under the Exchange Act, Seller Parent does not receive either (i) comments from the SEC on the preliminary Proxy Statement or (ii) notice from the SEC that it will review the preliminary Proxy Statement, then Seller Parent shall file definitive proxy materials (including the definitive Proxy Statement) with the SEC and cause the definitive Proxy Statement to be mailed to Seller Parent's stockholders as soon as reasonably practicable, and in any event not later than two (2) Business Days after the expiration of such waiting period. Seller Parent shall use commercially reasonable efforts to resolve all SEC comments, if any, with respect to the Proxy Statement as promptly as practicable after receipt thereof. Promptly following confirmation by the SEC that the SEC has no further comments, Seller Parent shall cause the definitive Proxy Statement to be filed with the SEC and mailed to Seller Parent's stockholders. Seller Parent shall be responsible for 100% of the fees, costs and expenses associated with the preparation, filing and mailing of the Proxy Statement.
- (b) Each of Seller, Seller Parent and Buyer shall promptly furnish all information concerning such Person and its Affiliates to the other, and provide such other assistance, as may be reasonably requested by such other Party to be included therein and will otherwise reasonably assist and cooperate with the other in the preparation, filing and distribution of the Proxy Statement and the resolution of any comments to either received from the SEC.
- (c) Seller Parent shall (i) notify Buyer promptly of the receipt of any comments, whether written or oral, from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements to the Proxy Statement or for additional information and (ii) supply Buyer with copies of all correspondence between it or any of its Representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to such filings. Prior to responding to any comments from the SEC or its staff, Seller Parent shall consult with Buyer and provide Buyer and its counsel a reasonable opportunity to review and comment on any written response thereto, and Seller Parent shall give reasonable and good faith consideration to all additions, deletions or changes suggested by Buyer or its counsel.
- (d) Except as provided in <u>Section 5.6(c)</u>, no amendment or supplement to the Proxy Statement shall be made by Seller or Seller Parent without the prior approval of Buyer (which shall not be unreasonably withheld, conditioned or delayed).
- (e) Seller Parent will take all action necessary in accordance with applicable Law, its organizational documents and the rules of the NYSE to establish a record date for, duly call, give notice of, convene and hold a meeting of its stockholders (the "Stockholder Meeting") as promptly as reasonably practicable following the mailing of the Proxy Statement to Seller Parent's stockholders, for the purpose of obtaining the Stockholder Approval. Notwithstanding anything to the contrary in this Agreement, Seller Parent shall convene, subject to Section 5.6(f), and hold the Stockholder Meeting as soon as practicable (but in no event more than 60 days) following the mailing of the Proxy Statement to Seller Parent's stockholders or on such other date elected by Seller Parent with Buyer's prior written consent (which shall not to be unreasonably withheld, conditioned or delayed). Seller Parent shall use its reasonable best efforts to solicit proxies to obtain the Stockholder Approval.
- (f) Seller Parent shall cooperate with and keep Buyer informed upon Buyer's reasonable request regarding its solicitation efforts and voting results following the dissemination of the Proxy Statement to Seller Parent's stockholders. Seller Parent shall be permitted to postpone or adjourn the Stockholder Meeting (after consultation with Buyer) if, but only if, (i) as of the time that the Stockholder Meeting is originally scheduled (as set forth in the Proxy Statement), there are holders of an insufficient number of shares of Seller Parent's common

stock present or represented by proxy at the Stockholder Meeting to constitute a quorum at the Stockholder Meeting, (ii) Seller Parent has determined in good faith (after consultation with outside legal counsel) that it is required by applicable Law or Governmental Order to postpone or adjourn the Stockholder Meeting, (iii) Seller Parent is requested to postpone or adjourn the Stockholder Meeting by the SEC or its staff or (iv) with the prior written consent of Buyer (which consent shall not be unreasonably withheld, conditioned or delayed).

- **5.7** <u>Interim Period Operations</u>. Nothing contained in this Agreement shall give any Party, directly or indirectly, the right to control or direct the operations of the other Party prior to the Closing. Prior to the Closing, each Party shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Affiliates' respective businesses and operations.
- 5.8 Assigned Locations. Schedule 5.8 (the "Assigned Real Property List") sets forth the following: (a) the patient service center locations that Buyer will acquire under this Agreement (the "Assigned Locations"), (b) the Assigned Real Property Leases related to such Assigned Locations that will be Assigned Contracts under this Agreement and (c) the Assigned Real Property Leases for which Buyer is requesting or will request a renewal, extension or amendment (each such renewal, extension or amendment, a "Lease Extension"). Buyer's requested terms and conditions of a Lease Extension will be those that are reasonably necessary to ensure that the leases for the Assigned Locations have rental terms and rental rates that are at fair market value, in each case as may be required, for compliance purposes and other applicable regulatory requirements. In addition Seller shall take such actions with respect to the Assigned Locations as described on Schedule 5.8. Seller shall use commercially reasonable efforts to seek an assignment of lease in a form and substance of the Assignment and Assumption of Lease, a landlord disclosure letter and lessor IRS Form W-9 for each Assigned Real Property Lease and any Consents, Lease Extensions or other documentation required to assign to Buyer the Assigned Real Property Leases and/or enter into any Lease Extension with the applicable landlord for each applicable Leased Real Property requested by Buyer as promptly as practicable after the Effective Date and in any event prior to the Closing Date. Each Consent and Lease Extension related to a Real Property Lease for an Assigned Location shall be in form and substance reasonably acceptable to Buyer and Seller shall provide Buyer and its Representatives with an opportunity to review and comment on all Consents, Lease Extensions and other documents related to the Assigned Locations and with periodic updates regarding the status of any required Consents or Lease Extensions.
- 5.9 <u>Transfer of Warranties</u>. As of the Closing Date, to the extent assignable, Seller shall be deemed (to the extent legally permissible) to have assigned to Buyer all of its right, title and interest in and to such warranties (express and implied) that continue in effect with respect to any of the Acquired Assets, and to have nominated Buyer as Seller's true and lawful attorney to enforce such warranties against such manufacturers, and Seller shall execute and deliver such specific assignments of such warranty rights as Buyer may reasonably request.
- 5.10 Seller Activities. Prior to the expiration of the term of the Amended STAT Services Agreement (or, if later, the term of the Transition Services Agreement (if any)), Seller and Seller Parent shall not sell or transfer (whether by sale of assets, assignment, rental or otherwise) to any Person any Excluded Asset (including STAT laboratories and any real property of Seller or an Affiliate located in Farmingdale, NY that is not an Assigned Location) that could reasonably be expected to be used by Seller and Seller Parent in connection with providing the STAT Services unless such Person acquires all or substantially all of the Excluded Business and expressly assumes all of the obligations of Seller and Seller Parent under the Amended STAT Services Agreement. Following the expiration of the term of the Amended STAT Services Agreement (or, if later, the term of the Transition Services Agreement (if any)), Seller and Seller Parent may sell or transfer (whether by sale of assets, assignment, rental or otherwise) to any Person any Excluded Asset (including STAT laboratories and any real property of Seller or an Affiliate located in Farmingdale, NY), subject to Section 7.14.
- **5.11** Notice of Developments. From the date of this Agreement until the Closing Date, Seller will give Buyer reasonably prompt notice upon becoming aware, and shall in any event give such notice no later than 48 hours of becoming aware, of (a) any material development that would cause the satisfaction of the conditions set forth in Article VI to not be met at the Closing, (b) any material development related to any Assigned Location (including with respect to the operations at such Assigned Location, any Assigned Real Property Lease for such Assigned Location or any Material Permit used or held for use in connection with such Assigned Location or the operation of the Business in connection with such Assigned Location) and/or (c) any material development with respect to the sales representatives and the senior-level managers of the Business listed in Schedule 5.11, including the resignation (or notification of an intent to resign) or termination of any such sales representatives or senior-level managers of the Business.

ARTICLE VI CLOSING CONDITIONS

- **6.1** <u>Seller's and Seller Parent's Closing Conditions</u>. The obligations of Seller and Seller Parent to effect the transactions contemplated by this Agreement (other than the Final Contemplated Transactions) shall be subject to the satisfaction and fulfillment, at or prior to the Closing, of each of the following conditions, any of which may be waived in writing by Seller and Seller Parent:
- (a) <u>Compliance with Agreements and Covenants</u>. Buyer shall have performed and complied in all material respects with each of the covenants, obligations, and agreements contained in this Agreement to be performed and complied with by Buyer on or prior to the Closing Date.
- (b) Representations and Warranties. Each of (i) the representations and warranties of Buyer contained in Section 4.1 (Organization, Standing and Power; No Subsidiaries), Section 4.2 (Authority), Section 4.3(a)(i) (No Conflict; Required Filings and Consents) and Section 4.4 (Brokers or Finders) shall be true and correct in all respects on and as of the Effective Date and shall also be true and correct in all respects on and as of the Closing Date with the same force and effect as though made on and as of the Closing Date (in each case, other than de minimis inaccuracies) and (ii) the other representations and warranties of Buyer contained herein shall be true and correct in all respects on and as of the Effective Date and shall also be true and correct in all respects on and as of the Closing Date with the same force and effect as though made on and as of the Closing Date, except in the case of clause (ii) where the failure of such other representations and warranties to be so true and correct, individually or in the aggregate, does not, and would not reasonably be expected to, prevent Buyer from consummating the transactions contemplated by this Agreement.
- (c) <u>Closing Documents</u>. Buyer shall have delivered to Seller duly executed (if applicable) counterparts to the documents and the deliveries as set forth in <u>Section 2.8(b)</u>.
- **6.2** Buyer's Closing Conditions. The obligations of Buyer to effect the transactions contemplated under this Agreement shall be subject to the satisfaction and fulfillment, at or prior to the Closing, of each of the following conditions, any of which may be waived in writing by Buyer:
- (a) <u>Compliance with Agreements and Covenants</u>. Each of Seller and Seller Parent shall have performed and complied in all material respects with each of the covenants, obligations, and agreements contained in this Agreement to be performed and complied with by Seller and Seller Parent, as applicable, on or prior to the Closing Date.
- (b) Representations and Warranties. Each of (i) the representations and warranties of Seller and Seller Parent contained in Section 3.5(f) through Section 3.5(g) (Legal Compliance), Section 3.8(a) (Good Title and Use of Assets), the second sentence of Section 3.8(b) (Good Title and Use of Assets), the first sentence of Section 3.28(b) (Intellectual Property) and the second sentence of Section 3.28(d) (Intellectual Property) shall be true and correct in all material respects on and as of the Effective Date and shall also be true and correct in all material respects on and as of the Closing Date with the same force and effect as though made on and as of the Closing Date (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" set forth therein), (ii) the remainder of the Fundamental Representations shall be true and correct in all respects on and as of the Effective Date and shall also be true and correct in all respects on and as of the Closing Date with the same force and effect as though made on and as of the Closing Date, and (iii) the other representations and warranties of Seller and Seller Parent contained herein shall be true and correct in all respects on and as of the Effective Date and shall also be true and correct in all respects on and as of the Closing Date with the same force and effect as though made on and as of the Closing Date (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" set forth therein), except in the case of clause (iii) where the failure of such other representations and warranties to be so true and correct, individually or in the aggregate, does not, and would not reasonably be expected to, have a Material Adverse Effect. Notwithstanding anything to the contrary herein, to the extent any Schedule or any section of the Disclosure Schedule is amended, supplemented, qualified or modified in any way prior to the Closing, (i) the condition set forth in this Section 6.2(b), (ii) the certificate delivered pursuant to Section 2.8(a)(ii) and (iii) any indemnification obligations arising under or in connection with Section 9.1(a)(i) shall each be with respect to the applicable representations and warranties as amended, supplemented, qualified or modified by any such Schedule or section of the Disclosure Schedule, as applicable.

- (c) <u>Closing Documents</u>. Seller and Seller Parent, as applicable, shall have delivered to Buyer duly executed (if applicable) counterparts to the documents and the deliveries as set forth in <u>Section 2.8(a)</u>.
- (d) NYS Form AU-196.10. Buyer shall not have received Form AU-197.2 from the NYS Department of Taxation and Finance in response to Buyer's filing of Form AU-196.10 pursuant to Section 7.5(e).
- (e) <u>Material Adverse Effect</u>. Since the Effective Date, there shall not have occurred and be continuing any Material Adverse Effect.
 - (f) Seller and Seller Parent Actions Seller and Seller Parent shall have delivered to Buyer:
 - (i) a certificate of the Secretary of each of Seller and Seller Parent certifying, as complete and accurate as of the Closing, (A) attached copies of the bylaws of such Person, (B) attached copies of all requisite resolutions of such Person's board of directors and stockholders approving the execution, delivery and performance of each Transaction Document to which such Person is a party and the obligations of such Person hereunder and thereunder with a statement that such resolutions are in full force and effect as of the Closing Date, and (C) certifying to the incumbency of the officers of such Person executing any Transaction Document; and
 - (ii) a certificate of status with respect to each of Seller and Seller Parent, issued by the State of New York, showing that such Person is in good standing in such jurisdiction, dated within ten Business Days prior to the Closing Date.
- **6.3** Conditions to Obligations of All Parties to Consummate the Closing. The obligations of each Party to consummate the transactions contemplated by this Agreement shall be subject to the fulfillment, at or prior to the Closing, of the following conditions:
- (a) No Governmental Order or Pending Actions No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any Governmental Order which is in effect and that prevents or prohibits the consummation of the transactions contemplated by this Agreement and the other Transaction Documents, and no Governmental Authority shall have instituted any Action (which remains pending at the Closing Date) which seeks to enjoin or prohibit the consummation of the transactions contemplated by this Agreement or the other Transaction Documents (collectively, a "Restraint"). For the avoidance of doubt, the receipt of a letter from the Federal Trade Commission (FTC) or the U.S. Department of Justice's Antitrust Division (DOJ) similar to the content of the form letter announced and disclosed by the FTC on August 3, 2021 shall not be deemed to be a "Restraint" for the purposes of this Agreement, nor shall it be used as, or deemed to be, the basis for concluding that any Closing condition is not, or will not be, satisfied for the purposes of this Article VI.
- (b) <u>HSR Clearance</u>. Any and all waiting periods applicable to the transactions contemplated by this Agreement and the other Transaction Documents under the HSR Act shall have expired or been terminated.
- (c) <u>Stockholder Approval</u>. The Stockholder Approval shall have been obtained in accordance with applicable Law.

ARTICLE VII ADDITIONAL AGREEMENTS

7.1 <u>Covenant to Satisfy Conditions</u>. Subject to the terms and conditions hereof, each Party shall use its commercially reasonable efforts to take all action required of it to satisfy the conditions set forth in <u>Article VI</u>, and otherwise to fulfill its obligations under the terms of this Agreement and to facilitate the consummation of the transactions contemplated hereby.

7.2 Confidentiality Agreement.

(a) The Parties acknowledge that Seller Parent and Buyer have previously executed that certain Confidentiality Agreement, dated as of November 22, 2022 (the "Confidentiality Agreement"), which shall continue in full force and effect in accordance with its terms except as expressly modified by this Agreement. For clarity, to the extent there is any conflict between the terms of the Confidentiality Agreement and the terms of this Agreement, the terms of this Agreement shall control.

- (b) Notwithstanding anything contained herein to the contrary, effective as of the Closing, all Confidential Information of Seller and Seller Parent included in the Acquired Assets or otherwise related to the Business will be deemed to be Confidential Information of Buyer and will be subject to the protections set forth herein and in the Confidentiality Agreement for the benefit of Buyer. Each of Seller and Seller Parent agrees that if the transactions contemplated by this Agreement are consummated, from and after the Closing, it shall, and it shall direct its Affiliates, directors, officers, partners, members and advisors to keep all such Confidential Information strictly confidential, and use such Confidential Information (including the terms and conditions of this Agreement) only for the purposes of fulfilling its obligations hereunder and for no other purpose whatsoever. Each Party shall be liable for all actions taken by its Affiliates, directors, officers, partners, members, Representatives and advisors that are inconsistent with this Section 7.2, and which are known by such Party at such time, as if the actions taken by any such Person were the actions of such Party.
- (c) To the extent either Party is required to disclose any materials, documents, correspondences or information provided to it by the other Party, such Party will provide the disclosing Party with prompt notice of any such request or requirement so that the disclosing Party may seek a protective order or other appropriate remedy (and, if the disclosing Party seeks such an order, the receiving Party (or its applicable Affiliates) agrees to provide such cooperation as the disclosing Party shall reasonably request).
- 7.3 Trading Prohibitions. Each Party hereby acknowledges that as a result of disclosures by the Parties contemplated under this Agreement, each Party and its Affiliates may, from time to time, have material, non-public information concerning the other Party and other Persons. Each Party confirms that it and its Affiliates are aware and it has advised its Representatives that (a) the United States securities Laws may prohibit a Person who has material, non-public information from purchasing or selling securities of any company to which such information relates, and (b) material non-public information shall not be communicated to any other Person except as permitted herein.
- 7.4 Public Statements and Internal Communications Except as may be required by applicable Law or the rules and regulations of any national securities exchange or national securities quotation system, the Parties shall consult with each other and obtain the prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed) of the other Party prior to issuing any press release or any written public statement with respect to this Agreement, the other Transaction Documents, or the transactions contemplated hereby or thereby, and shall not issue any such press release or written public statement prior to such consultation and obtaining such prior written consent; provided that, the Party issuing the press releases or statements may proceed without such comments or consent if such comments or consent are not timely provided. The Parties further agree that they will work together in good faith to develop a communications plan which shall memorialize the timing and manner in which messages related to this Agreement will be communicated to Seller's employees and Customers.

7.5 <u>Taxes</u>.

- (a) Each Party shall reasonably cooperate in the preparation, execution, and filing of all returns, questionnaires, applications or other documents regarding any property transfer or gains, sales, use, value added, business and occupancy, stock transfer and stamp Taxes, any transfer, recording, registration and other fees or any similar Taxes that become payable in connection with the transfer of Acquired Assets pursuant to this Agreement ("Transaction Taxes"). Seller shall file all Tax Returns with respect to such Transaction Taxes consistent with the principles and allocation set forth in Section 2.6 and pay any such Transaction Taxes and Buyer shall promptly reimburse Seller for one-half of such Transaction Taxes. For the avoidance of doubt, any sales and use Taxes payable in connection with the transactions contemplated by this Agreement shall be payable by Seller in connection with the first applicable sales or use Tax Return due after the relevant payment date. The Parties shall cooperate in good faith to minimize any such Transaction Taxes and complete any tax forms to qualify the transactions contemplated by this Agreement for reduction of or exemption from any otherwise applicable Transaction Taxes.
- (b) Seller shall be solely responsible for and shall pay, without any cost to Buyer, any and all Taxes assessed against or payable by Seller or Seller Parent, arising from the operation of Seller or Seller Parent, the Business, the Assumed Liabilities or the Acquired Assets (in each case regardless of whether the filing of any Tax Return with respect thereto or payment of any amount in respect thereof is filed, paid or due prior to, on or after the Closing Date, as applicable) or relating to the Excluded Assets or Excluded Liabilities. Liability for all real property Taxes, personal property Taxes and similar ad valorem obligations (the "Apportioned Obligations") levied with respect to the Business, the Acquired Assets and the Assumed Liabilities (individually or in the aggregate) for any Tax period beginning on or before the Closing Date and ending after the Closing Date (a "Straddle Period") shall

be apportioned between Seller and Buyer based on the number of days of such Straddle Period included in the Tax period ending on the Closing Date (a "Pre-Closing Tax Period") and the number of days of such Straddle Period included in the Tax period beginning after the Closing Date (a "Post-Closing Tax Period"). Seller shall be liable for the proportionate amount of such Apportioned Obligations that is attributable to the Pre-Closing Tax Period and Buyer shall be liable for the proportionate amount of such Apportioned Obligations that is attributable to the Post-Closing Tax Period.

- (c) Buyer, Seller and their respective Affiliates shall provide each other with such assistance as may reasonably be requested in connection with the preparation of any Tax Return, application for exemption or refund of Tax, audit or other examination by any Governmental Authority relating to Liabilities for Taxes, or action, suit, proceeding, claim, arbitration or investigation relating to Liability for Taxes, in all cases to the extent in connection with the Acquired Assets or the Business. Such assistance shall include (upon the request of Seller or Buyer, as applicable) the provision of records and information which are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Subject to Section 7.5(b), Seller and Seller Parent will prepare and timely file all Tax Returns and pay all Taxes owed in respect of the Acquired Assets and the Business for all Tax periods ending on or prior to the Closing Date including any Tax preparation fees. Buyer will prepare and timely file all other Tax Returns that are required to be filed in respect of the Acquired Assets or the Business, including any Tax preparation fees. Seller will pay any Apportioned Obligations with respect to the Business, the Acquired Assets and the Assumed Liabilities that are due and payable on or prior to the Closing Date (including any Tax preparation fees), and bill Buyer for any part of that amount apportioned to Buyer pursuant to Section 7.5(b) (including the proportionate amount of any Tax preparation fees). Buyer will pay such Apportioned Obligations with respect to the Business, the Acquired Assets and the Assumed Liabilities that are due and payable after the Closing Date and bill Seller for any part of that amount apportioned to Seller pursuant to Section 7.5(b) (including the proportionate amount of any Tax preparation fees).
- (d) Notwithstanding anything herein to the contrary, Buyer shall be entitled to deduct and withhold from any consideration otherwise payable pursuant to this Agreement, such amounts as may be required to be deducted and withheld with respect to the making of such payment under applicable Law; provided that Buyer shall use commercially reasonable efforts to notify Seller or the applicable payee in advance if it determines that any amounts are required to be withheld from any consideration otherwise payable hereunder (other than any amounts treated as compensation) and shall use commercially reasonable efforts to cooperate in obtaining any available exemptions or reductions from such withholding. To the extent that amounts are so deducted or withheld and paid over to the applicable Governmental Authority or other appropriate Person, such amounts will be treated for all purposes of this Agreement as having been paid hereunder.
- (e) At least 10 days prior to the Closing, Buyer shall file Form AU-196.10 with the NYS Department of Taxation and Finance.
- 7.6 Pre-Closing and Post-Closing Information; Employee Accessibility. During the Interim Period, and upon reasonable advance notice received from Buyer, Seller shall (a) afford Buyer Group reasonable access during normal business hours to Seller's properties, personnel, facilities, contracts, books and records, and other documents and data, in each case solely relating to the Business or the Acquired Assets, such rights of access to be exercised in a manner that does not unreasonably interfere with the operations of Seller, (b) furnish to Buyer Group copies of all such Contracts, books and records, and other existing documents and data, in each case solely relating to the Business, the Acquired Assets or Assumed Liabilities, that Buyer Group may reasonably request, (c) furnish Buyer Group with such additional financial, operating, and other relevant data and information, in each case solely relating to the Business, the Acquired Assets or the Assumed Liabilities, as Buyer Group may reasonably request, and (d) otherwise cooperate and assist, to the extent reasonably requested by Buyer Group, with Buyer Group's investigation of the properties, assets and financial condition of Seller and its Affiliates solely to the extent related to the Business; provided that such access or disclosure (x) shall be subject, in the case of competitively sensitive information, to any "clean-room" arrangements agreed between the Parties, and (y) may be restricted by Seller to the extent, in its reasonable judgment, may be required to preserve any applicable attorney-client privilege, to comply with any applicable Law, or to comply with any confidentiality obligation with any third party. Buyer Group will hold any such information that is non-public in confidence to the extent required by, and in accordance with, the provisions of the Confidentiality Agreement.

7.7 <u>Bulk Sales Laws</u>. The Parties hereby waive compliance with the provisions of any bulk sales, bulk transfer or similar Laws of any jurisdiction that may otherwise be applicable with respect to the sale of any or all of the Acquired Assets to Buyer.

7.8 Reconciliation. If, following the Closing, any Excluded Assets or Excluded Liabilities related to such Closing are inadvertently transferred or assumed, respectively, from Seller to Buyer or otherwise, or Buyer collects or receives any remittances, payments, mail or other communications relating to any such Excluded Asset, then Seller and Buyer shall execute, deliver and record (where appropriate) any and all instruments or other documents of transfer, conveyance and assignment, or amend or correct any such existing instruments or documents, and take such other action as Seller or Buyer may reasonably request, as may be reasonably necessary or advisable to effect or evidence the transfer of such Excluded Assets or Excluded Liabilities to Seller (or to any Person as directed by Seller) in accordance with the terms of this Agreement, and to thereafter deliver as soon as reasonably practicable after receipt of such Excluded Assets, and any remittances, payments, mail or other communications relating thereto, and such Excluded Liabilities to Seller. If, following the Closing, any Acquired Assets or Assumed Liabilities are inadvertently retained by Seller, Seller Parent or their Affiliates, or Seller, Seller Parent or their Affiliates collect or receive any remittances, payments, mail or other communications relating to any such Acquired Asset, then Seller, Seller Parent or their Affiliate and Buyer, as applicable, shall execute, deliver and record (where appropriate) any and all instruments or other documents of transfer, conveyance and assignment, or amend or correct any such existing instruments or documents, and take such other action as Seller or Buyer may reasonably request, as may be reasonably necessary or advisable to effect or evidence the transfer of such Acquired Assets or Assumed Liabilities to Buyer (or to any Person as directed by Buyer) in accordance with the terms of this Agreement, and to thereafter deliver as soon as reasonably practicable after receipt of any remittances, payments, mail or other communications relating to the Acquired Assets to Buyer. Seller hereby authorizes Buyer from and after the Closing Date to receive and open all mail and other communications received by Buyer which relate to the Business following the Closing or to the Acquired Assets, and to act with respect to such communications in such manner as Buyer may elect, and, if such communications do not so relate to the Business following the Closing or to the Acquired Assets, to forward them promptly to Seller. Subject to Section 7.10, Seller hereby authorizes Buyer from and after the Closing Date to endorse without recourse the name of Seller on any checks or other instruments of payment received by Buyer on account of any services rendered by Buyer to a Customer after the Closing Date; provided, however, that if any such check or other instrument of payment is provided in settlement of a dispute, then Buyer shall not be permitted to endorse the name of Seller on such instrument without Seller's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed.

7.9 Employees.

(a) Not less than two weeks prior to the Closing Date, Buyer shall notify Seller of those Business Employees to whom Buyer or an Affiliate of Buyer shall offer employment effective as of the Closing Date. Business Employees to be offered employment by Buyer must be employed by Seller or Seller Parent as of the Closing and must meet the standards of employment applicable to Buyer's employees generally, as Buyer reasonably determines in its sole discretion (each, an "Offered Employee"). Any Offered Employee who accepts and commences active employment with Buyer effective as of the Closing Date shall be referred to herein as a "Hired Employee", with such employment to be conditioned upon the consummation of the transactions contemplated by the Transaction Documents; provided, however, that with respect to any Offered Employee who is not actively employed at the time such offers are made because such employee is on a leave of absence (other than regularly scheduled vacation) (the "Leave of Absence Employees"), Buyer shall be under no obligation to make an offer to any such Leave of Absence Employees unless and until such Leave of Absence Employee returns to active employment on or within 12 months following the Closing Date; and provided, further, that, notwithstanding anything in this Agreement to the contrary, Buyer shall have no Liability with respect to any Leave of Absence Employee unless such employee returns to active employment on or within 12 months of the Closing Date and accepts Buyer's offer of employment within such time; and provided, further, that Buyer shall not use any such leave of absence as cause to modify, alter, or change any proposed offer of employment to the detriment of such Leave of Absence Employee. All Hired Employees, other than Leave of Absence Employees, shall be employed on a substantially uninterrupted basis by Buyer as of the Closing Date.

(b) As of the Closing Date each Hired Employee who is actively at work as of the Closing Date shall receive from Buyer, until the first anniversary of the Closing Date, salary, bonus, severance and employee benefit plan opportunities that are substantially comparable to the salary, bonus, severance and employee benefits provided to similarly situated employees of Buyer.

(c) Subject to Information Privacy and Security Laws and other applicable Laws, Seller will make available to Buyer, at least six weeks prior to the Closing, the following true, accurate, and complete information regarding each Business Employee: (i) name and employee ID number, (ii) employing entity, (iii) salary or other rate of base pay, if applicable, (iv) other compensation (including bonus target, incentive compensation target, deferred compensation, and commissions arrangements), (v) severance and employee benefit plan opportunities, (vi) status as an exempt or non-exempt under the Fair Labor Standards Act and any similar wage and hour law, (vii) status as a full-or part-time employee, (viii) job title or position, (ix) status as an active or inactive employee (and, if on leave, the nature of the leave and the expected return date, if known), (x) work authorization, if not a citizen or permanent resident, (xi) address, and (xii) whether such Business Employee is primarily or exclusively working for the Business. Unless Buyer otherwise consents in writing, every two weeks following the delivery of the information pursuant to the preceding sentence Seller shall provide Buyer with written notice of any updates to the Business Employee information provided pursuant to the preceding sentence and Buyer and Seller shall in good faith mutually agree upon the form of a written communication to be provided by Buyer to the applicable Business Employees describing the offer process; provided, that Buyer shall allow access to the information requested in this Section 7.9(c) only those Representatives of Buyer who need to know such information for purposes of making employment offers to the Business Employees.

(d) Seller, Seller Parent and Buyer intend that the transactions contemplated by this Agreement shall not result in a severance of employment of any Hired Employee for purposes of any benefit plan maintained by Seller or Seller Parent ("Seller Plans"), that the Hired Employees shall have continuous and uninterrupted employment immediately before, on, and immediately after the Closing Date, and Buyer, Seller and Seller Parent shall use reasonable efforts to ensure the same. For purposes of eligibility to participate, vesting, benefit accruals and level of benefits (except as otherwise specifically provided below) under any benefit plan maintained by Buyer and/or its Affiliates providing benefits to any Hired Employee after the Effective Date, each Hired Employee shall be credited with his or her years of service with Seller and its Affiliates (and any predecessors) prior to the Closing Date, to the same extent as such Hired Employee was (or would have been) entitled, before the Effective Date, to credit for such service under Seller Plans; provided, however, that such service need not be recognized (i) to the extent that such recognition would result in any duplication of benefits for the same period of service, (ii) for any purpose under any defined benefit retirement plan, retiree welfare plan, equity-based incentive plan or long-term incentive plan (including any cash balance plan or employee stock purchase plan of Buyer or its Affiliates), (iii) to the extent not recognized by Seller or Seller Parent for similar purposes, (iv) for purposes of any plan, program or arrangement (A) under which similarly situated employees of Buyer and its Affiliates do not receive credit for prior service or (B) that is grandfathered or frozen, either with respect to level of benefits or participation, or (v) discretionary contributions under Buyer's employees' retirement savings plan. To the extent that Buyer maintains a similar Employee Plan providing health and welfare benefits, for each Hired Employee, Buyer and/or its Affiliates shall take commercially reasonable steps waive any applicable waiting periods, pre-existing conditions or actively-at-work requirements and shall give such employees credit under the new coverages or benefit plans for deductibles, coinsurance and out-of-pocket payments that have been paid during the year in which each Hired Employee begins participation in the applicable Employee Plan. With respect to vacation benefits, effective as of the Closing Date, Buyer shall credit any Hired Employee with up to ten days of accrued, but unused, vacation time under any benefit plan maintained by Buyer and/or its Affiliates providing similar benefits after the Effective Date. Any credit of vacation days provided to Hired Employees pursuant to the foregoing sentence shall be at Seller's expense and taken into account in the calculation of the Vacation Rollover Amount. Seller shall be solely responsible, and Buyer shall have no obligations whatsoever for, any amounts accrued or payable to any current or former employees (including Hired Employees), including accrued vacation for which Buyer is not providing credit for any period relating to the service with Seller or Seller Parent, as applicable, at any time on or prior to the Closing Date or due to termination by Seller or its Affiliates on or prior to the Closing Date.

(e) If, as a result of the transactions contemplated by this Agreement, Seller or any of its Affiliates are required before, on or after the Closing to provide notice of any employment loss to any Business Employee or other Person under the WARN Act or any foreign, state or local laws, regulations or ordinances related to plant closings, relocations, mass layoffs or employment losses, Seller shall provide, or arrange for the provision of, all such required notices and make any required payments within the time periods required by applicable Law (collectively, "WARN Laws"). Seller shall provide Buyer with a reasonable opportunity to review and comment upon all such notices before the notices required by WARN Laws are provided to affected employees, governmental agencies or other Persons. Seller shall bear any and all obligations and Liability under the WARN Laws resulting

from employment losses arising in connection with the transactions contemplated in this Agreement, including without limitation the provision or failure to provide notice requirements, including any required notices to Hired Employees, and make any required payments under the WARN Laws. Buyer shall use its commercially reasonable efforts to assist Seller and Seller Parent in mitigating all such obligations and Liabilities under WARN Laws.

- (f) The provisions of this <u>Section 7.9</u> are solely for the benefit of the Parties, and no provision of this <u>Section 7.9</u> is intended to, or shall, constitute the establishment or adoption of or an amendment to any employee benefit plan for purposes of ERISA or otherwise, or result in any current or former director, employee, consultant of Seller or Seller Parent or any other individual associated therewith being regarded for any purposes as a third party beneficiary of this Agreement or have the right to enforce the provisions hereof.
- 7.10 Accounts Receivable. Seller shall retain all accounts receivable arising out of the operation of the Business and specimens processed by Seller prior to the Closing (subject to collection consistent with past practices), plus accounts receivable from the Add-On Testing Services, and Buyer shall retain all accounts receivable arising out of the operation of the Business and specimens processed by Buyer on or after the Closing. After the Closing, Buyer and Seller shall promptly forward to the other Party any funds which are received by such Party but relate to the accounts receivable of the other Party. Notwithstanding anything to the contrary stated herein, neither Party shall have any responsibility to collect any of the other Party's accounts receivable; provided, that Buyer shall use commercially reasonable efforts to cooperate with and assist Seller and its Affiliates and Representatives in the prompt collection of all accounts receivable owing to Seller.
- 7.11 <u>Data Room</u>. Promptly following the Closing, Seller shall deliver to Buyer (at Buyer's expense) a DVD, CD, flash drive or similar electronic storage medium containing the documents in the "data room" created and maintained by Seller and hosted by Venue by DFIN in connection with Buyer's review of the Business (the "Data Room").
- 7.12 <u>R&W Policy</u>. At or prior to the Closing, Buyer shall obtain and conditionally bind coverage under the R&W Policy. Buyer shall pay or cause to be paid, all costs and expenses related to the R&W Policy, including the total premium, underwriting costs, brokerage commissions, and other fees and expenses under the R&W Policy. The R&W Policy shall provide that the insurer of such policy shall waive its rights to bring any claim against Seller or its Affiliates by way of subrogation, claim for contribution or otherwise (other than in the case of Fraud), and Seller and its Affiliates shall be express third party beneficiaries of such provision.
- 7.13 Buyer Non-Solicitation. If this Agreement is terminated pursuant to Article VIII, Buyer agrees that, except as expressly permitted by this Agreement or with the prior written consent of Seller, during the Nonsolicit Period, Buyer shall not, directly or indirectly through one or more of its Affiliates, do any one or more of the following: cause, induce or solicit, or attempt to cause, induce or solicit any of the employees, officers, or directors of Seller to terminate such Person's employment relationship with Seller, Seller Parent or any of their Affiliates; provided, however, that nothing contained herein shall be deemed to prohibit Buyer or any of its Affiliates from (a) conducting any general solicitation or general advertisement of employment opportunities (including through the use of employment agencies or the internet) not specifically targeted at any such Persons, (b) hiring any such Person who responds to any such general solicitation or general advertising, (c) contact or solicitations (or resulting employment or engagement) made by any of Buyer's or its Affiliate's employees, agents or independent contractors who are not aware of the transactions contemplated by this Agreement, this Agreement or the restrictions contained herein, and who are not directed or encouraged to make such contact or solicitations by Buyer's employees, agents or independent contractors who have knowledge of the transactions contemplated by this Agreement, this Agreement or the restrictions contained herein, or (d) contact with or the solicitation or hiring of any director or officer who independently initiates employment discussions with Buyer, its Affiliates, or their employees agents or independent contractors. As used in this Section 7.13, "Nonsolicit Period" shall mean the 18 month period following the termination of this Agreement pursuant to Article VIII. Upon termination of this Agreement pursuant to Article VIII, Section 7.13 shall supersede and replace similar terms set forth in the Confidentiality Agreement.

7.14 Restrictive Covenants.

(a) Except as set forth on <u>Schedule 7.14(a)</u>, for a period of five years after the Closing Date (the "*Restricted Period*"), Seller and Seller Parent hereby covenant and agree that they shall not, directly or indirectly through one or more of their Affiliates or Representatives, (i) engage, invest in, own, manage, operate, finance, control, advise, render services to, be employed by any Person (including direct competitors of Buyer) engaged in any

business or activity that competes with the Business as conducted on the Effective Date or on the Closing Date in the States of New York, New Jersey and Connecticut; (ii) use any names licensed to Buyer in any business that competes with the Business; or (iii) use the trade names "Enzo Clinical Labs" or "Enzo Labs" or any other confusingly similar trade name that combines the names "Enzo" and "labs" in connection with any business of any kind.

- (b) During the Restricted Period, Seller and Seller Parent hereby covenant and agree that they shall not, directly or indirectly through one or more of their Affiliates or Representatives cause, induce or solicit, or attempt to cause, induce or solicit, (i) any Person who is or was an employee (including any temporary or leased employees), consultant or independent contractor of the Business on, or during the 12 months prior to, the Closing Date; or (ii) any Person who is or was an employee (including any temporary or leased employees), consultant or independent contractor of Buyer or any of its Affiliates at any time during the Restricted Period with whom Seller or Seller Parent (as applicable) has or has had contact (or with respect to whom Seller or Seller Parent obtains or obtained confidential information) during the Restricted Period to (A) terminate such Person's employment or consulting relationship with Buyer or any of Buyer's Affiliates; (B) in any way interfere with the relationship between such Person and Buyer or any of its Affiliates; or (C) hire, retain, employ or otherwise engage, or attempt to hire, retain, employ or otherwise engage, such individual as an employee, independent contractor or otherwise, unless (x) such individual's employment with Buyer has been involuntarily terminated by Buyer or its Affiliate or (y) such individual's employment has been voluntarily terminated by such employee not less than 12 months prior to any solicitation from Seller or Seller Parent or any of its Affiliates.
- (c) During the Restricted Period, Seller and Seller Parent hereby covenant and agree that they shall not, directly or indirectly through one or more of their Affiliates or Representatives: (i) solicit, induce or otherwise cause, or attempt to solicit, induce or otherwise cause, any customer, supplier, licensor, licensee or any prospective customer, supplier, licensor or licensee that has been contacted or targeted for contact by Seller or Seller Parent on or before the Closing Date, or any other Person that had, has or is expected to have a business relationship with Seller or Seller Parent, to (A) terminate, curtail or otherwise modify its relationship with the Business or (B) engage in business with a competitor of the Business; or (ii) interfere in any way with the relationship between the Business and any of its customers, suppliers, licensoes, licensees or any such prospective customers, suppliers, licensors or licensees that have been contacted or targeted for contact by the Business on or before the Closing Date, or any other Person engaged in a business relationship with the Business; provided, that nothing in this Section 7.14 shall prohibit Seller or Seller Parent from soliciting any such existing or prospective customer, supplier, licensor or licensee solely with respect to the Excluded Business.
- (d) No Party shall make any disparaging statement, either orally or in writing, regarding the other Parties, the Business, or any products or services thereof, or any Affiliates or Representatives of the Parties or the Business; <u>provided</u> that such prohibited statements do not include statements that are made for the purpose of protecting or enforcing their rights or interests hereunder or defending against any claim brought by either Party hereunder, and the Parties shall not be prohibited from making any truthful statements in any Action, or as otherwise required by applicable Law. Further, nothing herein shall be deemed to limit competitive speech or commercial comparisons regarding services or products of the Parties following the expiration of the Restricted Period.
- (e) Seller and Seller Parent, by their respective execution of this Agreement, hereby agree that this <u>Section 7.14</u>, including the provisions relating to duration, geographical area and scope, is reasonable and necessary to protect and preserve Buyer's legitimate business interests and the value of the Business, and to prevent an unfair advantage from being conferred on Seller or Seller Parent.
- (f) If any provision of this <u>Section 7.14</u> is held to be excessively broad as to duration, geographical area, scope, activity or subject, for any reason, such provision shall be modified, by limiting and reducing it, so as to be enforceable to the extent allowed by applicable Law.
- (g) Seller and Seller Parent, by their respective execution of this Agreement, hereby acknowledge that any breach of this Section 7.14 may result in serious and irreparable injury to Buyer, Buyer could not be adequately compensated by monetary damages alone, and Buyer's remedy at law would not be adequate. Therefore, Seller and Seller Parent hereby acknowledge and agree that, in the event of a breach of this Section 7.14. Buyer shall be entitled to seek, without proof of actual damages, in addition to any other remedy at law or in equity to which Buyer may be entitled, to equitable relief against Seller or Seller Parent, including temporary restraining orders and preliminary and permanent injunctions to restrain Seller or Seller Parent from such breach and to compel compliance with the obligations of Seller and Seller Parent.

- (h) In the event that all or substantially all of the Excluded Business is acquired by any Person or Persons other than Buyer or its Affiliates, whether in one transaction or a series of transactions and whether by merger, stock sale, sale of assets or otherwise, in lieu of the restrictions on competition set forth in Section 7.14(a) being applicable to such Person or any of its Affiliates (other than Seller and Seller Parent), Seller and Seller Parent shall ensure that any such Person and any of its Affiliates or their respective Representatives does not directly or indirectly have access to or use any Confidential Information retained by Seller and Seller Parent concerning Buyer or the Business (including data, books or records or other information related thereto), and no Acquired Assets or Confidential Information retained by Seller and Seller Parent concerning Buyer or the Business (including data, books or records or other information related thereto) is accessed or used directly or indirectly by such Person or any of its Affiliates or their respective Representatives, for the purposes of conducting any activities that would be prohibited by Section 7.14(a) or Section 7.14(b) if conducted by Seller or Seller Parent.
- 7.15 Payor Arrangements. (a). If and to the extent that any Governmental Program, any Private Program or any other payor, makes any claims or any offsets, withholds funds from Buyer, or requires Buyer to refund any payments for services rendered on or after the Closing Date, due to overpayments, duplicate payments, fraud, incorrect billing, retroactive denials or for any other reason relating to any action or inaction of Seller or its Affiliates in connection with the operation of the Business prior to the Closing Date (whether known or unknown as of the Closing) or pursuant to the Add-On Testing Services or STAT Services, Seller shall fully compensate Buyer for any Losses incurred and also satisfy any claims of Medicare, Medicaid, or such other payor.
- 7.16 <u>Transition Services</u>. If requested by Buyer no later than five Business Days prior to the Closing Date, Seller shall enter into an agreement(s) with Buyer to provide transition services reasonably necessary to allow Buyer to transition any remaining Customers of the Business to Buyer (if any Interface Configurations have not been completed five Business Days prior to the Final Transaction Expiration Date such that not all Customers of the Business can be transitioned to Buyer on or before the Closing Date) or to provide Buyer with access to any Assigned Locations not transferred to Buyer on the Closing Date, in each case, for up to six months following the Closing. Such transition services shall be provided to Buyer pursuant to a transition services agreement in a form to be mutually agreed (acting reasonably) by Buyer and Seller (the "*Transition Services Agreement*") prior to the Closing Date.

ARTICLE VIII TERMINATION

- **8.1** <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the transactions contemplated by this Agreement may be abandoned at any time prior to the Closing by the mutual written agreement of Buyer, Seller and Seller Parent.
- **8.2** <u>Termination by Seller or Buyer</u>. This Agreement may be terminated and the transactions contemplated by this Agreement may be abandoned by Seller, Seller Parent, on one hand, or Buyer, on the other hand, upon written notice to the other parties if:
- (a) the Closing shall not have occurred at or before 11:59 p.m. (New York City time) on the date that is 210 days after the Effective Date (the "Outside Date"); provided, that
 - (i) if, as of 11:59 p.m. (New York City time) on the Outside Date a Specified Circumstance exists and each of the conditions set forth in Section 6.3 (other than the conditions set forth in Sections 6.3(a) and (b)) and Section 6.1 (other than any requirement of receipt of a certificate regarding compliance with any such condition), in each case other than with respect to a Specified Circumstance, is satisfied or has been waived, then Buyer may, by providing written notice thereof to Seller at or prior to 11:59 p.m. (New York City time) on the Outside Date, extend the Outside Date to 270 days after the Effective Date and such date shall become the "Outside Date" for all purposes of this Agreement; provided further that if, on such extended Outside Date, a Specified Circumstance still exists but all other conditions to the Closing either have been satisfied or waived or would be satisfied if the Closing were to occur on such date, then Buyer may, by providing written notice thereof to Seller at or prior to 11:59 p.m. (New York City time) on the extended Outside Date, extend the Outside Date to 365 days after the Effective Date and such date shall become the "Outside Date" for all purposes of this Agreement;

<u>provided, however</u>, that the right to terminate this Agreement pursuant to this <u>Section 8.2(a)</u> shall not be available to any Party if: the failure of such Party (or any Affiliate of such Party) to perform any of its obligations under this Agreement has been a principal cause of or resulted in the failure of the Closing to occur on or before such date;

- (b) there shall be any Law that makes consummation of the transactions contemplated by this Agreement illegal or otherwise prohibited:
- (c) the Stockholder Meeting (including any postponements and adjournments thereof) shall have concluded without the Stockholder Approval having been obtained by reason of the failure to obtain the required vote of the holders of shares of Seller Parent's common stock; or
- (d) any Governmental Authority of competent jurisdiction shall have issued a Governmental Order permanently restraining or enjoining the consummation of the transactions contemplated by this Agreement, and such Governmental Order shall have become final and non-appealable.
- **8.3** Termination by Buyer. This Agreement may be terminated and the transactions contemplated by this Agreement may be abandoned at any time prior to the Closing by Buyer upon written notice to Seller if (a) there has been a material breach by Seller or Seller Parent of any representation, warranty or covenant contained in this Agreement which breach, if curable prior to the Outside Date, has not been cured within the earlier of (x) ten Business Days following notice of the breach from Buyer to Seller and (y) the period of time that remains between the date such written notice is provided and the Outside Date, which would result in the failure of any of the conditions set forth in Section 6.2 or Section 6.3 and Buyer has not waived such condition in writing; provided, however, that Buyer may not terminate this Agreement pursuant to this Section 8.3 if Buyer is then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement (or if any representations or warranties of Buyer have become, after the Effective Date, and at that time remain, untrue) such that it would result in the failure of any of the conditions set forth in Section 6.1 or Section 6.3; or (b) at any time prior to obtaining the Stockholder Approval, in the event that (x) the board of directors of Seller Parent shall have failed to include the Board Recommendation in the Proxy Statement or shall have effected an Adverse Recommendation Change or (y) Seller or Seller Parent shall have breached Section 5.2 in any material respect.
- 8.4 Termination by Seller. This Agreement may be terminated and the transactions contemplated by this Agreement may be abandoned at any time prior to the Closing by Seller upon written notice to Buyer (a) if there has been a material breach by Buyer of any representation, warranty or covenant contained in this Agreement which breach, if curable prior to the Outside Date has not been cured within the earlier of (x) ten Business Days following notice of the breach from Seller to Buyer and (y) the period of time that remains between the date such written notice is provided and the Outside Date, which would result in the failure of any of the conditions set forth in Section 6.1 or Section 6.3 and neither Seller nor Seller Parent has waived such condition in writing; provided, however, that Seller may not terminate this Agreement pursuant to this Section 8.4 if either Seller or Seller Parent is then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement (or if any representations or warranties of Seller or Seller Parent have become after the Effective Date, and at that time remain, untrue) such that it would result in the failure of any of the conditions set forth in Section 6.2 or Section 6.3, or (b) at any time prior to obtaining the Stockholder Approval, in order to enter into a binding agreement for a Superior Proposal, if prior to or concurrently with such termination, Seller or Seller Parent pays the Seller Termination Fee pursuant to Section 8.6(a), provided that such Superior Proposal was not solicited in violation ofSection 5.2.
- **8.5** Effect of Termination and Abandonment. In the event of termination of this Agreement and the abandonment of the transactions contemplated by this Agreement pursuant to this Article VIII, no Party (or any of its directors or officers) shall have any liability or further obligation to any Party, except (a) with respect to Section 2.3(f) through 2.3(h), Section 7.2, Section 7.4, Section 7.5, Section 7.13, this Section 8.5, Section 8.6 and Article X, and (b) except as set forth inSection 2.3(h), that termination of this Agreement will not preclude a Party from bringing a claim against any other Party for a breach arising prior to such termination pursuant to the terms and conditions set forth herein, and nothing herein will relieve any Party from liability for any material breach of this Agreement, subject to any limitations on liability or remedies set forth in this Agreement.

8.6 Termination Fee.

- (a) Seller Parent shall pay to Buyer \$5,000,000 (the "Seller Termination Fee") if (x) this Agreement is terminated pursuant to Section 8.3(b) or Section 8.4(b), or (y) (i) after the date of this Agreement, and prior to the termination of this Agreement, a proposal for an Alternative Transaction shall have been made to Seller Parent or Seller, or an intention (whether or not conditional) to make a proposal for an Alternative Transaction was publicly announced or otherwise became publicly known; (ii) thereafter, this Agreement is terminated (A) by Buyer or Seller pursuant to Section 8.2(a) or Section 8.2(c), or (B) by Buyer pursuant to Section 8.3(a); and (iii) within twelve (12) months after such termination, Seller enters into a definitive agreement to effect an Alternative Transaction or consummates an Alternative Transaction.
- (b) In the event that this Agreement is terminated by Buyer or Seller pursuant to Section 8.2(c), Seller Parent shall pay to Buyer all of the reasonable and documented out-of-pocket expenses incurred by Buyer and its Affiliates in connection with this Agreement and the transactions contemplated thereby up to a maximum of \$5,000,000 (the "Expense Reimbursement"). In no event shall Seller or Seller Parent pay both the Seller Termination Fee and the Expense Reimbursement.
- (c) The amounts due pursuant to this <u>Section 8.6</u> shall be paid within two (2) Business Days after termination of this Agreement in immediately available funds in lawful currency of the United States of America to the account designated by Buyer.
- (d) If Seller Parent fails to promptly pay any amounts due pursuant to this Section 8.6 and, in order to obtain such payment, Buyer commences a suit that results in a final and non-appealable judgment against Seller Parent, for payment of amounts owed pursuant to this Section 8.6, then Seller Parent shall pay to Buyer its reasonable and documented out-of-pocket costs and expenses (including reasonable and documented out-of-pocket attorneys' fees) in connection with such suit, together with interest on such amount at the prime lending rate prevailing during such period as published in The Wall Street Journal, calculated on a daily basis from the date that such amounts were required to be paid to the date of actual payment; provided, that if such suit results in a final and non-appealable judgment against Buyer, then Buyer shall pay to Seller Parent its reasonable and documented out-of-pocket costs and expenses (including reasonable and documented out-of-pocket attorneys' fees) in connection with such suit, together with interest on such amount at the prime lending rate prevailing during such period as published in The Wall Street Journal, calculated on a daily basis from the date that such amounts were required to be paid to the date of actual payment. Any amount payable pursuant to this Section 8.6 shall be paid by wire transfer of immediately available funds prior to or on the date such payment is required to be made under this Section 8.6.
- (e) The Parties acknowledge and agree that the agreements contained in this Section 8.6 are an integral part of this Agreement, and that, without these agreements, the parties would not enter into this Agreement. Each of the Parties further acknowledges that the payment of the amounts specified in this Section 8.6 is not a penalty, but, in each case, is liquidated damages in a reasonable amount that will compensate such Party in the circumstances in which such amounts are payable for the efforts and resources expended and the opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement, which amount would otherwise be impossible to calculate with precision. In no event shall Seller Parent be required to pay the Seller Termination Fee in connection with the termination of this Agreement more than once.
- (f) Notwithstanding anything to the contrary set forth in this Agreement or any other Transaction Document, but subject to Section 10.8, each of the Parties acknowledges and agrees that in the event the Seller Termination Fee or the Expense Reimbursement is paid to Buyer in circumstances in which such amounts are payable pursuant to this Section 8.6 payment of the Seller Termination Fee or the Expense Reimbursement, as applicable, pursuant to this Section 8.6 (and any costs, expenses and interest payable by Seller Parent to Buyer pursuant to Section 8.6(d)) shall constitute the sole and exclusive monetary remedy of Buyer, and its Affiliates and former, current or future stockholders, directors, employees, agents or Representatives, against Seller Parent, and its Affiliates and former, current or future stockholders, directors, officers, employees, agents or Representatives ("Related Parties"), for all Losses and damages in respect of this Agreement, the transactions contemplated by this Agreement, any breach of any covenant or agreement or otherwise in respect

of this Agreement, or the failure to consummate the transactions contemplated hereby, and upon payment the amounts due pursuant to this Section 8.6, neither Seller Parent nor its Related Parties, shall have any further liability or obligation relating to or arising out of this Agreement, the Transaction Documents or the transactions contemplated hereby or thereby (except that the Parties will remain subject to the Confidentiality Agreement and any covenants or obligations that expressly survive the termination of this Agreement), and none of Buyer or its Related Parties shall seek to recover any other damages or seek any other remedy, whether based on a claim at law or in equity, in contract, tort or otherwise, with respect to any Losses suffered in connection with this Agreement or the transactions contemplated hereby (except with respect to the Confidentiality Agreement and any covenants or obligations that expressly survive the termination of this Agreement). For the avoidance of doubt, this Section 8.6(f) applies only in the event of a valid termination under this Article VIII. Notwithstanding the foregoing, nothing in this Section 8.6(f) shall relieve either Party or its Related Parties from liability for damages arising from Fraud by the other Party or its Related Parties.

(g) While Buyer may pursue a grant of specific performance or other equitable relief under Section 10.8 and payment of the Seller Termination Fee or Expense Reimbursement as applicable, pursuant to this Section 8.6, under no circumstances shall Buyer be permitted or entitled to receive both a grant of specific performance or other equitable relief and the payment of the Seller Termination Fee or the Expense Reimbursement, as applicable, in connection with the termination of this Agreement (except with respect to the Confidentiality Agreement and any covenants or obligations that expressly survive the termination of this Agreement).

ARTICLE IX INDEMNIFICATION

9.1 Indemnification by the Parties.

- (a) <u>Indemnification by Seller</u>. From and after the Closing, subject to <u>Section 9.2</u>, Seller shall indemnify and defend Buyer against, and shall hold Buyer and its Affiliates and their respective directors, officers, employees, Representatives and assigns (collectively, the "*Buyer Indemnified Parties*") harmless from and against, and shall reimburse the Buyer Indemnified Parties for, any and all Losses which the Buyer Indemnified Parties shall suffer or incur or shall have imposed upon them in connection with, based upon, related to or arising out of:
 - (i) any breach of or inaccuracy in any representation or warranty made by Seller or Seller Parent in this Agreement or in any Schedule or certificate delivered pursuant to this Agreement, as of the Effective date or as of the Closing Date;
 - (ii) any Excluded Liabilities or Excluded Assets or the Excluded Business;
 - (iii) any breach or default of, or failure by Seller or Seller Parent to perform, any covenant, agreement or obligation to be performed by Seller or Seller Parent pursuant to this Agreement; and
 - (iv) (A) any Taxes of Seller, Seller Parent or any Affiliate of Seller or Seller Parent that Buyer was required to withhold from any payments hereunder and which Buyer did not withhold and (B) any Taxes Seller or Seller Parent are responsible for pursuant to Section 7.5(a), Section 7.5(b), or Section 7.5(c).
- (b) <u>Indemnification by Buyer</u>. From and after the Closing, subject to <u>Section 9.2</u>, Buyer shall indemnify and hold harmless Seller and its Affiliates and their respective directors, officers, employees, Representatives and assigns (collectively, the "*Seller Indemnified Parties*") from and against any and all Losses which the Seller Indemnified Parties shall suffer or incur or shall have imposed upon them in connection with, based upon, related to or arising out of:
 - (i) any breach of or inaccuracy in any representation or warranty made by Buyer in this Agreement or in any Schedule or certificate delivered pursuant to this Agreement, as of the Effective Date or as of the Closing Date;
 - (ii) any Assumed Liabilities;

(iii) any breach or default of, or failure by Buyer to perform, any covenant, agreement or obligation to be performed by Buyer pursuant to this Agreement or any document delivered by Buyer in connection with this Agreement; and

(iv) any Taxes Buyer is responsible for pursuant to <u>Section 7.5(a)</u>, <u>Section 7.5(b)</u> or <u>Section 7.5(c)</u>.

9.2 Procedure for Indemnification Claims - Third Party Claims.

(a) Promptly after receipt by a Person entitled to indemnification under<u>Section 9.1</u> (the "Indemnitee") of notice of the commencement of any Action against it, such Indemnitee will, if a claim is to be made against a Party obligated to indemnify under such section (the "Indemnitor"), give notice within 20 days to the Indemnitor of the commencement of such Action, but the failure to notify the Indemnitor will not relieve the Indemnitor of any liability that it may have to any Indemnitee, except to the extent that the Indemnitor demonstrates that it is actually prejudiced by the Indemnitee's failure to give such notice, except that in no event may an Indemnitee make a claim for indemnification after the expiration of the applicable survival period in accordance with Section 9.5.

(b) If any Action is brought against an Indemnitee and it gives notice to the Indemnitor of the commencement of the Action, the Indemnitor will be entitled to participate in such Action and to assume the defense of such Action with counsel satisfactory to the Indemnitee at the Indemnitor's expense, if Indemnitor so notifies Indemnitee within 20 days after the Indemnitor receives notice of such Action; provided, however that the Indemnitor may not assume the defense of an Action if the Indemnitor is also a party to such Action and the Indemnitee determines in good faith that joint representation would be inappropriate. Notwithstanding the foregoing, no Indemnitor shall have the right to assume the defense of any third party claim (i) that is asserted directly or indirectly on behalf of a Person that is a current or prospective customer, third party payor or supplier of an Indemnitee, or (ii) if the third party claim (A) seeks non-monetary relief, (B) involves criminal or quasi criminal allegations, (C) involves a claim of which the Indemnitee reasonably believes an adverse determination would be materially detrimental or injurious to the Indemnitee, (D) involves a claim which, upon petition by the Indemnitee, the appropriate court or arbitrational body rules that the Indemnitor failed or is failing to vigorously prosecute or defend, or (E) involves a claim in an amount which, together with previous and pending claim amounts, would exceed the Purchase Price. In the event that the Indemnitee shall in good faith determine that the Indemnitee may have available to it one or more defenses or counterclaims that are inconsistent with one or more of those that may be available to the Indemnitor in respect of such third party claim, the Indemnitee shall have the right, but not the obligation, at all times to take over and assume control over the defense, settlement, negotiations or proceedings relating to any such third party claim; provided that if the Indemnitee does so take over and assume control, the Indemnitee shall not settle such third party claim without the prior written consent of the Indemnitor, not to be unreasonably withheld, conditioned or delayed. In the event that the Indemnitor assumes the defense of any third party claim, subject to this Article IX, it shall have the right to take such action as it deems necessary to avoid, dispute, defend, appeal or make counterclaims pertaining to any such third party claim in the name and on behalf of the Indemnitee. The Indemnitee shall have the right, at its own cost and expense, to participate in the defense of any third party claim with counsel selected by it subject to the Indemnitor's right to control the defense thereof. If the Indemnitor elects not to compromise or defend such third party claim or fails to promptly notify the Indemnitee in writing of its election to defend as provided in this Agreement, the Indemnitee may, subject to this Section 9.2(b), pay, compromise, defend such third party claim and seek indemnification for any and all Losses based upon, arising from or relating to such third party claim. If the Indemnitor assumes the defense of an Action, no compromise or settlement of such claims may be effected by the Indemnitor without the Indemnitee's consent unless (1) there is no finding or admission of any violation of any Law or order of any Governmental Authority or any violation of the rights of any Person and no effect on any other claims that may be made against the Indemnitee, and (2) the sole relief provided is monetary damages that are paid in full by the Indemnitor, and (3) the Indemnitee will have no liability or obligation with respect to any compromise or settlement of such claims effected without its consent, and such compromise or settlement provides for a complete, unconditional release in customary form from all obligations and liabilities of the Indemnitee with respect to such claim. With respect to any third party claim subject to indemnification under this Article IX, both the Indemnitee and the Indemnitor, as the case may be, shall

use commercially reasonable efforts keep the other Person fully informed in all material respects of the status of such third party claim and any related proceedings at all stages thereof where such Person is not represented by its own counsel.

- 9.3 Procedure for Indemnification Claims Other Claims. In the event of a claim for indemnification under this Article IX not based upon a third party claim (which is addressed in Section 9.2), the Indemnitor shall have 30 Business Days from receipt of Indemnitee's prompt written notice of such claim for indemnification (which notice shall specify in reasonable detail the basis of such claim and be accompanied by supporting documentation sufficient to permit the Indemnitor to evaluate such claim) to respond to such claim. If the Indemnitor has timely disputed the claim, in whole or in part with respect to such claim as provided above, the Indemnitor and the Indemnitee shall proceed in good faith to negotiate a resolution of such dispute for 60 days following receipt of the dispute notice by the Indemnitee, and if not resolved through such negotiations within such 60-day period, either party shall be entitled to pursue resolution of such dispute by litigation in accordance with this Agreement. Following the final resolution of the merits and amounts of such claim (whether by mutual agreement or litigation in accordance with this Agreement), the Indemnitor shall pay the amount of such Losses due and owing to the Indemnitee on demand.
- **9.4** Nature of Payments. Any indemnification payments made pursuant to this Article IX shall be treated for Tax purposes as an adjustment to the Purchase Price, including for purposes of Section 2.6, unless otherwise required by applicable Laws.
- 9.5 Survival. The representations and warranties of Seller and Seller Parent contained in this Agreement or in any certificate delivered at the Closing pursuant hereto shall survive the Closing for a period of 15 months; provided, however, that (a) the representations and warranties of Seller and Seller Parent contained in Section 3.5(f) through Section 3.5(g) (Legal Compliance), Section 3.8(a) (Good Title and Use of Assets), the second sentence of Section 3.8(b) (Good Title and Use of Assets), the first sentence of Section 3.28(b) (Intellectual Property), the second sentence of Section 3.28(d) (Intellectual Property), Section 3.1 (Organization, Standing and Power; No Subsidiaries), Section 3.2 (Authority), Section 3.3(b)(i) (No Conflict; Required Filings and Consents) and Section 3.15 (Brokers or Finders) (collectively, the "Fundamental Representations") shall survive the Closing until the later of: (i) the sixth anniversary of the Closing Date and (ii) the expiration of the applicable statute of limitations, plus 90 days and (b) the representations and warranties of Seller and Seller Parent contained in Section 3.18 (Tax Matters) shall survive the Closing until the expiration of the applicable statute of limitations, plus 90 days. The representations and warranties of Buyer shall survive the Closing for a period of 15 months following the Closing Date. The covenants and agreements contained in this Agreement which by their terms are to be performed prior to the Closing shall survive the Closing for a period of 6 months after the Closing Date and the covenants and other agreements contained in this Agreement which by their terms or as stated contemplate performance in whole or in part after the Closing shall survive the Closing for the period contemplated by its terms or as stated expressly or until fully performed or fulfilled. Any action based on a breach of representations and warranties made herein must be brought on or before such representations and warranties terminate in accordance with the preceding sentences of this Section 9.5 or forever be barred, regardless of whether a longer statute of limitations may exist by Law; provided, however, that if an action or claim is brought on or before such representations and warranties terminate, such action or claim shall survive until final settlement or judgment. The limitations in this Section 9.5 shall not apply in the case of Fraud.

9.6 Limitations on Liability; Order of Recourse; Recovery from Escrow Account

(a) Notwithstanding anything to the contrary contained in this Agreement:

(i) Seller shall not have any indemnity obligation under Section 9.1(a)(i) unless and until the Losses incurred by the Buyer Indemnified Parties with respect to any matter or series of related matters for which indemnification is to be provided under Section 9.1(a)(i) exceed \$584,000 (the "Deductible"), in which case Seller shall be liable for all Losses with respect to such matter or matters in excess of such amount (subject to all other applicable requirements and limitations herein); provided that the foregoing limitation shall not apply to any claim based on any inaccuracy in or breach of Fundamental Representation or the representations and warranties of Seller and Seller Parent in Section 3.18 (Tax Matters);

- (ii) except with respect to any claim by a Buyer Indemnified Party based on any inaccuracy in or breach of Fundamental Representations or the representations and warranties of Seller in <u>Section 3.18</u> (Tax Matters), the aggregate liability of Seller under <u>Section 9.1(a)(i)</u> in excess of the Deductible shall not exceed \$584,000 (the "*Cap*");
- (iii) except with respect to any claim by a Buyer Indemnified Party based on any inaccuracy in or breach of Fundamental Representations or the representations and warranties of Seller and Seller Parent in Section 3.18 (Tax Matters), any claim for indemnification under Section 9.1(a)(i) shall be satisfied (A) first, from the Escrow Fund in the Escrow Account, or if the Escrow Fund has been exhausted or is otherwise insufficient or unavailable to fully satisfy such claim, from Seller directly, until the retention under the R&W Policy in excess of the Deductible has been satisfied, and (B) second, from the R&W Policy;
- (iv) with respect to any claim by a Buyer Indemnified Party with respect to any Fundamental Representation or the representations and warranties of Seller and Seller Parent in Section 3.18 (Tax Matters) or pursuant to Section 9.1(a)(iv), subject to the other limitations set forth herein, the Buyer Indemnified Parties shall be entitled to satisfy the Losses therefrom (A) first, from the Escrow Fund in the Escrow Account, or if the Escrow Fund has been exhausted or is otherwise unavailable or insufficient to fully satisfy such claim, from Seller directly, until the retention under the R&W Policy has been satisfied, (B) second, from the R&W Policy if and to the extent recovery thereunder is available, and (C) third, from the Escrow Fund in the Escrow Account, or if the Escrow Fund has been exhausted or is otherwise unavailable or insufficient to fully satisfy such claim, from Seller directly up to the amount of the Purchase Price; and
- (v) with respect to any claim by a Buyer Indemnified Party pursuant to Section 9.1(a)(ii) through Section 9.1(a)(iv) Buyer shall be entitled to satisfy the Losses therefrom from the Escrow Fund in the Escrow Account, or if the Escrow Fund has been exhausted or is otherwise unavailable or insufficient to fully satisfy such claim, from Seller directly.
- (b) On or before the Business Day after the 15-month anniversary of the Closing Date (the "Release Date"), Buyer shall notify Seller in writing of the amount of all claims made by the Buyer Indemnified Parties pursuant to this Article IX (that have been validly asserted, but not fully and finally resolved prior to such date in accordance with this Article IX (such unresolved claims being referred to as the "Unresolved Claims" and such aggregate amount being referred to as the "Unresolved Claims Amount"). Promptly after the Release Date, Buyer and Seller shall jointly instruct the Escrow Agent to release to Seller from the Escrow Fund in accordance with the Escrow Agreement the portion of the Escrow Fund, if any, in excess of the sum of (i) the amount, if any, that Buyer and Seller have instructed the Escrow Agent to disburse from the Escrow Account that has not actually been disbursed as of such date, and (ii) the Unresolved Claims Amount as of the Release Date.
- (c) Following the release from the Escrow Fund in Section 9.6(b), if any Unresolved Claim is finally resolved, then Buyer and Seller shall, within ten Business Days after the final resolution of such Unresolved Claim and the delivery to the Buyer Indemnified Party of the amount to be delivered to the Buyer Indemnified Party from the Escrow Fund (and/or directly from Seller, as applicable) pursuant to this Article IX, Buyer and Seller shall jointly instruct the Escrow Agent to disburse from the Escrow Account to Seller, the portion of the Unresolved Claims Amount retained in the Escrow Account in respect of such Unresolved Claim in excess of the amount determined to constitute Losses with respect to Unresolved Claim, if any; provided, that in no event shall the aggregate amount disbursed to Seller with respect to any Unresolved Claim exceed the amount, if any, that Buyer and Seller have instructed the Escrow Agent to disburse from the Escrow Account but that has not actually been disbursed as of such date; provided, further, that, to the extent that any such release to Seller would result in the Escrow Fund at such time not being sufficient with respect to all other Unresolved Claims, if any, then the Escrow Agent shall continue to retain the portion of the Escrow Amount as Buyer reasonably determines in good faith may be the maximum amount payable to any Buyer Indemnified Party under the terms of this Article IX in respect of any such Unresolved Claims until resolution thereof. The procedures set forth in this Section 9.6(c) shall be reiterated for each Unresolved Claim until all amounts in the Escrow Fund have either been paid to the Buyer Indemnified Parties or released to Seller.

- (d) Any Indemnitee that becomes aware of any Losses for which it seeks indemnification hereunder shall be required to use commercially reasonable efforts to mitigate such Losses consistent with the common law doctrine of mitigation of damages and an Indemnitor shall not be liable for any Losses to the extent that it is attributable to the Indemnitee's failure to so mitigate; provided, that an Indemnitee's obligation to mitigate shall not include any obligation or requirement that such Indemnitee or any of his or its Affiliates assume or incur any liability.
- (e) From and after the Closing Date, subject to the terms and limitations set forth in this Agreement, Seller Parent, as a primary obligor and not as a surety, hereby absolutely, unconditionally and irrevocably guarantees to Buyer the complete and prompt payment (and not of collection) and performance of, and compliance by Seller of all of Seller's obligations under this Agreement (the "Guaranteed Obligations"). Seller Parent acknowledges and agrees that its guaranty is full, absolute and unconditional and is in no way conditioned or contingent upon any attempt to collect from Seller. Seller Parent understands that Buyer is relying on this guarantee in entering into this Agreement. Seller Parent hereby waives, for the benefit of Buyer, to the fullest extent permitted by applicable Law, any defenses or benefits that may be derived from or afforded by Law that limit the liability of or exonerate guarantors or sureties, including those which would otherwise require any election of remedies by Buyer (other than payment of the applicable Guaranteed Obligations) and further waives any right to require Buyer to proceed against Seller with respect to any of the applicable Guaranteed Obligations.
- 9.7 <u>Calculation of Losses</u>. For purposes of determining any inaccuracy or breach of any representation or warranty and calculating the amount of any Loss with respect thereto, such representation or warranty shall be read without giving effect to any qualifications as to materiality, Material Adverse Effect or similar materiality qualifications. In no event will any party be entitled to recover or make a claim for indemnification under this <u>Article IX</u> for any amounts in respect of punitive damages other than those paid or required to be paid to a third party.
- 9.8 Insurance Proceeds. If a claim for indemnification is made by any Indemnitee pursuant to this Article IX in respect of any Losses that are covered by such Indemnitee's insurance policies (including any tail insurance policy), the amount of Losses shall be reduced by any third party insurance proceeds actually received by the Indemnitee from a third party, net of the cost of recovery (including reasonable attorneys' fees) and any retroactive premium adjustments or other increased insurance costs resulting therefrom; provided, however, that the Indemnitee shall have sole and absolute discretion to determine whether (a) to submit or pursue a claim for damages with the insurer under the applicable insurance policy and (b) in the case of a denial of coverage or issuance of reservation of rights, whether to challenge such denial or issuance.
- 9.9 Exclusive Remedies. After the Closing, the provisions of this Article IX shall constitute the sole and exclusive remedy (other than (a) with respect to Fraud, (b) injunctive relief or specific performance as contemplated by Section 10.8, (c) with respect to the covenants and agreements that by their respective terms anticipate performance following the Closing Date and (d) as otherwise set forth in this Agreement or any other Transaction Documents) of the Parties against each other with respect to any breach or non-fulfillment of any representation, warranty, agreement, covenant, condition or any other obligation contained in this Agreement or in any certificates or similar documents delivered pursuant to this Agreement. In furtherance of the foregoing, except for claims arising from Fraud, each Party hereby waives, to the fullest extent permitted by applicable Law, any and all other rights, claims and causes of action (including rights of contributions, if any) known or unknown, foreseen or unforeseen, which exist or may arise in the future, that it may have against Seller, Seller Parent or any of their respective Affiliates or Buyer or any of its Affiliates, as the case may be, arising under or based upon any Law. The obligations of the Parties set forth in this Section 9.9 shall be conditioned upon the Closing having occurred.

ARTICLE X GENERAL PROVISIONS

10.1 Notices. All notices and other communications hereunder shall be in writing and shall be deemed provided to a Party (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient or (d) on the third day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Electronic mail addresses of the Parties shall be exchanged prior to the execution of this Agreement. Notices and other communications to a Party shall be addressed as follows:

If to Seller or Seller Parent:

Enzo Biochem, Inc.

81 Executive Blvd., Suite # 3 Farmingdale, NY 11735

Attn: Hamid Erfanian; Matthew Kupferberg

Email: HErfanian@enzo.com; MKupferberg@enzo.com

With a copy to (which shall not constitute notice):

McDermott Will & Emery LLP

One Vanderbilt Avenue New York, NY 10017

Attn: Robert H. Cohen; Meir Lewittes

Email: rcohen@mwe.com; mlewittes@mwe.com

If to Buyer to:

Laboratory Corporation of America Holdings

531 South Spring Street

Burlington, NC 27215

Attn: Chief Legal Officer
Email: vaarts@labcorp.com

With a copy to (which shall not constitute notice):

Hogan Lovells US LLP 390 Madison Avenue New York, NY 10017

Attention: Michael J. Silver and Gabrielle M. Witt Email: michael.silver@hoganlovells.com

gabrielle.witt@hoganlovells.com

10.2 Interpretation. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. The Disclosure Schedules and Exhibits referred to herein shall be construed with, and as an integral part of, this Agreement to the same extent as if they were set forth verbatim herein.

10.3 Rules of Construction. The Parties agree that they have been represented by counsel during the negotiation, preparation and execution of this Agreement, and therefore waive the application of any Law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document. Unless otherwise qualified, all references in this Agreement that provide that information, documents or materials have been "made available to Buyer", "delivered to Buyer" or "provided to Buyer," and phrases of similar import, means that such information, document or material was made available for Buyer's review in the Data Room, prior to 5:00 P.M. Eastern Time on the Business Day prior to the Effective Date.

- **10.4 Entire Agreement**. This Agreement (including the documents and instruments referred to herein and the exhibits and schedules attached hereto and thereto) and the Confidentiality Agreement constitute the entire agreement, and supersede all other prior agreements and understandings (both written and oral), among the Parties, or any of them, with respect to the subject matter hereof.
- 10.5 Jurisdiction; Governing Law. This Agreement, and all claims or causes of action (whether in contract, tort or statute) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by, and enforced in accordance with, the internal laws of the State of Delaware, including its statutes of limitations. Any legal action or other proceeding brought for the enforcement of this Agreement or because of an alleged breach, dispute, default or misrepresentation regarding any provision of this Agreement shall be brought exclusively in the state and federal courts sitting in The City of New York, Borough of Manhattan, and any appellate courts from any appeal thereof, and each Party irrevocably submits to the jurisdiction of each such courts in any such proceeding, and waives any objection it may now or after the Effective Date have to venue or to convenience of forum. The Parties hereto agree that any of them may file a copy of this paragraph with any court as written evidence of the knowing, voluntary and bargained agreement of the Parties hereto irrevocably to waive any objections to venue or to convenience of forum. Process in any such proceeding may be served on any Party hereto anywhere in the world.
- 10.6 Waiver of Jury Trial. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT OR THE OTHER TRANSACTION DOCUMENTS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES AND, THEREFORE, EACH SUCH PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LEGAL ACTION ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY TO THIS AGREEMENT CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT SEEK TO ENFORCE THE FOREGOING WAIVER IN THE EVENT OF A LEGAL ACTION, (B) SUCH PARTY HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.6.
- 10.7 Severability. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is deemed unenforceable, such provision shall be deemed to be reformed and modified to the minimum amount required to make such provision enforceable.
- 10.8 Equitable Relief. The Parties hereto agree that irreparable damage would occur in the event that any Party fails to consummate the transactions contemplated by this Agreement in accordance with the terms of this Agreement and that the Parties shall be entitled to specific performance in such event, in addition to any other remedy at law or in equity. The Parties hereto agree that, in the event of any breach or threatened breach by another Party of any covenant or obligation contained in this Agreement, including the obligation of the Parties to consummate each Closing as required hereunder, the non-breaching Party shall be entitled to obtain (a) a decree or order of specific performance to enforce the observance and performance of such covenant or obligation, and (b) an injunction restraining such breach or threatened breach. In circumstances where a Party is obligated to consummate the transactions contemplated by this Agreement and such transactions have not been consummated, each of the Parties expressly acknowledges and agrees that the other Party shall have suffered irreparable harm, that monetary damages will be inadequate to compensate such other Party, and that such other Party shall be entitled to seek specific performance of such Party's obligation to consummate such transactions. The Parties further agree that neither the other Party, nor any other Person, shall be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section 10.8, and both Parties irrevocably waive any right they may have to require the obtaining, furnishing or posting of any such bond or similar instrument. No exercise of a remedy available to a Party shall be deemed an election excluding any other remedy available to such Party (any such claim by any other Party being hereby waived).

- 10.9 Expenses. Except as otherwise expressly provided herein, each Party shall pay their own costs and expenses relating to this Agreement and the other Transaction Documents, the negotiations leading up to this Agreement, and the performance of this Agreement and the other Transaction Documents.
- 10.10 Assignment. No Party may assign or delegate any rights or obligations set forth in this Agreement without the prior written consent of the other Party, except that Buyer may assign any of its rights and delegate any of its obligations under this Agreement (a) to any Affiliate of Buyer or (b) in connection with the sale of all or substantially all of the assets of or any business combination transaction involving Buyer. No assignments shall relieve the assigning Party of any of its obligations hereunder. This Agreement shall be binding upon the Parties and their respective successors and permitted assigns.
- 10.11 No Third Party Beneficiaries Except as set forth in Article IX, this Agreement is for the sole benefit of Seller, Seller Parent and Buyer and their respective successors and permitted assigns, and is not for the benefit of any third party. Except as set forth in Article IX, nothing contained in this Agreement is intended to relieve or discharge the obligations or liability of any third party to Seller, Seller Parent or Buyer, nor shall this Agreement give any third party any right of subrogation or action or any other rights over or against Seller, Seller Parent or Buyer.
- 10.12 Further Assurances and Records. From time to time after the Closing, each Party, at the request of the other Party, and without further consideration, shall sign and deliver or cause to be signed and delivered such other instruments of transfer and take such other actions as reasonably may be requested by the other in order further to effectuate the consummation of the transactions contemplated by this Agreement. In addition, the Parties acknowledge that there may be occasions in the future when a Party or Affiliates of a Party will need access to certain documentation of the other in order to prepare financial statements, Tax Returns or other reports to third parties, or in order to facilitate audits or legal proceedings, comply with Laws or Permits or otherwise conduct its affairs in a proper manner. Accordingly, the Parties shall exercise their respective reasonable efforts to achieve the purposes of this Section 10.12; provided, that such access or disclosure may be restricted by Seller to the extent, in its reasonable judgment, may be required to preserve any applicable attorney-client privilege, to comply with any applicable Law, or to comply with any confidentiality obligation with any third party.
- 10.13 Amendments and Waivers. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing signed by Buyer, Seller and Seller Parent. No waiver by any Party of any provision of this Agreement or any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be valid unless the same shall be in writing and signed by Seller Parent (if such waiver is made by Seller or Seller Parent) or Buyer (if such waiver is made by Buyer) nor shall such waiver be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.
- 10.14 <u>Independence of Covenants</u>. All covenants hereunder shall be given independent effect so that if a certain action or condition constitutes a default under a certain covenant, the fact that such action or condition is permitted by another covenant shall not affect the occurrence of such default, unless expressly permitted under an exception to such initial covenant. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation and warranty hereunder.
- 10.15 Attorney-Client Privilege and Conflict Waiver. McDermott Will & Emery LLP ("Seller Counsel") has represented Seller and Seller Parent in connection with the transactions contemplated by this Agreement and the other Transaction Documents. Buyer agrees and acknowledges that neither it nor its Affiliates or Representatives will seek to have Seller Counsel disqualified from representing Seller, Seller Parent or their respective Affiliates or Representatives in any dispute (whether in contract or tort) that may arise between Seller, Seller Parent, or any of their respective Affiliates or Representatives, on the one hand, and Buyer or its Affiliate or Representatives, on the other hand, based upon, arising out of or related to this Agreement or the other Transaction Documents or any of the transactions contemplated hereby or thereby (including in relation to the Business, the Acquired Assets, the Assumed Liabilities, the Excluded Assets or the Excluded Liabilities), in whole or in part, and Seller or Seller Parent shall have the right to decide whether or not to waive any attorney-client privilege that may apply to any communications between Seller or Seller Parent, on the one hand, and Seller Counsel, on the other hand. Buyer further agrees that, as to all pre-Closing privileged or otherwise protected materials relating to the Business, the Acquired Assets, the Assumed Liabilities, the Excluded Assets or the Excluded Liabilities, or relating to Seller, Seller Parent and their

Affiliates, including materials protected by the attorney-client privilege, work product, the expectation of client confidence and all other rights to any applicable privilege or immunity are hereby deemed the sole property of Seller, Seller Parent or such Affiliate, as applicable, and shall only be controlled, reviewed, utilized or otherwise accessed by thereby and shall not pass to or be claimed by Buyer on any ground, including waiver.

10.16 <u>Counterparts</u>. This Agreement may be executed and delivered in counterpart signature pages executed and delivered via electronic mail, and any such counterpart executed and delivered via electronic mail will be deemed an original for all intents and purposes.

SIGNATURES ON NEXT PAGE

IN WITNESS WHEREOF, the Parties have caused this Agreement to be signed as of the Effective Date.

BUYER:

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: /s/ Sandra D. van der Vaart

Name: Sandra D. van der Vaart

Title: Executive Vice President, Chief Legal Officer,

Chief Compliance Officer and Secretary

SELLER:

ENZO CLINICAL LABS, INC.

By: /s/ Hamid Erfanian

Name: Hamid Erfanian

Title: Chief Executive Officer

SELLER PARENT:

ENZO BIOCHEM, INC.

By: /s/ Hamid Erfanian

Name: Hamid Erfanian

Title: Chief Executive Officer

[Signature Page to Asset Purchase Agreement]

Schedule 1.1(b)(vi)

Schedule 1.1(b)(vi)

Prepaid Expenses

Schedule 1.1(b)(vx)

Laboratory-Related IT Assets

Schedule 1.1(0) Assigned Contracts

Schedule 1.1(uuu)

Excluded Liabilities

Schedule 1.1(nnnnn)

Permitted Liens

Schedule 2.6

Schedule 2.8(a)(iv)

Released Liens

Schedule 5.8

Schedule 5.11

Schedule 7.14(a)

EXHIBIT A

FORM OF ASSIGNMENT AND ASSUMPTION OF LEASE

EXHIBIT B

FORM OF ASSIGNMENT, ASSUMPTION AND BILL OF SALE

EXHIBIT C

FORM OF ESCROW AGREEMENT

EXHIBIT D

FORM OF INTELLECTUAL PROPERTY AGREEMENT

Opinion of Jefferies LLC

March 15, 2023

The Board of Directors Enzo Biochem, Inc. 81 Executive Blvd., Suite 3 Farmingdale, NY 11735

The Board of Directors:

We understand that Enzo Biochem, Inc., a New York corporation ("Enzo Biochem"), Enzo Clinical Labs, Inc., a New York corporation and wholly owned subsidiary of Enzo Biochem ("Enzo Clinical Labs"), and Laboratory Corporation of America Holdings, a Delaware corporation ("Labcorp"), propose to enter into an Asset Purchase Agreement (the "Agreement") pursuant to which, among other things, Enzo Clinical Labs and, solely with respect to certain intellectual property assets, Enzo Biochem and its affiliates, will sell to Labcorp on a debt-free and cash-free basis Enzo Biochem's business, conducted through Enzo Clinical Labs, of performing diagnostics tests in clinical and anatomical pathology laboratories and providing related laboratory testing support services (the "Clinical Lab Business" and, such sale, the "Transaction"). As more fully described in the Agreement, Enzo Clinical Labs and Enzo Biochem and its affiliates will sell to Labcorp specified assets, and Labcorp will assume specified liabilities, related to the Clinical Lab Business for an aggregate purchase price of \$146 million in cash (the "Purchase Price"), subject to certain adjustments, allocations and escrow and ticking fee arrangements, as specified in the Agreement (as to which adjustments, allocations and escrow and ticking fee arrangements we express no opinion). The terms and conditions of the Transaction are more fully set forth in the Agreement and related documents.

You have asked for our opinion as to whether the Purchase Price to be received for the Clinical Lab Business in the Transaction pursuant to the Agreement is fair, from a financial point of view, to Enzo Biochem.

In arriving at our opinion, we have, among other things:

- (i) reviewed a draft, dated March 15, 2023, of the Agreement;
- (ii) reviewed certain publicly available information relating to the Clinical Lab Business;
- (iii) reviewed certain internal financial and other information furnished to us by the management of Enzo Biochem relating to the Clinical Lab Business and its operations and prospects on a standalone basis, including certain historical financial and other information, and financial forecasts and estimates, provided to or discussed with us by the management of Enzo Biochem;
- (iv) held discussions with members of the senior management of Enzo Biochem regarding the Clinical Lab Business and its operations and prospects and the other matters described in clauses (ii) and (iii) above;
- (v) considered the discussions undertaken at the direction of Enzo Biochem with selected third parties to solicit indications of interest in the possible acquisition of all or a portion of Enzo Biochem;
- (vi) reviewed the implied trading multiples of certain publicly traded companies with operations that we deemed relevant in evaluating the Clinical Lab Business;
- (vii) reviewed, to the extent publicly available, financial terms of certain transactions that we deemed relevant in evaluating the Transaction; and
- (viii) conducted such other financial studies, analyses and investigations as we deemed appropriate.

The Board of Directors Enzo Biochem, Inc. March 15, 2023

In our review and analysis and in rendering this opinion, we have assumed and relied upon, but have not assumed any responsibility to independently investigate or verify, the accuracy and completeness of all financial and other information that was supplied or otherwise made available by or on behalf of Enzo Biochem or that was publicly available to us (including, without limitation, the information described above) or otherwise reviewed by us. We have relied on assurances of the management and other representatives of Enzo Biochem that they are not aware of any facts or circumstances that would make such information incomplete, inaccurate or misleading. We have not made or obtained an independent evaluation or appraisal of or relating to any of the assets or liabilities (contingent, accrued, derivative, off-balance sheet or otherwise), nor have we conducted a physical inspection of any of the properties or facilities, of the Clinical Lab Business or any entity, and we assume no responsibility to obtain or conduct any such evaluations, appraisals or physical inspections. We also have not evaluated the solvency of the Clinical Lab Business or any entity under any state or federal laws relating to bankruptcy, insolvency or similar matters. In addition, our analyses and opinion do not consider any actual or potential arbitration, litigation, audits, claims or possible unasserted claims, investigations or other proceedings involving or affecting the Clinical Lab Business or any entity.

We have been advised that there are no audited standalone financial statements relating solely to the Clinical Lab Business and we have assumed that if such financial statements were available, they would not reflect any information that would be meaningful in any respect to our analyses or opinion. We also have assumed that there have been no material changes in the assets or liabilities associated with, or in the financial condition or prospects of, the Clinical Lab Business since the dates on which financial and other information relating to the Clinical Lab Business were last made available to us that would be meaningful in any respect to our analyses or opinion. With respect to the financial forecasts and estimates provided to and reviewed by us, we note that projecting future results of any company or business is inherently subject to uncertainty. However, we have been advised, and we have assumed, that the financial forecasts and estimates relating to the Clinical Lab Business that we have been directed to utilize for purposes of our analyses and opinion have been reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of Enzo Biochem as to, and are an appropriate basis upon which to evaluate, the future financial performance of the Clinical Lab Business and the other matters covered thereby. We express no opinion as to any financial forecasts or estimates or the assumptions on which they are based.

We have relied upon the assessments of the management of Enzo Biochem as to, among other things, (1) the potential impact on the Clinical Lab Business of market, competitive, seasonal, macroeconomic and other conditions, trends and developments in and prospects for, and governmental, regulatory and legislative matters relating to or otherwise affecting, the clinical lab services industry, including healthcare reform regulations and changes in healthcare reimbursement models and products, government payment and reimbursement systems and third-party payor policies, (ii) implications of the global COVID-19 pandemic on the Clinical Lab Business, (iii) the ability to operate, secure leases and maintain sufficient patient service centers and other facilities for the operations of the Clinical Lab Business, (iv) required certifications and licenses for, and the technology and other intellectual property utilized in, the Clinical Lab Business and associated risks, including the validity and duration of such certifications, licenses, technology and other intellectual property, and (v) existing and future agreements and other arrangements involving, and ability to attract, retain and/or replace, key employees, customers, suppliers and other commercial relationships relating to the Clinical Lab Business. We have assumed that there will no developments with respect to any such matters, or any adjustments to or allocations of the Purchase Price, that would have an adverse effect on the Clinical Lab Business or the Transaction or that otherwise would be meaningful in any respect to our analyses or opinion. We also have assumed that the full escrowed portion of the Purchase Price will be received by Enzo Clinical Labs in connection with the

Our opinion is based on economic, monetary, regulatory, market and other conditions existing, and which can be evaluated, as of the date hereof. We expressly disclaim any undertaking or obligation to advise any person of any change in any fact or matter affecting our opinion of which we become aware after the date hereof. As you are aware, the credit, financial and stock markets, and the industry in which the Clinical Lab Business operates, have experienced and may continue to experience volatility and we express no view or opinion as to any potential effects of such volatility on the Clinical Lab Business or the Transaction.

The Board of Directors Enzo Biochem, Inc. March 15, 2023

We have made no independent investigation of, and we express no view or opinion as to, any legal, regulatory, accounting or tax matters affecting or relating to the Clinical Lab Business or the Transaction and we have assumed the correctness in all respects meaningful to our analyses and opinion of all legal, regulatory, accounting and tax advice given to Enzo Biochem and/or the Board of Directors of Enzo Biochem (the "Board"), including, without limitation, with respect to changes in, or the impact of, accounting standards or tax and other laws, regulations and governmental and legislative policies affecting the Clinical Lab Business or the Transaction and legal, regulatory, accounting and tax consequences to Enzo Biochem or its securityholders or to Enzo Clinical Labs of the terms of, and transactions contemplated by, the Agreement and related documents. We have assumed that the Transaction will be consummated in accordance with its terms without waiver, modification or amendment of any material term, condition or agreement and in compliance with all applicable laws, documents and other requirements and that, in the course of obtaining the necessary governmental, regulatory or third-party approvals, consents, waivers and releases for the Transaction or otherwise, including with respect to any divestitures or other requirements, no delay, limitation, restriction or condition will be imposed or occur that would have an adverse effect on the Clinical Lab Business or the Transaction or that otherwise would be meaningful in any respect to our analyses or opinion. We also have assumed that appropriate indemnification and other arrangements or provisions have been or will be made in respect of liabilities to be assumed by Labcorp in connection with the Transaction and that neither Enzo Biochem nor Enzo Clinical Labs will directly or indirectly incur or retain any liabilities or other obligations that are contemplated to be excluded from the Transaction or otherwise. We further have assumed that the final Agreement, when signed by the parties thereto, will not differ from the draft thereof reviewed by us in any respect meaningful to our analyses or

Our opinion does not address the relative merits of the Transaction or other transactions contemplated thereby as compared to any alternative transaction or opportunity that might be available to Enzo Biochem or Enzo Clinical Labs or with respect to the Clinical Lab Business, nor does it address the underlying business decision by Enzo Biochem to engage in the Transaction or the terms of the Agreement or related documents, including the form or structure of the Purchase Price or the Transaction, any adjustments to or allocations of the Purchase Price or any term, aspect or implication of any indemnification, escrow or ticking fee arrangements, any voting, transition services, intellectual property or other ancillary agreements or any other agreement, arrangement or understanding entered into in connection with, or contemplated by or resulting from, the Transaction or otherwise. Our opinion is limited to the fairness, from a financial point of view and as of the date hereof, to Enzo Biochem of the Purchase Price (to the extent expressly specified herein). We have not been asked to, and our opinion does not, address the fairness, financial or otherwise, of any consideration to the holders of any class of securities, creditors or other constituencies of Enzo Biochem or any other party. Furthermore, we express no view or opinion as to the fairness, financial or otherwise, of the amount or nature of any compensation or other consideration payable to or to be received by any officers, directors or employees, or any class of such persons, in connection with the Transaction relative to the Purchase Price or otherwise. We also express no view or opinion as to the prices at which the securities of Enzo Biochem may trade or otherwise be transferable at any time, including following announcement or consummation of the Transaction. The issuance of our opinion has been authorized by the Fairness Committee of Jefferies LLC.

It is understood that our opinion is for the use and benefit of the Board (in its capacity as such) in its evaluation of the Purchase Price, from a financial point of view, to Enzo Biochem. Our opinion does not constitute a recommendation as to how the Board, any securityholder or any other party should vote or act with respect to the Transaction or any other matter.

We have been engaged to act as financial advisor to Enzo Biochem in connection with the Transaction and will receive a fee for our services, of which a portion is payable upon delivery of this opinion and the principal portion is contingent upon consummation of the Transaction. In addition, Enzo Biochem has agreed to reimburse us for expenses incurred in connection with our engagement and to indemnify us against liabilities arising out of or in connection with the services rendered and to be rendered by us under such engagement.

The Board of Directors Enzo Biochem, Inc. March 15, 2023

As you are aware, we currently are acting as financial advisor to Enzo Biochem in connection with a strategic review and related matters, and we and our affiliates in the future may provide other financial advisory services or financing services to Enzo Biochem and/or its affiliates, for which services we and/or our affiliates would expect to receive compensation. As you also are aware, although we and our affiliates have not provided financial advisory or financing services to Labcorp during the past two years for which we and our affiliates received compensation, we and our affiliates in the future may provide such services to Labcorp and/or its respective affiliates, for which services we and/or our affiliates would expect to receive compensation. In the ordinary course of business, we and our affiliates may trade or hold securities or financial instruments (including loans and other obligations) of Enzo Biochem, Labcorp and/or their respective affiliates for our own account and for the accounts of our customers and, accordingly, may at any time hold long or short positions or otherwise effect transactions in those securities or financial instruments.

Based upon and subject to the foregoing, we are of the opinion that, as of the date hereof, the Purchase Price to be received for the Clinical Lab Business in the Transaction pursuant to the Agreement is fair, from a financial point of view, to Enzo Biochem.

Very truly yours,

JEFFERIES LLC

Annex B-4

New York Business Corporation Law Sections 623 and 910

- § 623. Procedure to enforce shareholder's right to receive payment for shares
- (a) A shareholder intending to enforce his right under a section of this chapter to receive payment for his shares if the proposed corporate action referred to therein is taken shall file with the corporation, before the meeting of shareholders at which the action is submitted to a vote, or at such meeting but before the vote, written objection to the action. The objection shall include a notice of his election to dissent, his name and residence address, the number and classes of shares as to which he dissents and a demand for payment of the fair value of his shares if the action is taken. Such objection is not required from any shareholder to whom the corporation did not give notice of such meeting in accordance with this chapter or where the proposed action is authorized by written consent of shareholders without a meeting.
- (b) Within ten days after the shareholders' authorization date, which term as used in this section means the date on which the shareholders' vote authorizing such action was taken, or the date on which such consent without a meeting was obtained from the requisite shareholders, the corporation shall give written notice of such authorization or consent by registered mail to each shareholder who filed written objection or from whom written objection was not required, excepting any shareholder who voted for or consented in writing to the proposed action and who thereby is deemed to have elected not to enforce his right to receive payment for his shares
- (c) Within twenty days after the giving of notice to him, any shareholder from whom written objection was not required and who elects to dissent shall file with the corporation a written notice of such election, stating his name and residence address, the number and classes of shares as to which he dissents and a demand for payment of the fair value of his shares. Any shareholder who elects to dissent from a merger under section 905 (Merger of subsidiary corporation) or paragraph (c) of section 907 (Merger or consolidation of domestic and foreign corporations) or from a share exchange under paragraph (g) of section 913 (Share exchanges) shall file a written notice of such election to dissent within twenty days after the giving to him of a copy of the plan of merger or exchange or an outline of the material features thereof under section 905 or 913.
- (d) A shareholder may not dissent as to less than all of the shares, as to which he has a right to dissent, held by him of record, that he owns beneficially. A nominee or fiduciary may not dissent on behalf of any beneficial owner as to less than all of the shares of such owner, as to which such nominee or fiduciary has a right to dissent, held of record by such nominee or fiduciary.
- (e) Upon consummation of the corporate action, the shareholder shall cease to have any of the rights of a shareholder except the right to be paid the fair value of his shares and any other rights under this section. A notice of election may be withdrawn by the shareholder at any time prior to his acceptance in writing of an offer made by the corporation, as provided in paragraph (g), but in no case later than sixty days from the date of consummation of the corporate action except that if the corporation fails to make a timely offer, as provided in paragraph (g), the time for withdrawing a notice of election shall be extended until sixty days from the date an offer is made. Upon expiration of such time, withdrawal of a notice of election shall require the written consent of the corporation. In order to be effective, withdrawal of a notice of election must be accompanied by the return to the corporation of any advance payment made to the shareholder as provided in paragraph (g). If a notice of election is withdrawn, or the corporate action is rescinded, or a court shall determine that the shareholder is not entitled to receive payment for his shares, or the shareholder shall otherwise lose his dissenters' rights, he shall not have the right to receive payment for his shares and he shall be reinstated to all his rights as a shareholder as of the consummation of the corporate action, including any intervening preemptive rights and the right to payment of any intervening dividend or other distribution or, if any such rights have expired or any such dividend or distribution other than in cash has been completed, in lieu thereof, at the election of the corporation, the fair value thereof in cash as determined by the board as of the time of such expiration or completion, but without prejudice otherwise to any corporate proceedings that may have been taken in the interim.
- (f) At the time of filing the notice of election to dissent or within one month thereafter the shareholder of shares represented by certificates shall submit the certificates representing his shares to the corporation, or to its transfer agent, which shall forthwith note conspicuously thereon that a notice of election has been filed and shall return the certificates to the shareholder or other person who submitted them on his behalf. Any shareholder of shares

represented by certificates who fails to submit his certificates for such notation as herein specified shall, at the option of the corporation exercised by written notice to him within forty-five days from the date of filing of such notice of election to dissent, lose his dissenter's rights unless a court, for good cause shown, shall otherwise direct. Upon transfer of a certificate bearing such notation, each new certificate issued therefor shall bear a similar notation together with the name of the original dissenting holder of the shares and a transferee shall acquire no rights in the corporation except those which the original dissenting shareholder had at the time of transfer.

- (g) Within fifteen days after the expiration of the period within which shareholders may file their notices of election to dissent, or within fifteen days after the proposed corporate action is consummated, whichever is later (but in no case later than ninety days from the shareholders' authorization date), the corporation or, in the case of a merger or consolidation, the surviving or new corporation, shall make a written offer by registered mail to each shareholder who has filed such notice of election to pay for his shares at a specified price which the corporation considers to be their fair value. Such offer shall be accompanied by a statement setting forth the aggregate number of shares with respect to which notices of election to dissent have been received and the aggregate number of holders of such shares. If the corporate action has been consummated, such offer shall also be accompanied by (1) advance payment to each such shareholder who has submitted the certificates representing his shares to the corporation, as provided in paragraph (f), of an amount equal to eighty percent of the amount of such offer, or (2) as to each shareholder who has not yet submitted his certificates a statement that advance payment to him of an amount equal to eighty percent of the amount of such offer will be made by the corporation promptly upon submission of his certificates. If the corporate action has not been consummated at the time of the making of the offer, such advance payment or statement as to advance payment shall be sent to each shareholder entitled thereto forthwith upon consummation of the corporate action. Every advance payment or statement as to advance payment shall include advice to the shareholder to the effect that acceptance of such payment does not constitute a waiver of any dissenters' rights. If the corporate action has not been consummated upon the expiration of the ninety day period after the shareholders' authorization date, the offer may be conditioned upon the consummation of such action. Such offer shall be made at the same price per share to all dissenting shareholders of the same class, or if divided into series, of the same series and shall be accompanied by a balance sheet of the corporation whose shares the dissenting shareholder holds as of the latest available date, which shall not be earlier than twelve months before the making of such offer, and a profit and loss statement or statements for not less than a twelve month period ended on the date of such balance sheet or, if the corporation was not in existence throughout such twelve month period, for the portion thereof during which it was in existence. Notwithstanding the foregoing, the corporation shall not be required to furnish a balance sheet or profit and loss statement or statements to any shareholder to whom such balance sheet or profit and loss statement or statements were previously furnished, nor if in connection with obtaining the shareholders authorization for or consent to the proposed corporate action the shareholders were furnished with a proxy or information statement, which included financial statements, pursuant to Regulation 14A or Regulation 14C of the United States Securities and Exchange Commission. If within thirty days after the making of such offer, the corporation making the offer and any shareholder agree upon the price to be paid for his shares, payment therefor shall be made within sixty days after the making of such offer or the consummation of the proposed corporate action, whichever is later, upon the surrender of the certificates for any such shares represented by certificates.
- (h) The following procedure shall apply if the corporation fails to make such offer within such period of fifteen days, or if it makes the offer and any dissenting shareholder or shareholders fail to agree with it within the period of thirty days thereafter upon the price to be paid for their shares:
- (1) The corporation shall, within twenty days after the expiration of whichever is applicable of the two periods last mentioned, institute a special proceeding in the supreme court in the judicial district in which the office of the corporation is located to determine the rights of dissenting shareholders and to fix the fair value of their shares. If, in the case of merger or consolidation, the surviving or new corporation is a foreign corporation without an office in this state, such proceeding shall be brought in the county where the office of the domestic corporation, whose shares are to be valued, was located.
- (2) If the corporation fails to institute such proceeding within such period of twenty days, any dissenting shareholder may institute such proceeding for the same purpose not later than thirty days after the expiration of such twenty day period. If such proceeding is not instituted within such thirty day period, all dissenter's rights shall be lost unless the supreme court, for good cause shown, shall otherwise direct.

- (3) All dissenting shareholders, excepting those who, as provided in paragraph (g), have agreed with the corporation upon the price to be paid for their shares, shall be made parties to such proceeding, which shall have the effect of an action quasi in rem against their shares. The corporation shall serve a copy of the petition in such proceeding upon each dissenting shareholder who is a resident of this state in the manner provided by law for the service of a summons, and upon each nonresident dissenting shareholder either by registered mail and publication, or in such other manner as is permitted by law. The jurisdiction of the court shall be plenary and exclusive.
- (4) The court shall determine whether each dissenting shareholder, as to whom the corporation requests the court to make such determination, is entitled to receive payment for his shares. If the corporation does not request any such determination or if the court finds that any dissenting shareholder is so entitled, it shall proceed to fix the value of the shares, which, for the purposes of this section, shall be the fair value as of the close of business on the day prior to the shareholders' authorization date. In fixing the fair value of the shares, the court shall consider the nature of the transaction giving rise to the shareholder's right to receive payment for shares and its effects on the corporation and its shareholders, the concepts and methods then customary in the relevant securities and financial markets for determining fair value of shares of a corporation engaging in a similar transaction under comparable circumstances and all other relevant factors. The court shall determine the fair value of the shares without a jury and without referral to an appraiser or referee. Upon application by the corporation or by any shareholder who is a party to the proceeding, the court may, in its discretion, permit pretrial disclosure, including, but not limited to, disclosure of any expert's reports relating to the fair value of the shares whether or not intended for use at the trial in the proceeding and notwithstanding subdivision (d) of section 3101 of the civil practice law and rules.
- (5) The final order in the proceeding shall be entered against the corporation in favor of each dissenting shareholder who is a party to the proceeding and is entitled thereto for the value of his shares so determined.
- (6) The final order shall include an allowance for interest at such rate as the court finds to be equitable, from the date the corporate action was consummated to the date of payment. In determining the rate of interest, the court shall consider all relevant factors, including the rate of interest which the corporation would have had to pay to borrow money during the pendency of the proceeding. If the court finds that the refusal of any shareholder to accept the corporate offer of payment for his shares was arbitrary, vexatious or otherwise not in good faith, no interest shall be allowed to him.
- (7) Each party to such proceeding shall bear its own costs and expenses, including the fees and expenses of its counsel and of any experts employed by it. Notwithstanding the foregoing, the court may, in its discretion, apportion and assess all or any part of the costs, expenses and fees incurred by the corporation against any or all of the dissenting shareholders who are parties to the proceeding, including any who have withdrawn their notices of election as provided in paragraph (e), if the court finds that their refusal to accept the corporate offer was arbitrary, vexatious or otherwise not in good faith. The court may, in its discretion, apportion and assess all or any part of the costs, expenses and fees incurred by any or all of the dissenting shareholders who are parties to the proceeding against the corporation if the court finds any of the following: (A) that the fair value of the shares as determined materially exceeds the amount which the corporation offered to pay; (B) that no offer or required advance payment was made by the corporation; (C) that the corporation failed to institute the special proceeding within the period specified therefor; or (D) that the action of the corporation in complying with its obligations as provided in this section was arbitrary, vexatious or otherwise not in good faith. In making any determination as provided in clause (A), the court may consider the dollar amount or the percentage, or both, by which the fair value of the shares as determined exceeds the corporate offer.
- (8) Within sixty days after final determination of the proceeding, the corporation shall pay to each dissenting shareholder the amount found to be due him, upon surrender of the certificates for any such shares represented by certificates.
- (i) Shares acquired by the corporation upon the payment of the agreed value therefor or of the amount due under the final order, as provided in this section, shall become treasury shares or be cancelled as provided in section 515 (Reacquired shares), except that, in the case of a merger or consolidation, they may be held and disposed of as the plan of merger or consolidation may otherwise provide.

- (j) No payment shall be made to a dissenting shareholder under this section at a time when the corporation is insolvent or when such payment would make it insolvent. In such event, the dissenting shareholder shall, at his option:
- (1) Withdraw his notice of election, which shall in such event be deemed withdrawn with the written consent of the corporation; or
- (2) Retain his status as a claimant against the corporation and, if it is liquidated, be subordinated to the rights of creditors of the corporation, but have rights superior to the non-dissenting shareholders, and if it is not liquidated, retain his right to be paid for his shares, which right the corporation shall be obliged to satisfy when the restrictions of this paragraph do not apply.
- (3) The dissenting shareholder shall exercise such option under subparagraph (1) or (2) by written notice filed with the corporation within thirty days after the corporation has given him written notice that payment for his shares cannot be made because of the restrictions of this paragraph. If the dissenting shareholder fails to exercise such option as provided, the corporation shall exercise the option by written notice given to him within twenty days after the expiration of such period of thirty days.
- (k) The enforcement by a shareholder of his right to receive payment for his shares in the manner provided herein shall exclude the enforcement by such shareholder of any other right to which he might otherwise be entitled by virtue of share ownership, except as provided in paragraph (e), and except that this section shall not exclude the right of such shareholder to bring or maintain an appropriate action to obtain relief on the ground that such corporate action will be or is unlawful or fraudulent as to him.
- (l) Except as otherwise expressly provided in this section, any notice to be given by a corporation to a shareholder under this section shall be given in the manner provided in section 605 (Notice of meetings of shareholders).
- (m) This section shall not apply to foreign corporations except as provided in subparagraph (e) (2) of section 907 (Merger or consolidation of domestic and foreign corporations).
- § 910. Right of shareholder to receive payment for shares upon merger or consolidation, or sale, lease, exchange or other disposition of assets, or share exchange
- (a) A shareholder of a domestic corporation shall, subject to and by complying with section 623 (Procedure to enforce shareholder's right to receive payment for shares), have the right to receive payment of the fair value of his shares and the other rights and benefits provided by such section, in the following cases:
- (1) Any shareholder entitled to vote who does not assent to the taking of an action specified in clauses (A), (B) and (C).
- (A) Any plan of merger or consolidation to which the corporation is a party; except that the right to receive payment of the fair value of his shares shall not be available:
- (i) To a shareholder of the parent corporation in a merger authorized by section 905 (Merger of parent and subsidiary corporations), or paragraph (c) of section 907 (Merger or consolidation of domestic and foreign corporations); or
- (ii) To a shareholder of the surviving corporation in a merger authorized by this article, other than a merger specified in subclause (i), unless such merger effects one or more of the changes specified in subparagraph (b) (6) of section 806 (Provisions as to certain proceedings) in the rights of the shares held by such shareholder; or
- (iii) Notwithstanding subclause (ii) of this clause, to a shareholder for the shares of any class or series of stock, which shares or depository receipts in respect thereof, at the record date fixed to determine the shareholders entitled to receive notice of the meeting of shareholders to vote upon the plan of merger or consolidation, were listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc.

- (B) Any sale, lease, exchange or other disposition of all or substantially all of the assets of a corporation which requires shareholder approval under section 909 (Sale, lease, exchange or other disposition of assets) other than a transaction wholly for cash where the shareholders' approval thereof is conditioned upon the dissolution of the corporation and the distribution of substantially all of its net assets to the shareholders in accordance with their respective interests within one year after the date of such transaction.
- (C) Any share exchange authorized by section 913 in which the corporation is participating as a subject corporation; except that the right to receive payment of the fair value of his shares shall not be available to a shareholder whose shares have not been acquired in the exchange or to a shareholder for the shares of any class or series of stock, which shares or depository receipt in respect thereof, at the record date fixed to determine the shareholders entitled to receive notice of the meeting of shareholders to vote upon the plan of exchange, were listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc.
- (2) Any shareholder of the subsidiary corporation in a merger authorized by section 905 or paragraph (c) of section 907, or in a share exchange authorized by paragraph (g) of section 913, who files with the corporation a written notice of election to dissent as provided in paragraph (c) of section 623.
- (3) Any shareholder, not entitled to vote with respect to a plan of merger or consolidation to which the corporation is a party, whose shares will be cancelled or exchanged in the merger or consolidation for cash or other consideration other than shares of the surviving or consolidated corporation or another corporation.

Annex C-5

TO VOTE, MARK BLOCKS BELOW IN BLUE OR BLACK INK AS FOLLOWS:

Signature [PLEASE SIGN WITHIN BOX]

Date

ENZO BIOCHEM, INC. 81 EXECUTIVE BLVD., SUITE 3
FARMINGDALE, NY 11735



VOTE BY INTERNET

Before The Meeting - Go to www.proxyvote.com or scan the QR Barcode above

Use the Internet to transmit your voting instructions and for electronic delivery of information. Vote by 11:59 p.m. Eastern Time on May 21, 2023 for shares held directly and by 11:59 p.m. Eastern Time on May 17, 2023 for shares held in a 401(k) Plan. Have your proxy card in hand when you access the web site and follow the instructions to obtain your records and to create an electronic voting instruction form.

 $\textit{During The Meeting -} \textbf{Go to } \underline{\textbf{www.virtualshareholdermeeting.com/ENZ2023SM}}$

You may attend the meeting via the Internet and vote during the meeting. Have the information that is printed in the box marked by the arrow available and follow the instructions.

VOTE BY PHONE - 1-800-690-6903

Use any touch-tone telephone to transmit your voting instructions. Vote by 11:59 p.m. Eastern Time on May 21, 2023 for shares held directly and by 11:59 p.m. Eastern Time on May 17, 2023 for shares held in a 401(k) Plan. Have your proxy card in hand when you call and then follow the instructions.

VOTE BY MAIL
Mark, sign and date your proxy card and return it in the postage-paid envelope we have provided or return it to Vote Processing, c/o Broadridge, 51 Mercedes Way, Edgewood, NY 11717.

		P THIS POR	TION	FOR YOU	R RECORD
	THIS PROXY CARD IS VALID ONLY WHEN SIGNED AND DATED.		TURN	THIS POR	rtion onl'
ENZO	BIOCHEM, INC.			_	\neg
Th	e Board of Directors recommends you vote "FOR" Proposals 1 and 2.		For A	Against	Abstain
1.	To consider and vote on a proposal to approve the sale of substantially all the assets and assignment of certain liabilities of the Company's laboratory business and adopt the Asset Purchase Agreement, dated as of March 16, 2023 (the "Asset Sale Proposal"), and	clinical	0	0	0
2.	To consider and vote on a proposal to approve the adjournment of the Special Meeting, if necessary or appropriate, to solicit additional proxies are insufficient votes to approve the Asset Sale Proposal at the time of the Special Meeting (the "Adjournment Proposal").	f there	0	0	0
NC	TE: Please sign exactly as your name(s) appear(s) hereon. When signing as attorney, executor, administrator, trustee, guardian, custodian or other fic	luciary,			
ple pa	TE: Please sign exactly as your name(s) appear(s) hereon. When signing as attorney, executor, administrator, trustee, guardian, custodian or other ficase give full title as such. Joint owners should each sign personally. All holders must sign. If a corporation or partnership, please sign in full corporation and the such as a sign in full corporation or partnership, please sign in full corporation place.	rate or			

Signature (Joint Owners)

Date

PLEASE VOTE TODAY!

SEE REVERSE SIDE

FOR THREE EASY WAYS TO VOTE.

Important Notice Regarding the Availability of Proxy Materials for the Special Meeting:

The Notice and Proxy Statement and Annual Report are available at www.proxyvote.com.

V13593-S65479

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS OF ENZO BIOCHEM, INC. SPECIAL MEETING OF SHAREHOLDERS

The undersigned hereby appoints Hamid Erfanian as proxy, with full power of substitution, revoking all proxies previously given, and hereby authorizes him to represent and to vote, as designated on the reverse side of this form, all shares of common stock of Enzo Biochem, Inc. that the undersigned would be entitled to vote if present at the Special Meeting of Shareholders to be held at 9:00 a.m. Eastern Time on May 22, 2023 virtually at www.virtualshareholdermeeting.com/ENZ2023SM, and at any adjournment or postponement thereof (the "Special Meeting").

This proxy, when properly executed, will be voted in the manner specified on the reverse side. **IF THIS PROXY IS EXECUTED, BUT NO INSTRUCTION IS GIVEN, THIS PROXY WILL BE VOTED "FOR" PROPOSALS 1 AND 2.** The proposals are more fully described in the Proxy Statement. If a matter is not timely submitted pursuant to Rule 14a-4(c)(1), the proxy appointed in this proxy card will have discretionary authority to vote with respect to any such matter subsequently and properly raised at the Special Meeting. Only shareholders of record as of the close of business on April 17, 2023 may vote at the Special Meeting.

PLEASE MARK, SIGN, DATE AND RETURN THE PROXY CARD, USING THE ENCLOSED ENVELOPE.

YOUR VOTE IS IMPORTANT - PLEASE VOTE TODAY

Continued and to be signed and dated on reverse side